ON THE STRUCTURE OF ACTION-MINIMIZING SETS FOR LAGRANGIAN SYSTEMS

Alfonso Sorrentino

A Dissertation Presented to the Faculty of Princeton University in Candidacy for the Degree of Doctor of Philosophy

Recommended for Acceptance by the Department of Mathematics

June 2008

© Copyright by Alfonso Sorrentino, 2008.

All Rights Reserved

Abstract

In this thesis we investigate the topological properties of the action-miniziming sets that appear in the study of Tonelli Lagrangian and Hamiltonian systems.

In the first part, we will focus on understanding the topology of the quotient Aubry set, and in particular its total disconnectedness. This property, in fact, plays a key role in the variational methods developed for constructing orbits with a prescribed behavior or connecting different regions of the state space. We will show how this problem may be related to a Sard-like property for certain subsolutions of Hamilton-Jacobi equation and use this approach to show total disconnectedness under suitable assumptions on the Lagrangian.

In the second part, we will discuss some relations between the dynamics of the system and the underlying symplectic geometry of the space. In particular, we will point out how to deduce from weak KAM theory the symplectic invariance of the Aubry set and the quotient Aubry set, and we will study the action minimizing properties of invariant measures supported on Lagrangian graphs. We will then use these results to deduce uniqueness of invariant Lagrangian graphs in a fixed homology or cohomology class, with particular attention to the case of KAM tori and Herman's Tori.

Acknowledgements

I would like to express my most sincere gratitude to my advisor Prof. John Mather for his guidance and support over the last few years. I am indebted to him for all the time he spent listening to my mathematical nonsense and for his enlightening questions, which, although still unanswered, drove me toward a better understanding of this fascinating subject. I feel honored that he shared with me his peerless insight into mathematics.

Je souhaite exprimer toute ma reconnaissance au Professeur Albert Fathi pour son intérêt envers mon travail et pour ses explications, qui ont beaucoup contribué à ma formation. Je dois beaucoup à son aide et à sa bienveillance.

Я очень признателен Якову Григорьевичу Синаю, любезно согласившемуся оппонировать мне на защите диссертации. У меня остнанутся самые теплые воспоминания о семинарах, на которых он давал нам нечто большее, чем замечательную математику.

I am also grateful to Prof. Bo'az Klartag for accepting to be one of my readers, and to Prof. Elon Lindenstrauss for serving as a member on my defense committee. I apologize for not being able to write these acknowledgements in hebrew.

Come potrei mai dimenticarmi dei miei adorabili coinquilini e amici: Francesca-Fracchia, con i suoi mostri e il fedelissimo fantasma Goffredo; Martin(o) ed il suo "controcorrentismo preventivo antibuonista" (forse è virale, ahimé...); Adriano, l'ultimo ad arrivare e forse il primo a pentirsene. Grazie di cuore per aver inondato Magie 5V con il vostro affetto ed averlo trasformato in una vera "casa", dove il mio pensiero continuerà a tornare, non senza rimpianto.

Un ringraziamento particolare va ad Alessandro per molteplici motivi, non per ultimi matematici. Nonostante non mi abbia ancora pienamente convinto della sua visione C^{∞} del mondo, gli do atto di aver contribuito a migliorarlo in un "intorno" di Princeton. E per tutte le volte che mi ha dato del "pazzo": "though this be madness, yet there is method in 't" [Hamlet, Shakespeare].

Grazie anche a tutti gli altri membri della *Princeton Italian Family* (aka PIFA), che hanno animato in modo tutt'altro che marginale la mia vita princetoniana: Attilio, Corinna (e le sue (im)prevedibili "corinnate"), l'*iraconda* Elisa (ed i suoi dispetti), Francesco (*mon Dauphin*), Luca, Luca-Luha, Maria, Nicola (N..., dai te lo risparmio!), Valeria-Tittirilla e tutti quelli che in un modo o nell'altro hanno lasciato piacevoli tracce del loro passaggio.

Πολλά ευχαριστώ στη Μαρία Μιχαλογιωργάκη για την υποστήριξή της όλα τα χρόνια: Μαρία, θα ήταν πολύ δύσκολο αν δεν ήσουν εδώ.

Илья Виноградов должен сам благодарить меня, но и я в свою очередь благодарю его за интерес к моей диссертации и за помощь с этим переводом.

My special gratitude goes also to Jill LeClair and Matthew Ferszt for their constant assistance over these years and their infinite patience.

Non per ultimo, vorrei esprimere la mia riconoscenza ad Alessandra, Alessandro, Alice, Emanuela, Livia, Lorenzo, Marta, Marco e Vittoria per le molte "chiacchierate" di questi anni, che hanno virtualmente ridotto la distanza che ci separava (special thanks to Skype Technologies S.A.).

Infine, un ringraziamento di cuore alla mia famiglia per avermi sempre appoggiato e supportato (nonché sopportato) in tutte le mie scelte. Non sarei arrivato fin qui senza il loro sostegno e la loro incondizionata fiducia.

Ad Uno, Nessuno e Centomila.

Princeton, May 2008.

Or se' giunta, anima fella! (Dante, Inferno VIII, 18)

Contents

	Abs	tract	iii
	Ack	nowledgements	iv
In	trod	uction	s iv neory for Lagrangian systems 13 angians and optical Hamiltonians on compact manifolds 13 Symplectic structure of the cotangent space and Hamilto- 14 dows 21 nizing measures: Mather sets 23 Holonomic measures 31 n minimizing curves: Aubry and Mañé sets 33 Semi-static and Static curves 43 Minimizers in the non-autonomous time periodic case 44 theory 46 Regularity of critical subsolutions 57 Non-wandering points of the Mañé set 59 e of the quotient Aubry set 62
1	Ma	ther-Fathi theory for Lagrangian systems	13
	1.1	Tonelli Lagrangians and optical Hamiltonians on compact manifolds .	13
		Addendum: Symplectic structure of the cotangent space and Hamilto-	
		nian flows	21
	1.2	Action-minimizing measures: Mather sets	23
		Addendum: Holonomic measures	31
	1.3	Global action minimizing curves: Aubry and Mañé sets	33
		Addendum: Semi-static and Static curves	43
		Addendum: Minimizers in the non-autonomous time periodic case	44
	1.4	Weak KAM theory	46
		Addendum: Regularity of critical subsolutions	57
		Addendum: Non-wandering points of the Mañé set	59
2	On	the structure of the quotient Aubry set	62
	2.1	The quotient Aubry set	62
	2.2	Sard's Lemma and the quotient Aubry set	66
		Addendum: Total disconnectedness in low dimension	70

		Addendum: Counterexamples in high dimension	72		
	2.3	On the total disconnectedness in high dimension	77		
	2.4	Proof of Lemma 2.3.9 and Sard's lemma for critical subsolutions $\ . \ .$	85		
	2.5	Proof of a modified version of Kneser-Glaeser's Rough composition			
		theorem	92		
3	Dyn	namics and Symplectic geometry	109		
	3.1	Symplectic geometry of the phase space	109		
	3.2	Symplectic aspects of the Aubry set and the quotient Aubry set	114		
	3.3	Minimizing properties of Lagrangian graphs	117		
	3.4	Global uniqueness of KAM tori and Herman's tori	123		
	3.5	Schwartzman unique and strict ergodicity	128		
Bibliography 14					

Introduction

La Nature est un temple où de vivants piliers laissent parfois sortir de confuses paroles. (Charles Baudelaire, Correspondances)

It was about forty years ago when the striking, and somehow unexpected, phenomenon of "diffusion" pointed out by V. I. Arnol'd [3] (that now bears his name) shed a new light on the stable picture drawn by the works of Kolmogorov [44], Moser [68] and Arnol'd himself [2] few years earlier (from whose initials the acronym KAM). This new insight led to a change of perspectives and in order to make sense of the complex balance between stable and unstable motions that was looming out, new approaches needed to be exploited. Amongst these, variational methods turned out to be particularly successful. Mostly inspired by the so-called *least action principle*, a sort of widely accepted thriftiness of the Nature in all its actions, they provide the natural setting to get over the local view given by the analytical methods and make towards a global understanding of the dynamics.

Aubry-Mather theory represents probably one of the biggest triumphs in this direction. Developed independently by Serge Aubry [6] and John Mather [51] in the eighties, this novel approach to the study of the dynamics of twist diffeomorphisms of the annulus (which correspond to Poincaré maps of 1-dimensional Hamiltonian systems [68]) pointed out the existence of many action minimizing sets that generalize invariant rotational curves and continue to exist even after these curves disappear. Besides providing a detailed structure theory for these new sets, this powerful approach yielded to a better understanding of the destiny of invariant rotational curves and to the construction of interesting chaotic orbits as a result of their destruction [53, 38, 58].

Motivated by these achievements, John Mather [57, 59] - and later Ricardo Mañé [48, 23] and Albert Fathi [32] in different ways - developed a generalization of this theory to higher dimensional systems. Positive definite superlinear Lagrangians on compact manifolds, also called *Tonelli Lagrangians* (see Definition 1.1.1), were the appropriate setting to work in. Although it was still possible to show the existence of interesting action minimizing sets, the situation turned out to be more complicated and, even today, very little is known about their structure. This lack represents one of the biggest restraints to the potentiality of such approaches, in particular in view of proving the existence of chaotic orbits. In fact, these sets play a twofold role. Whereas on the one hand they may provide an obstruction to the existence of diffusion, on the other hand they have a fundamental importance in the variational methods developed for constructing orbits with a prescribed behavior (see for instance [11, 12, 25, 61, 63, 65, 78]). Most of these methods, in fact, are based on strong assumptions on the structure of these sets; however, understanding such properties is a very difficult task. It is in the light of this observation that our interest in the structural properties of these sets can be better explained and understood.

This thesis is organized as follows:

- In chapter 1, we will present a brief but comprehensive introduction to Mather's theory for Lagrangian systems and Fathi's weak KAM theory, providing the necessary ground for understanding the main results and techniques in the following chapters.
- In chapter 2, we will focus on understanding the topology of the quotient Aubry set.

This set plays a key role in the variational method developed by John Mather [61, 63, 65] for showing the generic existence of Arnol'd diffusion in two-and-half degrees of freedom. In order to extend this approach to higher dimensions, it would be useful to understand when this set is totally disconnected or small in some dimensional sense. In this chapter, we will address this problem and the results in [75], exploiting an existing relation with a Sard-like lemma for certain subsolutions of Hamilton-Jacobi equation.

• In chapter 3, we will discuss the relation between the dynamics of the system and the underlying symplectic geometry of the phase space. In particular, we will prove the symplectic invariance of the Aubry set and the quotient Aubry set and study the action minimizing properties of Lagrangian graphs. We will use these results to deduce some uniqueness properties [34] of invariant Lagrangian graphs in a fixed homology or cohomology class. Besides a dynamical and symplectic topological interest, these results were motivated by a question on the global uniqueness of KAM tori (and Herman's tori) with a given rotation vector.

Description of the main results.

Chapter 2: On the structure of the quotient Aubry set

The quotient Aubry set (see definition 2.1.1) and its topological properties play, as we already mentioned, an important role in the variational methods developed for the construction of orbits connecting different regions of the phase space. In particular, in the light of John Mather's studies on Arnol'd diffusion [61, 63, 65], it is important to understand when this set is totally disconnected. In fact, as we will see, one can think of it as the set of α and ω limits of global action-minimizers: each of these curves is asymptotic in the past and in the future to an element of this quotient set, also called *static class* [48, 59] (see also remark 1.3.13). From here the need for a clear understanding of its topology in order to control where the constructed orbit comes from or is heading to.

In [62] Mather showed that if the state space has dimension ≤ 2 (in the nonautonomous case) or the Lagrangian is the kinetic energy associated to a Riemannian metric and the state space has dimension ≤ 3 , then the quotient Aubry set is totally disconnected for all cohomology classes (see the addendum to section 2.2).

Unfortunately in higher dimension the situation turns out to be more complicated and in general total disconnectedness might fail. In [64], for instance, for each $d \geq 3$ and $0 < \varepsilon < 1$, John Mather constructed a $C^{2d-3,1-\varepsilon}$ mechanical Lagrangian on $T\mathbb{T}^d$ (*i.e.*, a Lagrangian which is the sum of the kinetic energy and a potential) whose associated quotient Aubry set - corresponding to the zero cohomology class - is isometric to a closed interval (see also the addendum to section 2.2 for similar counterexamples due to Albert Fathi [31]). What emerges from this construction is that the machinery breaks down when one tries to obtain a smoother Lagrangian, as if there were an intrinsic relation between the regularity of the Lagrangian, the dimension of the state space and the dimension of the quotient Aubry set.

In [75] we proved the following result.

Theorem 2.3.1. Let M be a compact connected manifold of dimension $d \ge 1$, L a Tonelli Lagrangian on TM and let $\mathcal{L} : TM \longrightarrow T^*M$ denote the Legendre transform (see (1.4)). Suppose that :

i) $\Lambda_L := \mathcal{L}(M \times \{0\})$ is a C^2 Lagrangian submanifold of T^*M , with the canonical symplectic form, and let $c_L \in H^1(M; \mathbb{R})$ be its Liouville (or cohomology) class.

ii)
$$L(\cdot, 0) \in C^{r}(M)$$
, with $r \geq 2d - 2$.

Then, the quotient Aubry set $(\bar{\mathcal{A}}_{c_L}, \bar{\delta}_{c_L})$ is totally disconnected.

It is worthy mentioning that a similar result has been also proven independently

by Albert Fathi, Alessio Figalli and Ludovic Rifford ([33], in preparation). Observe that this result clearly applies to mechanical Lagrangians and, more generally, to symmetric (or reversible) Lagrangians (*i.e.*, L(x, v) = L(x, -v) for all $(x, v) \in TM$).

Corollary 2.3.4. Let M be a compact connected manifold of dimension $d \ge 1$ and let L(x, v) be a symmetrical Tonelli Lagrangian on TM, such that $L(x, 0) \in C^r(M)$, with $r \ge 2d - 2$. Then, the quotient Aubry set $(\bar{A}_0, \bar{\delta}_0)$ is totally disconnected.

More specifically,

Corollary 2.3.5. Let M be a compact connected manifold of dimension $d \ge 1$ and let $L(x,v) = \frac{1}{2} ||v||_x^2 + U(x)$ be a mechanical Lagrangian on TM, such that the potential $U \in C^r(M)$, with $r \ge 2d - 2$. Then, the quotient Aubry set $(\bar{\mathcal{A}}_0, \bar{\delta}_0)$ is totally disconnected.

In these cases, in fact, Λ_L is the zero section of T^*M and $c_L = 0$. Moreover, because of Mather's counterexamples [64], the regularity condition is optimal (one can easily modify the proof of the theorem to include also the case $L(\cdot, 0) \in C^{2d-3,1}(M)$).

Observe that this result also applies to Mañé's Lagrangians (see section 1.1 for a definition) associated to *irrotational* vector fields.

Corollary 2.3.7. Let M be a compact connected manifold of dimension $d \ge 1$, equipped with a C^{∞} Riemannian metric g. Let X be a C^r vector field on M and consider the associated Mañé's Lagragian $L_X(x,v) = \frac{1}{2} ||v - X(x)||_x^2$. If $r \ge 2d - 2$ then the quotient Aubry set $(\bar{\mathcal{A}}_{c_{L_X}}, \bar{\delta}_{c_{L_X}})$ is totally disconnected, where $X^{\flat} = g(X, \cdot)$ is the associated 1-form and $c_{L_X} = [X^{\flat}] \in \mathrm{H}^1(M; \mathbb{R})$.

The main idea behind the proof of theorem 2.3.1 is that there exists a *liaison* between this problem and Sard's Lemma and this relation becomes more clear if one considers the Weak KAM theory setting (see section 1.4). Such an approach is based on the concept of *critical subsolutions* and *weak solutions* of Hamilton-Jacobi

equation and consider the Hamiltonian setting, rather than the Lagrangian one. From a symplectic geometric point of view, it can be equivalently interpreted as the study of particular Lagrangian graphs and their non-removable intersections (see [69] and chapter 3).

Critical subsolutions (see definition 1.4.6) are a particular class of subsolutions that correspond to what is called the *Mañé's critical energy level*. They carry important information about the dynamics of the system and allow one to recover most of the results previously obtained via Mather and Mañé's approaches. It is a nontrivial fact that such Lipschitz subsolutions exist; moreover, Albert Fathi and Antonio Siconolfi [37] showed that C^1 critical subsolutions do also exist and are "dense" (see theorem 1.4.24). Patrick Bernard [13] extended this result to $C^{1,1}$ critical subsolutions.

The relation with Sard's Lemma that we were mentioning above can be easily expressed in terms of a Sard-like property for differences of these critical subsolutions. Let us denote this set by $\mathcal{D}_c^{1,1}$ the set of differences of $C^{1,1}$ *c*-critical subsolutions (see (2.1)). In section 2.2 we showed:

Proposition 2.2.6. Let M be a compact connected manifold of dimension $d \ge 1$, L a Tonelli Lagrangian and $c \in H^1(M; \mathbb{R})$. If each $w \in \mathcal{D}_c^{1,1}$ is of Morse-Sard type, then the quotient Aubry set $(\bar{\mathcal{A}}_c, \bar{\delta}_c)$ is totally disconnected.

Observe that this lemma immediately implies Mather's result in dimension $d \leq 2$ (for the autonomous case). In fact, in dimension $d \leq 2$ Sard's lemma holds for $C^{1,1}$ functions [10].

As a result of this, the original problem can be rephrased as follows: under which conditions on L and c, are these differences of subsolutions (or a "dense" subset of it) of Sard type? Unfortunately, this is not so easy to tackle either. It is not possible to use directly the classical Sard's lemma (or one of the available versions), due to a lack of regularity of these critical subsolutions, since in general they will be at most $C^{1,1}$. In fact, although it is always possible to smooth them out of the Aubry set and obtain functions in $C^{\infty}(M \setminus \mathcal{A}_c) \cap C^{1,1}(M)$, the presence of the Aubry set (where the value of their differential is prescribed) represents an obstacle that in some cases becomes impossible to overcome (see remark 1.4.25). On the other hand, one could try to control the *complexity* of these functions near their critical value set, using the rigid structure provided by Hamilton-Jacobi equation and the smoothness of the Hamiltonian itself, rather than the regularity of the subsolutions. While there are several difficulties in pursuing this approach in the general case - mostly related to the nature of the Aubry set - under the hypotheses of theorem 2.3.1 we managed to prove the following result, which can be seen a version of Sard's lemma for critical subsolutions of certain Hamilton-Jacobi equations:

Proposition 2.3.10. Under the hypotheses of the theorem 2.3.1, if u is a η_L critical subsolution, then $|u(\mathcal{A}_{c_L})| = 0$. In particular, for any $w \in \mathcal{D}_{c_L}$ we have that $|w(\mathcal{A}_{c_L})| = 0$.

Chapter 3: Dynamics and Symplectic geometry

In this chapter we want to highlight some relations between the dynamics of the system and the underlying symplectic geometry of the phase space. We start in section 3.2, by discussing purely symplectic definitions of the Aubry set and the quotient Aubry set, which follow quite easily from weak KAM theory approach. In fact, one can reinterpret the notion of critical subsolution from a more geometric perspective, and introduce what we called *c*-subcritical Lagrangian graphs (see definition 3.2.1). It will follow immediately from weak KAM theory (in particular theorem 1.4.19), that for any given cohomology class $c \in H^1(M; \mathbb{R})$ the Aubry set \mathcal{A}_c^* can be obtained as the intersection of all these Lagrangian graphs (see also [69]). In a similar way we will get a symplectic characterization for Mather's pseudodistance δ_c , the quotient

Aubry set $\overline{\mathcal{A}}_c$ and the Mañé critical energy level \mathcal{E}_c^* . Using these characterizations, one can deduce the following result concerning the invariance of these sets under exact symplectomorphisms. Let us first recall that a diffeomorphism $\Psi : T^*M \longrightarrow T^*M$ is a *symplectomorphism* if it preserves the symplectic form ω , *i.e.*, $\Psi_*\omega = \omega$; in particular this is equivalent to say that $\Psi_*\lambda - \lambda$ is a closed 1-form (where λ denotes the *Liouville* form and $\omega = -d\lambda$; see section 3.1 and the addendum to section 1.1). We will say that Ψ is an *exact symplectomorphism* if $\Psi_*\lambda - \lambda$ is exact.

Theorem 3.2.3. Let H be an optical Hamiltonian on T^*M and let $\Psi : T^*M \longrightarrow T^*M$ be an exact symplectomorphism. Consider the new Hamiltonian $H' = H \circ \Psi^{-1}$. Then for all $c \in H^1(M; \mathbb{R})$: ¹

- $\mathcal{E}_c^{*\prime} = \Psi(\mathcal{E}_c^*)$ and therefore $\alpha'(c) = \alpha(c)$;
- $\mathcal{A}_c^{*\prime} = \Psi(\mathcal{A}_c^*);$
- δ_c(x, y) = δ'_c(Ψ(x), Ψ(y)); therefore it maps c-static classes of H' into c-static classes of H and the induced map Ψ : Ā_c → Ā'_c is an isometry.

A similar result has been also proven, with a different approach, by Patrick Bernard in [15].

In section 3.3 we will analyze the minimizing properties of invariant measures supported on Lagrangian graphs and use them to deduce some uniqueness results for invariant Lagrangian graphs within a fixed homology or cohomology class. These results are based on a joint work [34] with Albert Fathi and Alessandro Giuliani, originally motivated by a question on the global uniqueness of KAM tori with a given rotation vector.

We start with the following characterization of minimizing measures (or *Mather's measures*, see section 1.2):

¹We will indicate with a *prime* all quantities associated to H' (*e.g.*, $\mathcal{E}_c^{*'}, \mathcal{A}', \delta_c'$, etc...)

Lemma 3.3.1. Let μ be an invariant probability measure on TM and $\mu^* = \mathcal{L}_*\mu$ its push-forward to T*M, via the Legendre transform \mathcal{L} . Then, μ is a Mather's measure if and only if $\operatorname{supp} \mu^*$ is contained in the critical part of a subcritical Lagrangian graph. In particular, any invariant probability measure μ^* on T*M, whose support is contained in an invariant Lagrangian graph with Liouville class c, is the image of a c-action minimizing measure, via the Legendre transform.

Note that the fact that the orbits on an invariant Lagrangian graph are actionminimizing can be also deduced from a classical result of calculus of variations due to Weierstrass, as already pointed out by Jürgen Moser (see remark in [57]). In fact, Weierstrass method or the use of the Hamilton-Jacobi Equation (that we are using in our proof) are essentially two sides of the same coin.

This result easily implies an already-known uniqueness result for Lagrangian graphs supporting invariant measures of full support, in a fixed cohomology class (see also [57], in which a different proof is presented).

Theorem 3.3.3. If $\Lambda \subset T^*M$ is a Lagrangian graph on which the Hamiltonian dynamics admits an invariant measure μ^* with full support, then $\Lambda = \mathcal{L}(\widetilde{\mathcal{M}}_c) = \mathcal{A}_c^*$, where c is the cohomology class of Λ . Therefore, if Λ_1 and Λ_2 are two Lagrangian graphs as above, with the same cohomology class, then $\Lambda_1 = \Lambda_2$. In other words, for any given $c \in H^1(M; \mathbb{R})$, there exists at most one invariant Lagrangian graph Λ with cohomology class c, that carries an invariant measure whose support is the whole of Λ .

Moreover,

Theorem 3.3.4. If Λ and μ are as in Theorem 3.3.3 and ρ is the rotation vector of $\mu = \mathcal{L}^{-1}\mu^*$, then $\Lambda = \mathcal{L}(\widetilde{\mathcal{M}}^{\rho})$. Therefore, if Λ_1 and Λ_2 are two Lagrangian graphs supporting measures of full support and the same rotation vector ρ , then $\Lambda_1 = \Lambda_2$. Moreover, Mather's β -function is differentiable at ρ with $\partial\beta(\rho) = c$, where c is the

cohomology class of Λ .

Using the notion of rotation vector or Schwartzman asymptotic cycle of an invariant measure, which describes how asymptotically an average orbit winds around TM (see sections 1.2, 3.5 and [57, 74]), we introduce the notion of Schwartzman ergodicity. We shall say that a Lagrangian graph Λ is Schwartzman uniquely ergodic if all invariant measures supported on Λ have the same rotation vector ρ , which will be called homology class of Λ (see definition 3.5.5). Moreover, if there exists an invariant measure with full support, Λ will be called Schwarztman strictly ergodic (see definition 3.5.9). We will give a detailed description of these flows and their properties in section 3.5.

We will prove the following uniqueness result.

Theorem 3.3.6. Let Λ be a Schwartzman strictly ergodic invariant Lagrangian graph with homology class ρ . The following properties are satisfied:

- (i) if $\Lambda \cap \mathcal{A}_c^* \neq \emptyset$, then $\Lambda = \mathcal{A}_c^*$ and $c = c_\Lambda$, where c_Λ is the cohomology class of Λ .
- (ii) the Mather function α is differentiable at c_{Λ} and $\partial \alpha(c_{\Lambda}) = \rho$.

Therefore,

- (iii) any invariant Lagrangian graph that carries a measure with rotation vector ρ is equal to the graph Λ;
- (iv) any invariant Lagrangian graph is either disjoint from Λ or equal to Λ .

If $M = \mathbb{T}^d$, it is natural to ask what this implies in the setting of KAM theory. In section 3.4 we will use these results to discuss the problem of uniqueness of KAM tori (see definition 3.4.1) and, more generally, of the invariant tori belonging to the closure of the set of KAM tori (that we will call *Herman's tori*, [43]). In fact, while the proof of KAM theorem is constructive and the invariant torus one finds is locally unique (see for instance [72, 73]), the issue of global uniqueness of such tori is still an object of some debate and study (see, for instance, [17]).

Corollary 3.4.6. Every optical Hamiltonian H on T^*T^d possesses at most one Lagrangian KAM torus for any given rotation vector ρ .

The property of being Lagrangian plays a crucial role. When ρ is rationally independent (i.e., $\langle \rho, \nu \rangle \neq 0$, $\forall \nu \in \mathbb{Z}^d \setminus \{0\}$), every KAM torus with frequency ρ is automatically Lagrangian (this is a remark due to Michael Herman; see proposition 3.1.7).

We also extend this uniqueness result to generic invariant tori contained in the C^0 -closure $\overline{\Upsilon}$ of the set Υ of all Lagrangian KAM tori. These "new" tori were first studied by Michael Herman [43], who showed that generically they are not conjugated to rotations. Moreover, they represent the majority, in the sense of topology, and hence most invariant tori cannot be obtained by the KAM algorithm. More precisely, Herman showed that in $\overline{\Upsilon}$ there exists a dense G_{δ} set (*i.e.*, a dense countable intersection of open sets) of invariant Lagrangian graphs on which the dynamics is strictly ergodic and weakly mixing, and for which the rotation vector is not Diophantine. These invariant graphs are therefore not obtained by the KAM theorem, however our uniqueness result do still apply to these graphs since strict ergodicity implies Schwartzman strict ergodicity.

More generally, given any Tonelli Lagrangian on \mathbb{T}^d , we consider the set $\tilde{\Upsilon}$ of invariant Lagrangian graphs on which the dynamics of the flow is topologically conjugated to an *ergodic* linear flow on \mathbb{T}^d . We show the following proposition.

Proposition 3.4.7. There exists a dense G_{δ} set \mathcal{G} in the C^0 closure of $\tilde{\Upsilon}$ consisting of strictly ergodic invariant Lagrangian graphs. Any $\Lambda \in \mathcal{G}$ satisfies the following properties:

(i) the invariant graph Λ has a well-defined rotation vector $\rho(\Lambda)$.

- (ii) Any invariant Lagrangian graph that intersects Λ coincides with Λ .
- (iii) Any Lagrangian invariant graph that carries an invariant measure whose rotation is ρ(Λ) coincides with Λ.

Chapter 1

Mather-Fathi theory for Lagrangian systems

1.1 Tonelli Lagrangians and optical Hamiltonians on compact manifolds

In this section we want to introduce the basic setting that we will be considering hereafter. Let M be a compact and connected smooth manifold without boundary. Denote by TM its tangent bundle and T^*M the cotangent one. A point of TM will be denoted by (x, v), where $x \in M$ and $v \in T_xM$, and a point of T^*M by (x, p), where $p \in T^*_xM$ is a linear form on the vector space T_xM . Let us fix a Riemannian metric g on it and denote by d the induced metric on M; let $\|\cdot\|_x$ be the norm induced by g on T_xM ; we will use the same notation for the norm induced on T^*_xM .

We will consider functions $L : TM \longrightarrow \mathbb{R}$ of class C^2 , which are called *Lagrangians*. Associated to each Lagrangian, there is a flow on TM called the *Euler-Lagrange* flow, defined as follows. Let us consider the action functional A_L from the space of continuous piecewise C^1 curves $\gamma : [a, b] \to M$, with $a \leq b$, defined by:

$$A_L(\gamma) := \int_a^b L(\gamma(t), \dot{\gamma}(t)) \, dt.$$

Curves that extremize this functional among all curves with the same end-points are solutions of the *Euler-Lagrange equation*:

$$\frac{d}{dt}\frac{\partial L}{\partial v}(\gamma(t),\dot{\gamma}(t)) = \frac{\partial L}{\partial x}(\gamma(t),\dot{\gamma}(t)) \qquad \forall t \in [a,b].$$
(1.1)

Observe that this equation is equivalent to

$$\frac{\partial^2 L}{\partial v^2}(\gamma(t),\dot{\gamma}(t))\ddot{\gamma}(t) = \frac{\partial L}{\partial x}(\gamma(t),\dot{\gamma}(t)) - \frac{\partial^2 L}{\partial v \partial x}(\gamma(t),\dot{\gamma}(t))\dot{\gamma}(t) \,,$$

therefore, if the second partial vertial derivative $\partial^2 L/\partial v^2(x, v)$ is non-degenerate at all points of TM, we can solve for $\ddot{\gamma}(t)$. This condition

$$\det \frac{\partial^2 L}{\partial v^2} \neq 0$$

is called *Legendre condition* and allows one to define a vector field X_L on TM, such that the solutions of $\ddot{\gamma}(t) = X_L(\gamma(t), \dot{\gamma}(t))$ are precisely the curves satisfying the Euler-Lagrange equation. This vector field X_L is called the *Euler-Lagrange vector* field and its flow Φ_t^L is the *Euler-Lagrange flow* associated to L. It turns out that Φ_t^L is C^1 even if L is only C^2 .

Definition 1.1.1 (Tonelli Lagrangian). A function $L : TM \longrightarrow \mathbb{R}$ is called a Tonelli Lagrangian if:

- i) $L \in C^2(TM);$
- ii) L is strictly convex in the fibers, in the C^2 sense, i.e., the second partial vertical derivative $\partial^2 L/\partial v^2(x, v)$ is positive definite, as a quadratic form, for all (x, v);

iii) L is superlinear in each fiber, i.e.,

$$\lim_{\|v\|_x \to +\infty} \frac{L(x,v)}{\|v\|_x} = +\infty;$$

this condition is equivalent to ask that for each $A \in \mathbb{R}$ there exists $B(A) \in \mathbb{R}$ such that

$$L(x,v) \ge A \|v\| - B(A) \qquad \forall (x,v) \in TM.$$

Observe that since all Riemannian metrics on a compact manifold are equivalent, then condition iii is independent of the choice of the Riemannian metric g.

Remark 1.1.2. More generally, one can consider the case of a *time-periodic Tonelli* Lagrangian $L: TM \times \mathbb{T} \longrightarrow \mathbb{R}$ (also called *non-autonomous case*), as it was originally done by John Mather [57]. As it was pointed out to him by Jürgen Moser, this was the right setting to generalize Aubry and Mather's results for twist maps to higher dimensions; in fact, every twist map can be seen as the time one map associated to the flow of a periodic Tonelli Lagrangian on the one dimensional torus (see for instance [68]). In this case, a further condition on the Lagrangian is needed:

iv) The Euler-Lagrange flow is complete, i.e., every maximal integral curve of the vector field X_L has all \mathbb{R} as its domain of definition.

In the non-autonomous case, in fact, this condition is necessary in order to have that action-minimizing curves (or *Tonelli minimizers*, see section 1.3) satisfy the Euler-Lagrange equation. Without such an assumption Ball and Mizel [7] have constructed example of Tonelli minimizers that are not C^1 and therefore are not solutions of the Euler-Lagrange flow. The role of the completeness hypothesis can be explained as follows. It is possible to prove, under the above conditions, that action minimizing curves not only exist and are absolutely continuous, but they are C^1 on an open and dense full measure subset of the interval in which they are defined; it is easy to check that they satisfy the Euler-Lagrange equation on this set, while their velocity goes to infinity on the exceptional set on which they are not C^1 . Asking the flow to be complete, therefore, implies that Tonelli minimizers are C^1 everywhere and are actual solutions of the Euler-Lagrange equation. A sufficient condition for the completeness of the Euler-Lagrange flow, for example, can be expressed in terms of a growth condition for $\partial L/\partial t$:

$$-\frac{\partial L}{\partial t}(x,v,t) \le C\left(1 + \frac{\partial L}{\partial v}(x,v,t) \cdot v - L(x,v,t)\right) \quad \forall (x,v,t) \in \mathbf{T}M \times \mathbb{T}.$$

Examples of Tonelli Lagrangians.

• Riemannian Lagrangians. Given a Riemannian metric g on TM, the Riemannian Lagrangian on (M, g) is given by the Kinetic energy:

$$L(x,v) = \frac{1}{2} \|v\|_x^2.$$

Its Euler-Lagrange equation is the equation of the geodesics of g:

$$\frac{D}{dt}\dot{x} \equiv 0 \,,$$

and its Euler-Lagrange flow coincides with the geodesic flow.

• Mechanical Lagrangians. These Lagrangians play a key-role in the study of classical mechanics. They are given by the sum of the kinetic energy and a potential $U: M \longrightarrow \mathbb{R}$:

$$L(x,v) = \frac{1}{2} \|v\|_x^2 + U(x).$$

The associated Euler-Lagrange equation is given by:

$$\frac{D}{dt}\dot{x} = \nabla U(x) \,,$$

where ∇U is the gradient of U with respect to the riemannian metric g, *i.e.*,

$$d_x U \cdot v = \langle \nabla U(x), v \rangle_x \quad \forall (x, v) \in \mathrm{T}M.$$

 Mañé's Lagrangians. This is a particular class of Tonelli Lagrangians, introduced by Ricardo Mañé in [46] (see also [33]). If X is a C^k vector field on M, with k ≥ 2, one can embed its flow φ^X_t into the Euler-Lagrange flow associated to a certain Lagrangian, namely

$$L_X(x,v) = \frac{1}{2} \|v - X(x)\|_x^2$$

It is quite easy to check that the integral curves of the vector field X are solutions to the Euler-Lagrange equation. In particular, the Euler-Lagrange flow $\Phi_t^{L_X}$ restricted to $\operatorname{Graph}(X) = \{(x, X(x)), x \in M\}$ (that is clearly invariant) is conjugated to the flow of X on M and the conjugation is given by $\pi |\operatorname{Graph}(X)$, where $\pi : \operatorname{T} M \to M$ is the canonical projection. In other words, the following diagram commutes:

$$\begin{array}{ccc} \operatorname{Graph}(X) & \xrightarrow{\Phi_t^{L_X}} \operatorname{Graph}(X) \\ & \pi & & & & \\ \pi & & & & & \\ M & \xrightarrow{\varphi_t^X} & M \end{array}$$

that is, for every $x \in M$ and every $t \in \mathbb{R}$, $\Phi_t^{L_X}(x, X(x)) = (\gamma_x^X(t), \dot{\gamma}_x^X(t))$, where $\gamma_x^X(t) = \varphi_t^X(x)$.

In the study of classical dynamics, it turns often very useful to consider the as-

sociated Hamiltonian system, which is defined on the cotangent space T^*M . Let us describe how to define this new system and what is its relation with the Lagrangian one.

A standard tool in the study of convex functions is the so-called *Fenchel transform*, which allows one to transform functions on a vector space into functions on the dual space (see for instance [32, 71] for excellent introductions to the topics). Given a Lagrangian L, we can define the associated *Hamiltonian*, as its Fenchel transform (or *Fenchel-Legendre transfrom*):

$$H: \mathbf{T}^*M \longrightarrow \mathbb{R}$$
$$(x,p) \longmapsto \sup_{v \in \mathbf{T}_x M} \{ \langle p, v \rangle_x - L(x,v) \}$$

where $\langle \cdot, \cdot \rangle_x$ denotes the canonical pairing between the tangent and cotangent space.

If L is a Tonelli Lagrangian, one can easily prove that H is finite everywhere (as a consequence of the superlinearity of L), C^2 , superlinear and strictly convex in each fiber (in the C^2 sense). Such a Hamiltonian is called *optical* (or *Tonelli*) *Hamiltonian*.

Definition 1.1.3 (Optical Hamiltonian). A function $H : T^*M \longrightarrow \mathbb{R}$ is called an optical (or Tonelli) Hamiltonian if:

- i) H is of class C^2 ;
- ii) H is strictly convex in each fiber in the C² sense, i.e., the second partial vertical derivative ∂²H/∂p²(x, p) is positive definite, as a quadratic form, for any (x, p) ∈ T*M;
- iii) H is superlinear in each fiber, i.e.,

$$\lim_{\|p\|_x \to +\infty} \frac{H(x,p)}{\|p\|_x} = +\infty \,.$$

Examples of optical Hamiltonians.

Let us see what are the optical Hamiltonians associated to the Tonelli Lagrangians that we have introduced in the previous examples.

• Riemannian Hamiltonians. If $L(x, v) = \frac{1}{2} ||v||_x^2$ is the Riemannian Lagrangian associated to a Riemannian metric g on M, the corresponding Hamiltonian will be

$$H(x,p) = \frac{1}{2} \|p\|_x^2,$$

where $\|\cdot\|$ represents - in this last expression - the induced norm on the cotangent space T^*M .

• Mechanical Hamiltonians. If $L(x, v) = \frac{1}{2} ||v||_x^2 + U(x)$ is a mechanical Lagrangian, the associated Hamiltonian is:

$$H(x,p) = \frac{1}{2} \|p\|_x^2 - U(x),$$

that it is sometimes referred to as *mechanical energy*.

• Mañé's Hamiltonians. If X is a C^k vector field on M, with $k \ge 2$, and $L_X(x,v) = ||v - X(x)||_x^2$ is the associated Mañé Lagrangian, one can check that the corresponding Hamiltonian is given by:

$$H(x,p) = \frac{1}{2} ||p||_x^2 + \langle p, X(x) \rangle$$

Given a Hamiltionian one can consider the associated Hamiltonian flow Φ_t^H on T^*M . In local coordinates, this flow can be expressed in terms of the so-called

Hamilton's equations:

$$\begin{cases} \dot{x}(t) = \frac{\partial H}{\partial p}(x(t), p(t)) \\ \dot{p}(t) = -\frac{\partial H}{\partial x}(x(t), p(t)) . \end{cases}$$
(1.2)

It is easy to check that - only in the autonomous case - the Hamiltonian is a *prime integral of the motion*, *i.e.*, it is constant along the solutions of these equations.

Now, we would like to explain what is the relation between the Euler-Lagrange flow and the Hamiltonian one. It follows easily from the definition of Hamiltonian (and Fenchel transform) that for each $(x, v) \in TM$ and $(x, p) \in T^*M$ the following inequality holds:

$$\langle p, v \rangle_x \le L(x, v) + H(x, p); \tag{1.3}$$

this is called *Fenchel inequality* and plays a crucial role in the study of Lagrangian and Hamiltonian dynamics and, in particular, the variational methods that we are going to describe. In particular, equality holds if and only if $p = \partial L/\partial v(x, v)$. One can therefore introduce the following diffeomorphism between TM and T^{*}M:

$$\mathcal{L}: \ \mathrm{T}M \longrightarrow \mathrm{T}^*M$$
$$(x,v) \longmapsto \left(x, \frac{\partial L}{\partial v}(x,v)\right).$$
(1.4)

This is called the *Legendre transform* and the following relation with the Hamiltonian holds:

$$H \circ \mathcal{L}(x, v) = \left\langle \frac{\partial L}{\partial v}(x, v), v \right\rangle_{x} - L(x, v).$$

A crucial observation is that this diffeomorphism represents a conjugation between the two flows, namely the Euler-Lagrange flow on TM and the Hamiltonian one on T^*M ; in other words, the following diagram commutes:

$$\begin{array}{c|c} \mathrm{T}M & \xrightarrow{\Phi_t^L} \mathrm{T}M \\ \mathcal{L} & & & \downarrow \mathcal{L} \\ \mathrm{T}^*M & \xrightarrow{\Phi_t^H} \mathrm{T}^*M \end{array}$$

Therefore one can equivalently study the Euler-Lagrange flow or the Hamiltonian flow, obtaining in both cases information on the dynamics of the system. Each of these equivalent approaches will provide different tools and advantages, which may prove very useful to understand the dynamical properties of the system. For instance, the tangent space is the natural setting for the classical calculus of variations and for Mather and Mañé's approaches (sections 1.2 and 1.3); on the other hand, the cotangent space can be equipped with a canonical symplectic structure (see the addendum at the end of this section), which allows one to use many fine symplectic topological results, coming from the study of Lagrangian graphs, Hofer's theory, Floer Homology, *etc* ... Moreover, a particular fruitful approach in T^*M is the so-called *Hamilton-Jacobi method* (or *Weak KAM theory*), which concerns with the study of *solutions* and *subsolutions* of Hamilton-Jacobi equations and, in a certain sense, represents the functional analytical counterpart of the above-mentioned variational approach (section 1.4). In the following sections we will provide a complete description of these methods and their implications to the study of the dynamics of the system.

ADDENDUM

Symplectic structure of the cotangent space and Hamiltonian flows

There is a more geometric and intrinsic way to define the Hamiltonian flow. Remember that T^*M , as a cotangent bundle, may be equipped with a *canonical* symplectic structure. Namely, if $(\mathcal{U}, x_1, \ldots, x_d)$ is a local coordinate chart for M and $(T^*\mathcal{U}, x_1, \ldots, x_d, p_1, \ldots, p_d)$ the associated cotangent coordinates, one can define the 2-form

$$\omega = \sum_{i=1}^d dx_i \wedge dp_i \,.$$

It is easy to show that ω is a symplectic form (*i.e.*, it is non-degenerate and closed). In particular, one can check that ω is intrinsically defined (*i.e.*, it does not depend on the choice of coordinate charts), by considering the 1-form on T^{*} \mathcal{U}

$$\lambda = \sum_{i=1}^d p_i \, dx_i \,,$$

which satisfies $\omega = -d\lambda$ and is coordinate-independent; in fact, in terms of the natural projection

$$\begin{array}{rccc} \pi: \mathrm{T}^*M & \longrightarrow & M \\ & (x,p) & \longmapsto & x \end{array}$$

the form λ may be equivalently defined pointwise by

$$\lambda_{(x,p)} = (d\pi_{(x,p)})^* p \in \mathcal{T}^*_{(x,p)}\mathcal{T}^*M.$$

The 1-form λ is called the *Liouville form* (or the *tautological form*).

Since ω is non-degenerate and closed, the following relation determines a unique vector field X_H on T^*M :

$$\omega\left(X_H(x,p),\cdot\right) = d_x H(\cdot)\,.$$

This vector field is the *Hamiltonian vector field* and one can easily check that, in local coordinates, it coincides with the one given by Hamilton's equation (1.2). We will discuss more in depth the relation between the dynamics of the system and the symplectic properties of the underlying space in chapter 3.

1.2 Action-minimizing measures: Mather sets

This and the following sections are meant to provide a comprehensive introduction to Mather's theory for Lagrangian systems and Fathi's weak KAM theory. We will recall most of the results that we are going to use, trying to give - unless where it is needed - general ideas rather than rigourous proofs (for which we refer to [32, 57] and references therein).

Remark 1.2.1. Before entering into the details of Mather's theory, let us make a crucial remark, which is at the base of this approach. Observe that if η is a 1-form on M, we can see it as a function on the tangent space (linear on each fiber)

$$\begin{array}{rcl} \hat{\eta}:\mathrm{T}M & \longrightarrow & \mathbb{R} \\ & & (x,v) & \longmapsto & \langle \eta(x),\,v\rangle_x \end{array}$$

and consider a new Tonelli Lagrangian $L_{\eta} := L - \hat{\eta}$; the associated Hamiltonian will be $H_{\eta}(x,p) = H(x,\eta(x)+p)$. Moreover, if η is closed, then $\int L dt$ and $\int L_{\eta} dt$ will have the same extremals and therefore the Euler-Lagrange flows on TM associated to L and L_{η} will be the same. This last conclusion may be easily deduced observing that, since η is closed, the variational equations $\delta [\int L dt] = 0$ and $\delta [\int (L - \hat{\eta}) dt] = 0$ for the fixed end-point problem have the same solutions. Although the extremals are the same, this is not generally true for the orbits "minimizing the action" (we will give a precise definition of "minimizers" later in section 1.3). What one can say is that they stay the same when we change the Lagrangian by an exact 1-form. Thus, for a fixed L, the minimizers will depend only on the de Rham cohomology class $c = [\eta] \in H^1(M; \mathbb{R})$. Therefore, instead of studying the action minimizing properties of a single Lagrangian, one can consider a family of such "modified" Lagrangians, parameterized over $H^1(M;\mathbb{R})$. This idea represents the keystone of the approach that we are going to describe.

In order to generalize to more degrees of freedom Aubry and Mather's variational approach to twist maps, a first important notion is that of *minimal measure*, which replaces that of action minimizing orbit. Aubry-Mather theory in higher dimension, in fact, cannot deal with such orbits, due to a lack of them: a classical example due to Hedlund [40] shows the existence of a Riemannian metric on a three-dimensional torus, for which minimal geodesics exist only in three directions. Instead, Mather proposed to look at the closely related notion of action minimizing invariant probability measures. Let us try to describe this idea. Let $\mathfrak{M}(L)$ be the space of probability measures on TM that are invariant under the Euler-Lagrange flow of L. To each $\mu \in \mathfrak{M}(L)$, we may associate its *average action*

$$A_L(\mu) = \int_{\mathrm{T}M} L \, d\mu$$

Since L is bounded below (because of the superlinear growth condition), this integral exists although it might be $+\infty$. In [57] Mather showed the existence of $\mu \in \mathfrak{M}(L)$ such that $A_L(\mu) < +\infty$. The argument is mainly the same as Krylov-Bogoliubov's theorem concerning existence of invariant measures for flows on compact spaces. This argument is applied to a one-point compactification of TM, and the main step consists in showing that the measure provided by this construction has no atomic part supported at ∞ (which is a fixed point for the extended system).

Remark 1.2.2. Note that Mather's approach works also for periodic time-dependent Lagrangians. For time independent Lagrangians, finding such a μ is much easier. By conservation of energy, the levels of the energy function are compact and invariant under the Euler-Lagrange flow Φ_t^L , and therefore carry such measures.

In case $A_L(\mu) < \infty$, thanks to the superlinearity of L, the integral $\int_{TM} \hat{\eta} d\mu$ is well defined and finite for any closed 1-form η on M (see [57]). Moreover, it is quite easy to show (again see [57]) that since μ is invariant by the Euler-Lagrangian flow Φ_t^L , if $\eta = df$ is an exact 1-form, then $\int d\hat{f} d\mu = 0$. Therefore, we can define a linear functional

$$\begin{array}{rccc} \mathrm{H}^{1}(M;\mathbb{R}) & \longrightarrow & \mathbb{R} \\ \\ c & \longmapsto & \int_{\mathrm{T}M} \hat{\eta} d\mu \, , \end{array}$$

where η is any closed 1-form on M with cohomology class c. By duality, there exists $\rho(\mu) \in H_1(M; \mathbb{R})$ such that

$$\int_{\mathrm{T}M} \hat{\eta} \, d\mu = \langle c, \rho(\mu) \rangle \qquad \forall \, c \in \mathrm{H}^1(M; \mathbb{R})$$

(the bracket on the right-hand side denotes the canonical pairing between cohomology and homology). We call $\rho(\mu)$ the rotation vector of μ . It is the same as the Schwartzman's asymptotic cycle of μ (see section 3.5 and [74] for more details). A natural question is whether there exist invariant probability measures for any given rotation vector. The answer turns out to be affirmative. In fact, using that the action functional $A_L : \mathfrak{M}(L) \longrightarrow \mathbb{R} \cup \{+\infty\}$ is lower semicontinuous, one can prove the following:

Proposition 1.2.3. [57]. For every $h \in H_1(M; \mathbb{R})$ there exists $\mu \in \mathfrak{M}(L)$ with $A_L(\mu) < \infty$ and $\rho(\mu) = h$. In other words, the map $\rho : \mathfrak{M}(L) \longrightarrow H_1(M; \mathbb{R})$ is surjective.

Amongst the all probability measures with a prescribed rotation vector, a peculiar role - from a dynamical systems point of view - will be played by those minimizing the average action. Following Mather, let us consider the minimal value of the average action A_L over the probability measures with rotation vector h:

$$\beta : \mathrm{H}_{1}(M; \mathbb{R}) \longrightarrow \mathbb{R}$$
$$h \longmapsto \min_{\mu \in \mathfrak{M}(L): \, \rho(\mu) = h} A_{L}(\mu) \,. \tag{1.5}$$

Observe that this minimum is actually achieved (see [57]). This function β is what is generally known as *Mather's* β -function and it is also related to the notion of *stable* norm for a metric d (see for instance [49]).

We can now define what we mean by action minimizing measure with a given rotation vector.

Definition 1.2.4. A measure $\mu \in \mathfrak{M}(L)$ realizing the minimum in (1.5), i.e., such that $A_L(\mu) = \beta(\rho(\mu))$, is called an action minimizing (or minimal or Mather's) measure with rotation vector $\rho(\mu)$.

Dual to the concept of rotation vector, one can introduce the notion of *cohomology* of an action minimizing measure. Let us try to describe what we mean. Since the β -function is convex, one can consider its *conjugate* function (given by Fenchel's duality):

$$\alpha : \mathrm{H}^{1}(M; \mathbb{R}) \longrightarrow \mathbb{R}$$

$$c \longmapsto \max_{h \in \mathrm{H}_{1}(M; \mathbb{R})} \left(\langle c, h \rangle - \beta(h) \right). \tag{1.6}$$

In particular,

$$\alpha(c) := \max_{h \in \mathrm{H}_{1}(M;\mathbb{R})} \left(\langle c, h \rangle - \beta(h) \right) =$$

$$= -\min_{h \in \mathrm{H}_{1}(M;\mathbb{R})} \left(\beta(h) - \langle c, h \rangle \right) =$$

$$= -\min_{\mu \in \mathfrak{M}(L)} \left(A_{L}(\mu) - \langle c, \rho(\mu) \rangle \right) =$$

$$= -\min_{\mu \in \mathfrak{M}(L)} A_{L-c}(\mu). \qquad (1.7)$$

Note that for a given μ invariant under the Euler-Lagrange flow Φ_t^L , we have $\langle c, \rho(\mu) \rangle = \int_{\mathrm{T}M} \hat{\eta} \, d\mu$, for any closed 1-form η on M with cohomology class c; therefore, we have $A_{L-c}(\mu) = \int_{\mathrm{T}M} (L-\hat{\eta}) \, d\mu$. It is interesting to remark that the value of this α -function coincides with what is called $Ma\tilde{n}\acute{e}$'s critical value, which will be introduced later in sections 1.3 and 1.4. Analogously to what we have already done before, we want to single out the invariant measures that minimize this c-average action.

Definition 1.2.5. If $\mu \in \mathfrak{M}(L)$ and μ minimizes A_{L-c} , i.e., $A_L(\mu) = -\alpha(c)$, we will say that μ is a c-action minimizing measure (or c-minimal measure, or Mather's measure with cohomology c).

An important fact is next lemma, which will help to clearify the relation (and duality) between these two minimizing procedures. To state it, recall that, like any convex function on a finite-dimensional space, the Mather function β admits a subderivative at each point $h \in H_1(M; \mathbb{R})$, *i.e.*, we can find $c \in H^1(M; \mathbb{R})$ such that

$$\forall h' \in \mathrm{H}_1(M; \mathbb{R}), \quad \beta(h') - \beta(h) \ge \langle c, h' - h \rangle$$

As it is usually done, we will denote by $\partial\beta(h)$ the set of $c \in H^1(M;\mathbb{R})$ that are subderivatives of β at h, *i.e.*, the set of c which satisfy the inequality above. By Fenchel's duality, we have

$$c \in \partial \beta(h) \Longleftrightarrow \langle c, h \rangle = \alpha(c) + \beta(h).$$

Lemma 1.2.6. If $\mu \in \mathfrak{M}(L)$, then $A_L(\mu) = \beta(\rho(\mu))$ if and only if there exists $c \in \mathrm{H}^1(M;\mathbb{R})$ such that μ minimizes A_{L-c} (i.e., $A_{L-c}(\mu) = -\alpha(c)$). Moreover, if $\mu \in \mathfrak{M}(L)$ satisfies $A_L(\mu) = \beta(\rho(\mu))$, and $c \in \mathrm{H}^1(M;\mathbb{R})$, then μ minimizes A_{L-c} if and only if $c \in \partial\beta(\rho(\mu))$ (or $\langle c, h \rangle = \alpha(c) + \beta(\rho(\mu))$.

Proof. We will prove both statement at the same time. Assume $A_L(\mu_0) = \beta(\rho(\mu_0))$.

Let $c \in \partial \beta(\rho(\mu_0))$, by Fenchel's duality this is equivalent to

$$\alpha(c) = \langle c, \rho(\mu_0) \rangle - \beta(\rho(\mu_0))$$
$$= \langle c, \rho(\mu_0) \rangle - A_L(\mu_0)$$
$$= -A_{L-c}(\mu_0).$$

Therefore by (1.7): $A_{L-c}(\mu_0) = \min_{\mu \in \mathfrak{M}} A_{L-c}(\mu).$

Assume conversely that $A_{L-c}(\mu_0) = \min_{\mu \in \mathfrak{M}} A_{L-c}(\mu)$, for some given cohomology class c. Therefore if follows from (1.7) that

$$\alpha(c) = -A_{L-c}(\mu_0),$$

which can be written as

$$\langle c, \rho(\mu_0) \rangle = \alpha(c) + A_L(\mu_0).$$

It now suffices to use the Fenchel inequality $\langle c, \rho(\mu_0) \rangle \leq \alpha(c) + \beta(\rho(\mu_0))$, and the inequality $\beta(\rho(\mu_0) \leq A_L(\mu_0))$, given by the definition of β , to obtain the equality

$$\langle c, \rho(\mu_0) \rangle = \alpha(c) + \beta(\rho(\mu_0)).$$

In particular, we have $A_L(\mu_0) = \beta(\rho(\mu_0))$.

The above discussion leads to two equivalent formulations for the minimality of a measure μ :

- there exists a homology class $h \in H_1(M; \mathbb{R})$, namely its rotation vector $\rho(\mu)$, such that μ minimizes A_L amongst all measures in $\mathfrak{M}(L)$ with rotation vector h; *i.e.*, $A_L(\mu) = \beta(h)$;
- there exists a cohomology class $c \in H^1(M; \mathbb{R})$, namely any subderivative of β at $\rho(\mu)$ (*i.e.*, the slope of a supporting hyperplane of the epigraph of β at

 $\rho(\mu)$, such that μ minimizes A_{L-c} amongst all probability measures in $\mathfrak{M}(L)$; *i.e.*, $A_{L-c}(\mu) = -\alpha(c)$.

For $h \in H_1(M; \mathbb{R})$ and $c \in H^1(M; \mathbb{R})$, let us define

$$\mathfrak{M}^{h} := \mathfrak{M}^{h}(L) = \{ \mu \in \mathfrak{M}(L) : A_{L}(\mu) < +\infty, \ \rho(\mu) = h \text{ and } A_{L}(\mu) = \beta(h) \}$$
$$\mathfrak{M}_{c} := \mathfrak{M}_{c}(L) = \{ \mu \in \mathfrak{M}(L) : A_{L}(\mu) < +\infty \text{ and } A_{L-c}(\mu) = -\alpha(c) \}.$$

Observe that because of the superlinear growth condition in the fiber, $A_L(\mu) < +\infty$ implies $A_{L-c}(\mu) < +\infty$. Moreover, both procedures lead to the same sets of minimal measures:

$$\bigcup_{h\in\mathrm{H}_1(M;\mathbb{R})}\mathfrak{M}^h=\bigcup_{c\in\mathrm{H}^1(M;\mathbb{R})}\mathfrak{M}_c\,.$$

In fact, minimizing over a set of invariant measures with a fixed rotation vector or minimizing - globally - the modified Lagrangian (corresponding to a certain cohomology class) are dual problems, as the ones that often appears in linear programming and optimization.

We can now begin to discuss some dynamical implications of this approach. Let us start by defining a first important family of invariant sets: *Mather sets*. For a cohomology class $c \in H^1(M; \mathbb{R})$, we call *Mather set of cohomology class c* the set:

$$\widetilde{\mathcal{M}}_c := \overline{\bigcup_{\mu \in \mathfrak{M}_c} \operatorname{supp} \mu} \subset \mathrm{T}M$$
(1.8)

while the projection on the base manifold $\mathcal{M}_c = \pi\left(\widetilde{\mathcal{M}}_c\right) \subseteq M$ is called *projected* Mather set (with cohomology class c). In [57] Mather proved the celebrated graph theorem:

Theorem 1.2.7. Let $\widetilde{\mathcal{M}}_c$ be defined as in (1.8). The set $\widetilde{\mathcal{M}}_c$ is compact, invariant

under the Euler-Lagrange flow and $\pi | \widetilde{\mathcal{M}}_c$ is an injective mapping of $\widetilde{\mathcal{M}}_c$ into M, and its inverse $\pi^{-1} : \mathcal{M}_c \longrightarrow \widetilde{\mathcal{M}}_c$ is Lipschitz. Moreover this set is contained in the energy level corresponding to the value $\alpha(c)$, i.e.,

$$H \circ \mathcal{L}(x, v) = \alpha(c) \qquad \forall (x, v) \in \widetilde{\mathcal{M}}_c.$$
(1.9)

Remark 1.2.8. The last statement, corresponding to (1.9), is due to Dias-Carneiro [20] and holds only in the autonomous case.

Analogously, one can consider the Mather set corresponding to a rotation vector $h \in H_1(M; \mathbb{R})$ as

$$\widetilde{\mathcal{M}}^h := \overline{\bigcup_{\mu \in \mathfrak{M}^h} \operatorname{supp} \mu} \subset \mathrm{T}M\,, \qquad (1.10)$$

and the projected one $\mathcal{M}^h = \pi\left(\widetilde{\mathcal{M}}^h\right) \subseteq M$. Notice that by Lemma 1.2.6, if $c \in \partial\beta(h)$, we have

$$\widetilde{\mathcal{M}}^h \subseteq \widetilde{\mathcal{M}}_c.$$

Therefore, although this was not shown in [57], the set $\widetilde{\mathcal{M}}^h$ also has a Lipschitz graph over the base.

Theorem 1.2.9. Let $\widetilde{\mathcal{M}}^h$ be defined as in (1.10). $\widetilde{\mathcal{M}}^h$ is compact, invariant under the Euler-Lagrange flow and $\pi | \widetilde{\mathcal{M}}^h$ is an injective mapping of $\widetilde{\mathcal{M}}^h$ into M and its inverse $\pi^{-1} : \mathcal{M}^h \longrightarrow \widetilde{\mathcal{M}}^h$ is Lipschitz.

Remark 1.2.10. Though the graph property for $\widetilde{\mathcal{M}}^h$ is not proved in [57], it was shown there that the support of an action minimizing measure has the graph property. The graph property for $\widetilde{\mathcal{M}}^h$ can be also deduced from this last property. In fact, since the space of probability measures on TM is a separable metric space, one can take a countable dense set $\{\mu_n\}_{n=1}^{\infty}$ of Mather's measures with rotation vector h and consider the new measure $\tilde{\mu} = \sum_{n=1}^{\infty} \frac{1}{2^n} \mu_n$. This is still an invariant probability measure with rotation vector h and $\operatorname{supp} \tilde{\mu} = \widetilde{\mathcal{M}}_h$. Therefore, as the support of a single action minimizing measure, $\widetilde{\mathcal{M}}^h$ has the graph property.

Moreover, this remark points out that there always exist Mather's measures μ^h and μ_c of full support, *i.e.*, $\operatorname{supp} \mu^h = \widetilde{\mathcal{M}}^h$ and $\operatorname{supp} \mu_c = \widetilde{\mathcal{M}}_c$. We will say that an action minimizing measure μ with rotation vector h (resp. cohomology c) has maximal support if $\operatorname{supp} \mu = \widetilde{\mathcal{M}}^h$ (resp. $\operatorname{supp} \mu = \widetilde{\mathcal{M}}_c$).

ADDENDUM

Holonomic measures

Before concluding this section, we would like to stress that using the above approach the minimizing measures are obtained through a variational principle over the set of invariant probability measure. Because of the request of "invariance", this set clearly depends on the Lagrangian that one is considering.

An alternative approach, slightly different under this respect, was due to Ricardo Mañé [46] (see also [23]). This deals with the bigger set of *holonomic measures* and prove extremely advantageous when dealing with different Lagrangians at the same time. In this addendum we want to sketch the basic ideas behind it.

Let C^0_{ℓ} be the set of continuous functions $f : TM \to \mathbb{R}$ growing (fiberwise) at most linearly, *i.e.*,

$$||f||_{\ell} := \sup_{(x,v)\in \mathrm{T}M} \frac{f(x,v)}{1+||v||} < +\infty,$$

and let \mathfrak{M}_{M}^{ℓ} be the set of probability measures on the Borel σ -algebra of TM such that $\int_{\mathrm{TM}} \|v\| d\mu < \infty$, endowed with the unique metrizable topology given by:

$$\mu_n \longrightarrow \mu \qquad \Longleftrightarrow \qquad \int_{\mathrm{T}M} f(x,v) \, d\mu_n \longrightarrow \int_{\mathrm{T}M} f(x,v) \, d\mu \quad \forall f \in C^0_{\ell}$$

Let $(C_{\ell}^{0})^{*}$ be the dual of C_{ℓ}^{0} . Then \mathfrak{M}_{M}^{ℓ} can be naturally embedded in $(C_{\ell}^{0})^{*}$ and its topology coincides with that induced by the weak^{*} topology on $(C_{\ell}^{0})^{*}$. One can show that this topology is metrizable and a metric is, for instance:

$$d(\mu_1, \mu_2) = \left| \int_{\mathrm{T}M} \|v\| \, d\mu_1 - \int_{\mathrm{T}M} \|v\| \, d\mu_2 \right| + \sum_n \frac{1}{2^n c_n} \left| \int_{\mathrm{T}M} \varphi_n \, d\mu_1 - \int_{\mathrm{T}M} \varphi_n \, d\mu_2 \right| \,,$$

where $\{\varphi_n\}_n$ is a sequence of functions with compact support on C_ℓ^0 , which is dense on C_ℓ^0 (in the topology of uniform convergence on compact subsets of TM) and $c_n := \sup_{\mathrm{TM}} |\varphi_n(x, v)|$. The space of probability measures that we will be considering is a closed subset of \mathfrak{M}_M^ℓ (endowed with the induced topology), which is defined as follows. If $\gamma : [0, T] \to M$ is a closed absolutely continuous curve, let μ_γ be such that

$$\int_{\mathrm{T}M} f(x,v) \, d\mu_{\gamma} = \frac{1}{T} \int_0^T f(\gamma(t), \dot{\gamma}(t)) \, dt \qquad \forall f \in C_\ell^0 \, .$$

Observe that $\mu_{\gamma} \in \mathfrak{M}_{M}^{\ell}$ because if γ is absolutely continuous then $\int |\dot{\gamma}(t)| dt < +\infty$. Let $\mathcal{C}(M)$ be the set of such μ_{γ} 's and $\overline{\mathcal{C}(M)}$ its closure in \mathfrak{M}_{M}^{ℓ} . This set is convex and it is called the set of *holonomic measures* on M.

One can check that the following properties are satisfied:

- i) $\mathfrak{M}(L) \subseteq \overline{\mathcal{C}(M)} \subseteq \mathfrak{M}_{M}^{\ell}$. In particular, for every Tonelli Lagrangian L on M, all probabilities measures μ that are invariant with respect to the Euler-Lagrange flow and such that $\int_{\mathrm{T}M} L \, d\mu < +\infty$, are contained in $\overline{\mathcal{C}(M)}$.
- ii) To any given probability $\mu \in \mathcal{C}(M)$, one can associate a rotation vector $\rho(\mu) \in H_1(M; \mathbb{R})$. This map extends continuously to a map

$$\rho: \overline{\mathcal{C}(M)} \longrightarrow \mathrm{H}_1(M;\mathbb{R})$$

and this extension is surjective.

- iii) For each $C \in \mathbb{R}$ the set $\left\{ \mu \in \overline{\mathcal{C}(M)} : A_L(\mu) \leq C \right\}$ is compact.
- iv) If a measure $\mu \in \overline{\mathcal{C}(M)}$ satisfies

$$A_L(\mu) = \min\left\{A_L(\nu): \ \nu \in \overline{\mathcal{C}(M)}\right\},\$$

then $\mu \in \mathfrak{M}(L)$ (and in particular it is invariant). Observe that the existence of probabilities attaining the minimum follows from iii).

In view of these properties, it is clear that the corresponding minimizing problem, although on a bigger space of measures, will lead to the same results as before and the same definition of Mather sets.

1.3 Global action minimizing curves: Aubry and Mañé sets

In addition to the Mather sets, one can construct other compact invariant sets, which are also particularly significant from both a dynamical system and a geometric point of view: the *Aubry sets* and the *Mañé sets*. Instead of considering action minimizing invariant probability measures, we now shift our attention to *c*-action-minimizing curves, also called *c*-global minimizers, for any given cohomology class $c \in H^1(M; \mathbb{R})$.

In the light of remark 1.2.1, let us fix a cohomology class $c \in H^1(M; \mathbb{R})$ and choose a smooth 1-form η on M that represents c. As we have already pointed out in Section 1.1, there is a close relation between solutions of the Euler-Lagrange flow and extremals of the action functional $A_{L_{\eta}}$ for the fixed end-point problem (which are the same as the extremals of A_L). In general, these extremals are not minima (they are local minima only if the time length is very short). Nevertheless, such minima exist; this is a classical result of the calculus of variations, known as *Tonelli Theorem*, that has been reproven - in the setting of Tonelli Lagrangians - by John Mather [57].

Theorem 1.3.1 (Tonelli Theorem, [57]). Let M be a compact manifold and L a Tonelli Lagrangian on TM. For all $a < b \in \mathbb{R}$ and $x, y \in M$, there exists - in the set of absolutely continuous curves $\gamma : [a, b] \longrightarrow M$ such that $\gamma(a) = x$ and $\gamma(b) = y$ - a curve that minimizes the action $A_L(\gamma) = \int_a^b L(\gamma(t), \dot{\gamma}(t)) dt$.

A curve minimizing $A_{L\eta}(\gamma) = \int_a^b L_\eta(\gamma(t), \dot{\gamma}(t)) dt$ subject to the fixed end-point condition $\gamma(a) = x$ and $\gamma(b) = y$, is called a *c*-Tonelli minimizer. Recall that such minimizers do only depend on *c* and not on the chosen representative η (see remark 1.2.1). As Mañé pointed out in [48], for these minimizers to exist it is not necessary to assume the compactness of M: the superlinear growth condition with respect to some complete Riemannian metric on M is enough.

Remark 1.3.2. A Tonelli minimizer which is C^1 is C^r (if the Lagrangian L is C^r) and satisfies the Euler-Lagrange equation; this follows from the usual elementary arguments in the calculus of variations, together with Caratheodory's remark on differentiability. In the autonomous case, Tonelli minimizers will be always C^1 . In the non-autonomous time-periodic case (Tonelli Theorem holds also in this case [57]), as already remarked in remark 1.1.2, one needs to require that the Euler-Lagrange flow is complete.

Our interest will be for particular Tonelli minimizers that are defined for all times and whose action is minimal with respect to any given time length. We will see that these curves present a very rich structure.

Definition 1.3.3 (c-minimizers). An absolutely continuous curve $\gamma : \mathbb{R} \longrightarrow M$ is a c-(global) minimizer if for any given $a < b \in \mathbb{R}$

$$A_{L_n}(\gamma | [a, b]) = \min A_{L_n}(\sigma)$$

where the minimum is taken over all $\sigma : [a, b] \to M$ such that $\sigma(a) = \gamma(a)$ and $\sigma(b) = \gamma(b)$.

A first interesting property of such curves is the following [23].

Proposition 1.3.4. Any c-minimizer is contained in the energy level

$$\widetilde{E}_c = \{ (x, v) \in \mathrm{T}M : H \circ \mathcal{L}(x, v) = \alpha(c) \},\$$

where $\alpha : \mathrm{H}^1(M; \mathbb{R}) \to \mathbb{R}$ is Mather's α -function defined in (1.6).

This result can be also deduced from Carneiro result [20] (see also 1.9 in Theorem 1.2.7), observing that the energy is conserved along solutions and each *c*-minimizer contains *c*-action minimizing measures in its α and ω -limit sets.

Remark 1.3.5. It is a result by Albert Fathi [32], that these *c*-minimizers satisfy a stronger property: they are *time-free* minimizer for $L_{\eta} + \alpha(c)$, *i.e.*, for any $a < b \in \mathbb{R}$ they minimize the action of $L_{\eta} + \alpha(c)$ over the set of all curves with the same endpoints, independently of their time lengths. Namely, if γ is a *c*-minimizer, then for any $\sigma : [a', b'] \to M$ such that $\sigma(a') = \gamma(a)$ and $\sigma(b') = \gamma(b)$ we have

$$\int_{a}^{b} L_{\eta}(\gamma(t), \dot{\gamma}(t)) + \alpha(c) dt \leq \int_{a'}^{b'} L_{\eta}(\sigma(t), \dot{\sigma}(t)) + \alpha(c) dt$$

This is not true anymore in the non-autonomous case. See the addendum at the end of this section for more details.

We can now define another family of invariant sets.

Definition 1.3.6 (Mañé set). The Mañé set (with cohomology class c) is:

$$\widetilde{\mathcal{N}}_c = \bigcup \left\{ (\gamma(t), \dot{\gamma}(t)) : \gamma \text{ is a } c \text{-minimizer and } t \in \mathbb{R} \right\}.$$
(1.11)

One can show that $\widetilde{\mathcal{N}}_c$ is non-empty and compact, it is contained in the energy level \widetilde{E}_c corresponding to $\alpha(c)$ and, above all, it contains the Mather set $\widetilde{\mathcal{M}}_c$ (see theorem 1.3.17). Differently from what happens with the Mather set, in general, the Mañé set is not a graph.

Another important family of sets, which also play an extremely significant role in the study of the dynamics of Lagrangian system, consists of the *Aubry sets*. In order to define these sets, we need some extra tools that will allow us to single out particular kinds of *c*-minimizers. Let us start by introducing a so-called "barrier" function. As done by Mather in [59], it is convenient to define the following quantity for t > 0 and $x, y \in M$:

$$h_{\eta,t}(x,y) = \min \int_0^t L_\eta(\gamma(s), \dot{\gamma}(s)) \, ds \,,$$
 (1.12)

where the minimum is taken over all piecewise C^1 paths $\gamma : [0, t] \longrightarrow M$, such that $\gamma(0) = x$ and $\gamma(t) = y$. This minimum is achieved because of Tonelli theorem. We define the *Peierls barrier* as:

$$h_{\eta}(x,y) = \liminf_{t \to +\infty} (h_{\eta,t}(x,y) + \alpha(c)t).$$
 (1.13)

Remark 1.3.7. Observe that h_{η} does not depend only on the cohomology class c, but also on the choice of the representative η ; namely, if $\eta' = \eta + df$, then $h_{\eta'}(x, y) = h_{\eta}(x, y) + f(y) - f(x)$. Anyhow, this dependence will not be harmful for what we are going to do in the following.

This function is finite for all $x, y \in M$ and it can be shown that it is Lipschitz. Moreover, Albert Fathi [32] showed that - in the autonomous case - this limit can be replaced with a lim. This is not generally true in the non-autonomous timeperiodic case (see for instance [36] for some counterexamples); Tonelli Lagrangians for which this convergence result holds are called *regular*. Patrick Bernard [11] showed that under suitable assumptions on the Mather set it is possible to prove that the Lagrangian is regular. For instance, if the Mather set $\tilde{\mathcal{M}}_c$ is union of 1-periodic orbits, then L_η is regular. This problem turned out to be strictly related to the convergence of the so-called *Lax-Oleinik semigroup* (see [32] for its definition).

Remark 1.3.8. The function h_{η} is a generalization of Peierls Barrier introduced by Aubry [6] and Mather [52, 54, 58] in their study of twist maps. In some sense we are comparing, in the limit, the action of Tonelli minimizers of time length T with the corresponding average *c*-minimimal action $-\alpha(c)T$. Remember, in fact, that $-\alpha(c)$ is the "average action" of a *c*-minimal measure. In particular, the finiteness of such liminf can be used to get a further characterization of $\alpha(c)$:

$$\alpha(c) = \sup \left\{ k \in \mathbb{R} : \liminf_{t \to +\infty} (h_{\eta,t}(x,y) + kt) = -\infty \quad \text{for all } x, y \in M \right\} = \\ = \min \left\{ k \in \mathbb{R} : \liminf_{t \to +\infty} (h_{\eta,t}(x,y) + kt) \in \mathbb{R} \quad \text{for all } x, y \in M \right\}.$$

See [48, 22, 23] for these and other similar characterizations. This quantity is often referred to as $Ma\tilde{n}\acute{e}$ critical value) (see also the addendum at the end of this section and remark 1.4.5).

Many properties of this barrier have been studied by several authors, leading to a better understanding of its dynamical meaning and, we will see in section 1.4, its relation with "weak" solutions of Hamilton-Jacobi equations. We can summarize some of them in the following proposition (see for example [11, 23, 32, 59] for a proof).

Proposition 1.3.9. The values of the map h_{η} are finite. Moreover, the following properties hold:

- i) h_{η} is Lipschitz;
- ii) for each $x, y \in M$, $h_{\eta}(x, y) + h_{\eta}(y, x) \ge 0$.

- iii) for each $x \in M$, $h_{\eta}(x, x) \ge 0$;
- iv) for each $x \in \mathcal{M}_c$, $h_\eta(x, x) = 0$;
- v) for each $x, y, z \in M$ and t > 0, $h_{\eta}(x, y) \leq h_{\eta}(x, z) + h_{\eta,t}(z, y) + \alpha(c)t$ and $h_{\eta}(x, y) \leq h_{\eta,t}(x, z) + \alpha(c)t + h_{\eta,t}(z, y);$
- vi) for each $x, y, z \in M$, $h_{\eta}(x, y) \leq h_{\eta}(x, z) + h_{\eta}(z, y)$.

Inspired by these properties, one can consider its symmetrization:

$$\delta_c: M \times M \longrightarrow \mathbb{R}$$

(x,y) $\longmapsto h_\eta(x,y) + h_\eta(y,x).$ (1.14)

Observe that this function does now depend only on the cohomology class c and moreover it is non-negative, symmetric and satisfies the triangle inequality; therefore, it is a pseudometric on the set

$$\mathcal{A}_{c} = \{ x \in M : \delta_{c}(x, x) = 0 \}.$$
(1.15)

 \mathcal{A}_c is called the *projected Aubry set* associated to L and c, and δ_c is *Mather's pseudo*metric.

Remark 1.3.10. One can easily construct a metric space out of $(\mathcal{A}_c, \delta_c)$. We call *quotient Aubry set* the metric space $(\bar{\mathcal{A}}_c, \bar{\delta}_c)$ obtained by identifying two points in \mathcal{A}_c , if their δ_c -pseudodistance is zero. This set will the main object of study in chapter 2.

An equivalent definition of the projected Aubry set is the following (it follows immediately from the definition of δ_c):

Proposition 1.3.11. $x \in \mathcal{A}_c$ if and only if there exists a sequence of absolutely continuous curves $\gamma_n : [0, t_n] \to M$ such that:

- for each n, we have $\gamma_n(0) = \gamma_n(t_n) = x$;
- the sequence $t_n \to +\infty$, as $n \to +\infty$;

- as
$$n \to +\infty$$
, $\int_0^{t_n} L_\eta(\gamma_n(s), \dot{\gamma}_n(s)) ds + \alpha(c) t_n \to 0.$

Therefore, it consists of points that are contained in loops with period as long as we want and action as close as we want to the minimal average one.

Remark 1.3.12. An interesting property of the pseudometric δ_c is the following (see [59]). If d denotes the distance induced on M by the Riemannian metric g, then there exists C > 0 such that for each $x, y \in M$ we have

$$\delta_c(x,y) \le Cd(x,y)^2.$$

The same estimate continues to be true for the non-autonomous time-periodic case. In this case we have that

$$\delta_c((x,\tau_0),(y,\tau_1)) \le C[d(x,y) + \|\tau_1 - \tau_0\|]^2$$

for each $(x, \tau_0), (y, \tau_1) \in \mathcal{A}_c$, where

$$\|\tau_1 - \tau_0\| = \inf \{ |t_1 - t_0| : t_i \in \mathbb{R}, t_i \equiv \tau_i \pmod{1}, i = 0, 1 \}.$$

This estimate will be particularly useful in section 2.2 (addendum) to prove the total disconnectedness of the quotient Aubry set in low dimension.

We want to show now that \mathcal{A}_c is the projection of an invariant compact subset of TM, which will be called *Aubry set*. Let start with the following observation.

Remark 1.3.13. Let $\gamma : \mathbb{R} \longrightarrow M$ be a *c*-minimizer and consider x_{α}, x'_{α} in the

 α -limit set¹ of γ and x_{ω}, x'_{ω} in the ω -limit set² of γ . John Mather in [59] proved that $\delta_c(x_{\alpha}, x'_{\alpha}) = \delta_c(x_{\omega}, x'_{\omega}) = 0$. In general, it is not true that $\delta_c(x_{\alpha}, x_{\omega})=0$; what one can prove is that this value does not depend on the particular x_{α} and $x_{\omega}, i.e., \delta_c(x_{\alpha}, x_{\omega}) = \delta_c(x'_{\alpha}, x'_{\omega})$: it is a property of the limit sets rather than of their elements. Nevertheless, there will exist particular *c*-minimizers for which this value is equal to 0 and these will be the *c*-minimizers that we want to single out.

Definition 1.3.14 (c-regular minimizers). A c-minimizer $\gamma : \mathbb{R} \longrightarrow M$ is called a c-regular minimizer, if $\delta_c(x_{\alpha}, x_{\omega}) = 0$ for each x_{α} in the α -limit set of γ and x_{ω} in the ω -limit set of γ .

We can now define the $Aubry \ set$ as the union of the support of all these c-regular minimizers.

Definition 1.3.15 (Aubry set). The Aubry set (with cohomology class c) is:

$$\widetilde{\mathcal{A}}_c = \bigcup \left\{ (\gamma(t), \dot{\gamma}(t)) : \gamma \text{ is a } c\text{-regular minimizer and } t \in \mathbb{R} \right\}.$$
(1.16)

This set $\widetilde{\mathcal{A}}_c$ is clearly contained in the Mañé set $\widetilde{\mathcal{N}}_c$. In particular, each *c*-minimizer is asymptotic to $\widetilde{\mathcal{A}}_c$ (see also [23]). One of its most important properties is that, similarly to the Mather set, it is also a graph over the base (see [59, 32]):

Theorem 1.3.16. $\pi | \widetilde{\mathcal{A}}_c$ is a bi-Lipschitz homeomorphism and $\pi (\widetilde{\mathcal{A}}_c) = \mathcal{A}_c$.

Furthermore, there is a clear relation between the support of c-minimal measures and these sets.

¹Recall that a point z is in the α -limit set of γ , if there exists a sequence $t_n \to -\infty$ such that $\gamma(t_n) \to z$.

²Recall that a point z is in the ω -limit set of γ , if there exists a sequence $t_n \to +\infty$ such that $\gamma(t_n) \to z$.

Theorem 1.3.17. For any $c \in H^1(M; \mathbb{R})$, $\widetilde{\mathcal{M}}_c \subseteq \widetilde{\mathcal{A}}_c$. Moreover:

$$\mu \in \mathfrak{M}_c \iff \mu \in \mathfrak{M}(L) \text{ and } \operatorname{supp} \mu \subseteq \widetilde{\mathcal{A}}_c$$
$$\iff \mu \in \mathfrak{M}(L) \text{ and } \operatorname{supp} \mu \subseteq \widetilde{\mathcal{N}}_c. \tag{1.17}$$

The fact that an invariant measure whose support is contained in $\widetilde{\mathcal{N}}_c$ (or $\widetilde{\mathcal{A}}_c$) is *c*-minimizing might be seen, for example, as a consequence of the finiteness of Peierls barrier. The proof of the first equivalence in (1.17) goes back to Mañé [48]. As far as the second equivalence is concerned, it follows from the fact that the *non-wandering* set of $\Phi_t^L | \widetilde{\mathcal{N}}_c$ is contained in $\widetilde{\mathcal{A}}_c$ (we will prove it in the addendum in section 1.4).

Moreover, the following relation between these sets holds.

Proposition 1.3.18. Let ρ , c be respectively an arbitrary homology class in $H_1(M; \mathbb{R})$ and an arbitrary cohomology class $H^1(M; \mathbb{R})$. We have

(1)
$$\widetilde{\mathcal{M}}^{\rho} \cap \widetilde{\mathcal{A}}_c \neq \emptyset \quad \iff \quad (2) \widetilde{\mathcal{M}}^{\rho} \subseteq \widetilde{\mathcal{A}}_c \quad \iff \quad (3) \ \rho \in \partial \alpha(c) \,.$$

Proof. The implication (2) \implies (1) is trivial. Let us prove that (1) \implies (3). If $\widetilde{\mathcal{M}}^{\rho} \cap \widetilde{\mathcal{A}}_c \neq 0$, then there exists a *c*-minimizing invariant measure μ with rotation vector ρ . Let η be a closed 1-form with $[\eta] = c$; from the definition of α and β :

$$-\alpha(c) = \int_{\mathrm{T}M} (L - \hat{\eta}) \, d\mu = \int_{\mathrm{T}M} L \, d\mu - \langle c, \rho \rangle = \beta(\rho) - \langle c, \rho \rangle \,;$$

since β and α are convex conjugated, then ρ is a subderivative of α at c.

Finally, in order to show (3) \implies (2), let us prove that any action minimizing measure with rotation vector ρ is *c*-minimizing. In fact, if $\rho \in \partial \alpha(c)$ then $\alpha(c) = \langle c, \rho \rangle - \beta(\rho)$; therefore for any $\mu \in \mathfrak{M}^{\rho}$ and η as above:

$$-\alpha(c) = \beta(\rho) - \langle c, \rho \rangle = \int_{\mathrm{T}M} (L - \hat{\eta}) \, d\mu.$$

This proves that $\mu \in \mathfrak{M}_c$ and concludes the proof.

We can summarize what discussed so far in the following diagram:³

We have already observed that the Mañé set $\widetilde{\mathcal{N}}_c$ in general is not a graph over the base. Anyhow, one can prove a sort of "graph" property over the Aubry set (see [11]). More precisely, $\widetilde{\mathcal{N}}_c \cap \pi^{-1}(\mathcal{A}_c) = \widetilde{\mathcal{A}}_c$, where $\pi : \mathrm{T}M \to M$ denotes the canonical projection. In other words, there is a Lipschitz section $\mathcal{V}_c : \mathcal{A}_c \longrightarrow \mathrm{T}M$, such that for each $x \in \mathcal{A}$ there exists one and only *c*-minimizer $\gamma : \mathbb{R} \to M$ satisfying $\gamma(0) = x$; this minimizer is *regular* and is given by $\gamma(t) = \pi \left(\Phi_t^L(x, \mathcal{V}_c(x))\right)$.

Moreover, one can prove the following properties of $\widetilde{\mathcal{A}}_c$ and $\widetilde{\mathcal{N}}_c$.

Proposition 1.3.19 ([22]). $\widetilde{\mathcal{N}}_c$ is chain transitive ⁴ and $\widetilde{\mathcal{A}}_c$ is chain recurrent ⁵.

ADDENDA

In these addenda we want to discuss some other definitions of minimizers that appear in the literature and some differences between the autonomous and nonautonomous case.

³The typographical "coincidence", honoring Ricardo Mañé, was noticed by Albert Fathi.

⁴Namely, for each $\varepsilon > 0$ and for all $(x, v), (y, w) \in \widetilde{\mathcal{N}}_c$, there exists an ε -pseudo-orbit for the flow Φ^L connecting them. In other words, there exist $\{(x_n, v_n)\}_{n=0}^{k_{\varepsilon}} \subset \widetilde{\mathcal{N}}_c$ and positive times $t_1, \ldots, t_{k_{\varepsilon}} > 0$ such that $(x_0, v_0) = (x, v), (x_{k_{\varepsilon}}, v_{k_{\varepsilon}}) = (y, w)$ and dist $(\Phi^L_{t_{i+1}}((x_i, v_i)), (x_{i+1}, v_{i+1})) \leq \varepsilon$ for all $i = 0, \ldots, k_{\varepsilon}$.

⁵As in the definition of chain transitivity, but in this case the end-points are the same

Semi-static and Static curves

In some of the literature, in particular in the production of Ricardo Mañé and his school (see for instance [48, 22, 23]), *c*-minimizers and *c*-regular minimizers are replaced by the notions of *c*-semi-static curves and *c*-static curves. They used these curves for defining the equivalent of Aubry and Mañé sets that we have introduced above. It turns out that - at least in the autonomous case - these different kinds of curves do indeed coincide. Let us try to give an idea of these definitions and their equivalence.

First of all they were not considering Peierls barrier as in our definition, but something different, that is commonly called $Ma\tilde{n}\acute{e}$ potential:

$$\phi_{\eta}(x,y) = \inf_{t>0} \left(h_t(x,y) + \alpha(c)t \right)$$

Clearly $\phi_{\eta}(x, y) \leq h_{\eta}(x, y)$ everywhere. Moreover, it shares many properties with Peierls barrier: it is finite, Lipschitz, it satisfies the triangle inequality (vi) in proposition 1.3.9) and, above all, $\phi_{\eta}(x, x) = 0$ if and only if $h_{\eta}(x, x) = 0$ (see for instance [11]).

Definition 1.3.20 (Semi-static and static curves). An absolutely continuous curve $\gamma : \mathbb{R} \longrightarrow M$ is a c-semi-static if for any given $a < b \in \mathbb{R}$

$$\int_{a}^{b} L_{\eta}(\gamma(t), \dot{\gamma}(t)) dt + \alpha(c)(b-a) = \phi_{\eta}(\gamma(a), \gamma(b))$$

Moreover, an absolutely continuous curve $\gamma : \mathbb{R} \longrightarrow M$ is c-static if for any given $a < b \in \mathbb{R}$

$$\int_{a}^{b} L_{\eta}(\gamma(t), \dot{\gamma}(t)) dt + \alpha(c)(b-a) = -\phi_{\eta}(\gamma(b), \gamma(a)).$$

It is easy to check that c-static curves are c-semistatic. Moreover, in the au-

tonomous case, c-semi-static curves coincide with c-minimizers, while c-static curves with c-regular minimizers (see for instance [32]).

Remark 1.3.21. One can therefore give the definition of *c*-semi-static and static curves also in terms of Peierls barrier. Namely, using remark 1.3.5 it is easy to check that $\gamma : \mathbb{R} \to M$ is *c*-semi-static if for any a < b

$$\int_{a}^{b} L_{\eta}(\gamma(t), \dot{\gamma}(t)) dt + \alpha(c)(b-a) = h_{\eta}(\gamma(a), \gamma(b)).$$

In particular, a *c*-semi-static curve is *c*-static (or equivalently, a *c*-minimizer is regular) if $h_{\eta}(\gamma(a), \gamma(b)) = -h_{\eta}(\gamma(b), \gamma(a))$, for any a < b.

Minimizers in the non-autonomous time-periodic case

In the non-autonomous time-periodic case, the notion of *c*-minimizer that we have given above is a weak notion and the invariant set of curves that one obtains in this way is too big to enjoy many of the properties that the Mañé set does. The main problem is that the property highlighted in remark 1.3.5 fails to be true in the non-autonomous case. For this reason, one needs to introduce a stronger notion of action-minimizing curves and distinguish between *weak c-minimizers* and *c-minimizers* (although it would rather make more sense to call this new class of curves *c-strong minimizers*).

Let M be a compact manifold and $L : \mathbb{T}M \times \mathbb{T} \longrightarrow \mathbb{R}$ a Tonelli Lagrangian, whose Euler-Lagrange flow is complete. If η is a closed 1-form, we let $[\eta] = ([\eta]_M, [\eta]_{\mathbb{T}}) \in$ $\mathrm{H}^1(M \times \mathbb{T}; \mathbb{R}) = \mathrm{H}^1(M; \mathbb{R}) \times \mathbb{R}$ denote its de Rham cohomology class. We will say that η is Mañé critical if $[\eta]_{\mathbb{T}} = -\alpha([\eta]_M)$. All the theory developed so far in this chapter, can be extended to the non-autonomous case considering Mañé critical 1-forms.

In the following, let us fix η to be a Mañé critical 1-form with $[\eta]_M = c$.

Definition 1.3.22 (c-weak minimizers). An absolutely continuous curve $\gamma : \mathbb{R} \longrightarrow M$ is a c-weak minimizer if for any given $a < b \in \mathbb{R}$

$$\int_{a}^{b} (L-\hat{\eta})(\gamma(t),\dot{\gamma}(t),t \operatorname{mod.} 1) dt \leq \int_{a}^{b} (L-\hat{\eta})(\sigma(t),\dot{\sigma}(t),t \operatorname{mod.} 1) dt$$

for any $\sigma : [a, b] \to M$ such that $\sigma(a) = \gamma(a)$ and $\sigma(b) = \gamma(b)$.

Definition 1.3.23 (c-minimizers). An absolutely continuous curve $\gamma : \mathbb{R} \longrightarrow M$ is a c-minimizer if for any given $a < b \in \mathbb{R}$

$$\int_{a}^{b} (L - \hat{\eta})(\gamma(t), \dot{\gamma}(t), t \bmod 1) \, dt \le \int_{a'}^{b'} (L - \hat{\eta})(\sigma(t), \dot{\sigma}(t), t \bmod 1) \, dt$$

for any $\sigma : [a', b'] \to M$ such that $a' < b', a' - a \in \mathbb{Z}, b' - b \in \mathbb{Z}$ and $\sigma(a') = \gamma(a)$ and $\sigma(b') = \gamma(b)$.

One can define the following invariant (compact) subsets:

- $\widetilde{\mathcal{W}}_c = \bigcup \{ (\gamma(t), \dot{\gamma}(t)) : \gamma \text{ is a } c \text{-weak minimizer and } t \in \mathbb{R} \};$
- $\widetilde{\mathcal{N}}_c = \bigcup \{ (\gamma(t), \dot{\gamma}(t)) : \gamma \text{ is a } c \text{-minimizer and } t \in \mathbb{R} \} (Ma \widetilde{n} \acute{e} set);$
- $\widetilde{\mathcal{A}}_c = \bigcup \{ (\gamma(t), \dot{\gamma}(t)) : \gamma \text{ is a } c \text{-regular minimizer and } t \in \mathbb{R} \} (Aubry set).$

Obviously every *c*-minimizer is a *c*-weak minimizer, therefore $\widetilde{\mathcal{N}}_c \subseteq \widetilde{\mathcal{W}}_c$. On the other hand, this inclusion could be strict, since there might exist *c*-weak minimizers that are not *c*-minimizers (see for example [11, 36]). As already remarked above, in the autonomous case these two definitions/sets coincide. Patrick Bernard in [11] proved that for *regular* Lagrangians, *i.e.*, when the liminf in the definition of Peierls barrier is a limit, these two sets coincide: $\widetilde{\mathcal{N}}_c = \widetilde{\mathcal{W}}_c$.

Summarizing, also in the non-autonomous case we have a similar diagram:

Furthermore, the Aubry and Mather sets still satisfy the graph property and proposition 1.3.19 continues to hold, *i.e.*, $\widetilde{\mathcal{N}}_c$ is chain transitive and $\widetilde{\mathcal{A}}_c$ chain recurrent.

1.4 Weak KAM theory

Another interesting approach to the study of these invariant sets is provided by *weak KAM theory*, which represents the functional analytical counterpart of the variational methods discussed in the previous sections. This is mainly based on the concept of "critical" subsolutions and "weak" solutions of Hamilton-Jacobi equation and, we will see, can be interpreted from a symplectic geometric point of view as the study of particular Lagrangian graphs and their non-removable intersections (see [69] and chapter 3). This latter approach is particularly interesting, since it relates the dynamics of the system to the geometry of the space and might potentially open the way to a "symplectic" definition of Aubry-Mather theory (see also section 3.2 and [15]). In this section we want to provide a brief presentation of this theory, omitting most of the proves, for which we remand the reader to the excellent - and self-contained - presentation [32].

The main object of investigation is represented by Hamilton-Jacobi (H-J) equation:

$$H_{\eta}(x, d_x u) = H(x, \eta(x) + d_x u) = k,$$

where η is a closed 1-form on M with a certain cohomology class c. Observe that

considering H-J equations for different 1-forms corresponding to different cohomology classes, is equivalent to Mather's idea of changing Lagrangian (see remark 1.2.1).

From now on, we will consider L to be a Tonelli Lagrangian on a compact manifold M and H its associated Hamiltonian. Let us fix η to be a closed 1-form on M with cohomology class c, and, as before, denote by L_{η} and H_{η} the modified Lagrangian and Hamiltonian (see remark 1.2.1). In classical mechanics, one is interested in studying solutions of this equation, *i.e.*, C^1 functions $u: M \to \mathbb{R}$ such that $H_{\eta}(x, d_x u) = k$. It is immediate to check that for any given cohomology class there exists at most one value of k for which these C^1 solutions may exist. In fact, it is enough to observe that if u and v are two C^1 functions on a compact manifold, there will exist a point x_0 at which their differentials coincide (take any critical point of u - v). We will see that this value of k for which solutions might exist, coincides with $\alpha(c)$ (see theorem 1.4.15).

The existence of such solutions has significant implications to the dynamics of the system and it is, consequently, quite rare. In particular, they correspond to Lagrangian graphs, which are invariant under the Hamiltonian flow Φ_t^H (Hamilton-Jacobi theorem). For instance, in the case of $M = \mathbb{T}^d$ and nearly-integrable systems these solutions correspond to KAM tori (this might give an idea of their rareness).

One of the main results of weak KAM theory is that, in the case of optical Hamiltonians, a weaker kind of solutions do always exist. In the following we are going to define these generalized solutions and their relation with the dynamics of the systems. It is important to point out that one of the main ingredient in the proof of all these results is provided by Fenchel inequality (1.3).

Let us start by generalizing the concept of subsolutions. In the C^1 -case it is easy to check - using Fenchel inequality - that the following property holds.

Proposition 1.4.1. Let $u: M \to \mathbb{R}$ be C^1 ; u satisfies $H(x, \eta(x) + d_x u) \leq k$ for all

 $x \in M$ if and only if for all $\gamma : [a, b] \to M$

$$u(\gamma(b)) - u(\gamma(a)) \le \int_a^b L_\eta(\gamma(t), \dot{\gamma}(t)) \, dt + k(b-a).$$

This last inequality provides the ground for to the definition of *dominated func*tions, which will generalize subsolutions in the C^0 -case.

Definition 1.4.2 (Dominated functions). Let $u : M \to \mathbb{R}$ be a continuous function; u is dominated by $L_{\eta} + k$, and we will write $u \prec L_{\eta} + k$, if for all $\gamma : [a, b] \to M$

$$u(\gamma(b)) - u(\gamma(a)) \le \int_a^b L_\eta(\gamma(t), \dot{\gamma}(t)) dt + k(b-a).$$
(1.18)

One can check that if $u \prec L_{\eta} + k$ then u is Lipschitz and its Lipschitz constant can be bounded by a constant C(k) independent of u; in particular, all dominated functions for values of k in a compact set are equiLipschitz. On the other hand, it is easy to check that each Lipschitz function is dominated by $L_{\eta} + k$, for a suitable kdepending on its Lipschitz constant: this shows that dominated functions exist.

Dominated functions generalize subsolutions of H-J to the continuous case. In fact:

Proposition 1.4.3. If $u \prec L_{\eta} + k$ and $d_x u$ exists, then $H(x, \eta(x) + d_x u) \leq k$. Moreover, if $u: M \to \mathbb{R}$ is Lipschitz and $H(x, \eta(x) + d_x u) \leq k$ a.e., then $u \prec L_{\eta} + k$.

Remark 1.4.4. Using the fact that any Lipschitz function is differentiable almost everywhere (Rademacher theorem), one could equivalently define subsolutions in the following way: a locally Lipschitz function $u : M \longrightarrow \mathbb{R}$ is a subsolution of $H_{\eta}(x, d_x u) = k$, with $k \in \mathbb{R}$, if $H_{\eta}(x, d_x u) \leq k$ for almost every $x \in M$.

Remark 1.4.5. One interesting question is: for which values of k do there exist functions dominated by $L_{\eta} + k$ (or equivalently subsolutions of $H(x, \eta(x) + d_x u) = k$)? It is easy to show that there exists a value $k_c \in \mathbb{R}$ such that $H(x, \eta + d_x u) = k$ does not admit any subsolution for $k < k_c$, while it has subsolutions for $k \ge k_c$, see [45, 32]. In particular, if $k > k_c$ there exist C^{∞} subsolutions. The constant k_c is called $Ma\tilde{n}\acute{e}'s$ critical value and coincides with $\alpha(c)$, where $c = [\eta]$ (see [23, 32]).

Functions corresponding to this "critical domination" play an important role, since they encode significant information about the dynamics of the system.

Definition 1.4.6 (Critical subsolutions). A function $u \prec L_{\eta} + \alpha(c)$ is said to be critically dominated. Equivalently, we will also call it an η -critical subsolution, since $H(x, \eta(x) + d_x u) \leq \alpha(c)$ for almost every $x \in M$.

Remark 1.4.7. The above observation provides a further definition of $\alpha(c)$:

$$\alpha(c) = \inf_{u \in C^{\infty}(M)} \max_{x \in M} H(x, \eta(x) + d_x u)$$

This infimum is not a minimum, but it becomes a minimum over the set of Lipschitz functions on M (also over the smaller set of $C^{1,1}$ functions, see the addendum at the end of this section). This characterization has the following geometric interpretation. If we consider the space T^*M equipped with the canonical symplectic form, the graph of the differential of a $C^1 \eta$ -critical subsolution (plus the 1-form η) is nothing else than a *c*-Lagrangian graph (*i.e.*, a Lagrangian graph with cohomology class *c*). Therefore Mañé *c*-critical energy level $\mathcal{E}_c^* = \{(x, p) \in T^*M : H(x, p) = \alpha(c)\}$ corresponds to a (2d-1)-dimensional hypersurface, such that the region it bounds is convex in each fiber and does not contain in its interior any *c*-Lagrangian graph, while any of its neighborhoods does. See also chapter 3 for more details.

Next step consists in analyzing curves for which equality in (1.18) holds.

Definition 1.4.8 (Calibrated curves). Let $u \prec L_{\eta} + k$. A curve $\gamma : I \rightarrow M$ is (u, L_{η}, k) -calibrated if for any $[a, b] \subseteq I$

$$u(\gamma(b)) - u(\gamma(a)) = \int_a^b L_\eta(\gamma(t), \dot{\gamma}(t)) dt + k(b-a).$$

These curves are very special curves; in fact:

Proposition 1.4.9. If $u \prec L_{\eta} + k$ and $\gamma : [a, b] \rightarrow M$ is (u, L_{η}, k) -calibrated, then γ is a c-Tonelli minimizer, i.e.,

$$\int_{a}^{b} L_{\eta}(\gamma(t), \dot{\gamma}(t)) dt \leq \int_{a}^{b} L_{\eta}(\sigma(t), \dot{\sigma}(t)) dt$$

for any $\sigma : [a, b] \to M$ such that $\sigma(a) = \gamma(a)$ and $\sigma(b) = \gamma(b)$. Most of all, this implies that γ is a solution of the Euler-Lagrange flow and therefore it is C^r (if L is C^r).

Moreover, the following differentiability result holds.

Proposition 1.4.10. Let $u \prec L_{\eta} + k$ and $\gamma : [a, b] \rightarrow M$ be (u, L_{η}, k) -calibrated.

- i) If $d_{\gamma(t)}u$ exists for some $t \in [a, b]$, then $H(\gamma(t), \eta(\gamma(t)) + d_{\gamma(t)}u) = k$ and $d_{\gamma(t)}u = \frac{\partial L}{\partial u}(\gamma(t), \dot{\gamma}(t)).$
- ii) If $t \in (a, b)$, then $d_{\gamma(t)}u$ exists.

Remark 1.4.11. Calibrated curves are "Lagrangian gradient lines" of $\operatorname{grad}_L u$ (where $\operatorname{grad}_L u$ is a multivalued vector field given by the equation $d_x u = \frac{\partial L}{\partial v} (x, \operatorname{grad}_L u)$. Therefore, there is only one possibility for calibrated curves, at each point of differentiability of u.

In order to define *weak solutions*, let us recall this property of classical solutions.

Proposition 1.4.12. Let $u : M \to \mathbb{R}$ a C^1 function and $k \in \mathbb{R}$. The following conditions are equivalent:

- 1. *u* is solution of $H(x, \eta(x) + d_x u) = k$;
- 2. $u \prec L_{\eta} + k$ and for each $x \in M$ there exists $\gamma_x : (-\infty, +\infty) \to M$ such that $\gamma_x(0) = x$ and γ_x is (u, L_{η}, k) -calibrated;

- 3. $u \prec L_{\eta} + k$ and for each $x \in M$ there exists $\gamma_x : (-\infty, 0] \to M$ such that $\gamma_x(0) = x$ and γ_x is (u, L_{η}, k) -calibrated;
- 4. $u \prec L_{\eta} + k$ and for each $x \in M$ there exists $\gamma_x : [0, +\infty) \to M$ such that $\gamma_x(0) = x$ and γ_x is (u, L_{η}, k) -calibrated.

This suggests the following definitions.

Definition 1.4.13 (weak KAM solutions). Let $u \prec L_{\eta} + k$.

- u is a weak KAM solution of negative type (or backward Weak KAM solution)
 if for each x ∈ M there exists γ_x : (-∞, 0] → M such that γ_x(0) = x and γ_x is
 (u, L_η, k)-calibrated;
- u is a weak KAM solution of positive type (or forward Weak KAM solution) if for each x ∈ M there exists γ_x : [0, +∞) → M such that γ_x(0) = x and γ_x is (u, L_η, k)-calibrated.

Remark 1.4.14. Observe that any weak KAM solution of negative type u_- (resp. of positive type u_+) for a given Lagrangian L, can be seen as a weak KAM solution of positive type (resp. of negative type) for the *symmetrical* Lagrangian $\tilde{L}(x, v) := L(x, -v)$.

Let us denote with S_{η}^{-} the set of Weak KAM solutions of negative type and S_{η}^{+} the ones of positive types. Albert Fathi [28, 32] proved that these sets are always non-empty.

Theorem 1.4.15 (Weak KAM theorem). There is only one value of k for which weak KAM solutions of positive or negative type of $H(x, \eta(x) + d_x u) = k$ exist. This value coincides with $\alpha(c)$, where $\alpha : H^1(M; \mathbb{R}) \to \mathbb{R}$ is Mather's α -function defined in (1.6). In particular, for any $u \prec L_{\eta} + \alpha(c)$ there exist a weak KAM solution of negative type u_- and a weak KAM solution of positive type u_+ , such that $u_- = u = u_+$ on the projected Aubry set \mathcal{A}_c . Therefore, for any given weak KAM solution of negative type u_- (resp. of positive type u_+), there exists a weak KAM solution of positive type of positive type u_+ (resp. of negative type u_-) such that $u_- = u_+$ on the projected Aubry set \mathcal{A}_c . Such solutions are said to be *conjugate*. In particular, for any u_- or u_+ there exists a unique conjugate solutions. This follows from the following result.

Proposition 1.4.16. The projected Mather set \mathcal{M}_c is the uniqueness set for weak KAM solutions of the same type. Namely, if u_-, v_- are weak KAM solutions of negative type (resp. u_+, v_+ are weak KAM solutions of positive type) and $u_- = v_-$ on \mathcal{M}_c (resp. $u_+ = v_+$ on \mathcal{M}_c), then they coincide everywhere on M.

Let us try to understand the dynamical meaning of such solutions. Albert Fathi [32] - using these generalized solutions - proved a *weak* version of Hamilton-Jacobi theorem, showing the relation between these weak solutions and the dynamics of the associated Hamiltonian system. We will state it for weak KAM solution of negative type, but - using remark 1.4.14 - one can deduce an analogous statement for weak KAM solutions of positive type.

Theorem 1.4.17 (Weak Hamilton-Jacobi Theorem). Let $u_{-} : M \to \mathbb{R}$ be a weak KAM solution of negative type and consider

$$\operatorname{Graph}(\eta + du_{-}) := \{ (x, \eta(x) + d_x u_{-}), \text{ where } d_x u_{-} \text{ exists} \}.$$

Then:

i) $\overline{\operatorname{Graph}(\eta + du_{-})}$ is compact and is contained in the energy level $\mathcal{E}_{c}^{*} = \{(x, p) \in \operatorname{T}^{*}M : H(x, p) = \alpha(c)\};$ ii) $\Phi_{-t}^{H}\left(\overline{\operatorname{Graph}(\eta + du_{-})}\right) \subseteq \operatorname{Graph}(\eta + du_{-})$ for each t > 0;iii) $M = \pi\left(\overline{\operatorname{Graph}(\eta + du_{-})}\right),$ where $\pi : \operatorname{T}^{*}M \to M$ is the canonical projection. Moreover, let us define:

$$\mathfrak{I}^*(u_-) := \bigcap_{t \ge 0} \Phi^H_{-t} \left(\overline{\operatorname{Graph}(\eta + du_-)} \right).$$

 $\mathfrak{I}^*(u_-)$ is non-empty, compact and invariant under Φ_t^H . Furthermore, its "unstable set" contains $\overline{\operatorname{Graph}(\eta + du_-)}$; i.e.,

$$\overline{\operatorname{Graph}(\eta + du_{-})} \subseteq W^{u}\left(\mathfrak{I}^{*}(u_{-})\right) := \left\{ (y, p) : \operatorname{dist}\left(\Phi^{H}_{-t}(y, p), \mathfrak{I}^{*}(u_{-})\right) \xrightarrow{t \to +\infty} 0 \right\}.$$

There is a relation between these invariant sets $\mathfrak{I}^*(u_-)$ (or $\mathfrak{I}^*(u_+)$) and the Aubry set $\widetilde{\mathcal{A}}_c$ (recall that $\mathfrak{I}^*(u_-), \mathfrak{I}^*(u_+) \subset \mathrm{T}^*M$, while $\widetilde{\mathcal{A}}_c \subset \mathrm{T}M$):

$$\mathcal{A}_{c}^{*} := \mathcal{L}\left(\widetilde{\mathcal{A}}_{c}\right) = \bigcap_{u_{-} \in \mathcal{S}_{\eta}^{-}} \mathfrak{I}^{*}(u_{-}) = \bigcap_{u_{-} \in \mathcal{S}_{\eta}^{-}} \operatorname{Graph}(\eta + du_{-}) = \\ = \bigcap_{u_{+} \in \mathcal{S}_{\eta}^{+}} \mathfrak{I}^{*}(u_{+}) = \bigcap_{u_{+} \in \mathcal{S}_{\eta}^{+}} \operatorname{Graph}(\eta + du_{+})$$
(1.19)

where $\mathcal{L}: \mathrm{T}M \to \mathrm{T}^*M$ denotes the Legendre transform (1.4).

Since it is easier to work with subsolutions rather than weak solutions, we want to discuss now how η -critical subsolutions, although they contain less dynamical information than weak KAM solutions, can be used to characterize Aubry and Mañé sets in a similar way.

Consider $u \prec L_{\eta} + \alpha(c)$. For $t \geq 0$ define

$$\widetilde{\mathfrak{I}}_t(u) := \left\{ (x,v) \in \mathrm{T}M : \ \gamma_{(x,v)}(s) := \pi \Phi_s^L((x,v)) \text{ is } (u,L_\eta,\alpha(c)) - \text{calibr. on } (-\infty,t] \right\}.$$

We will call the Aubry set of u: $\widetilde{\mathfrak{I}}(u) := \bigcap_{t \ge 0} \widetilde{\mathfrak{I}}_t(u)$, that can be also defined as

$$\widetilde{\mathfrak{I}}(u) := \left\{ (x, v) \in \mathrm{T}M : \gamma_{(x, v)}(s) := \pi \Phi_s^L((x, v)) \text{ is } (u, L_\eta, \alpha(c)) - \text{ calibr. on } \mathbb{R} \right\}.$$

These sets $\widetilde{\mathfrak{I}}(u)$ are non-empty, compact and invariant. Moreover, here there are some properties of these sets (compare with theorem 1.4.17).

Proposition 1.4.18. Let $u \prec L_{\eta} + \alpha(c)$.

- 1. $\widetilde{\mathfrak{I}}_t(u)$ is compact;
- 2. $\widetilde{\mathfrak{I}}_{t'}(u) \subseteq \widetilde{\mathfrak{I}}_t(u) \subseteq \widetilde{\mathfrak{I}}_0(u)$ for all $t' \ge t \ge 0$;
- 3. $\mathcal{L}\left(\widetilde{\mathfrak{I}}_{0}(u)\right)$ is contained in the energy level \mathcal{E}_{c}^{*} corresponding to $\alpha(c)$;

4.
$$\mathcal{L}\left(\widetilde{\mathfrak{I}}_t(u)\right) \subseteq \operatorname{Graph}(\eta + du) \text{ for all } t > 0;$$

5. $\mathcal{L}\left(\widetilde{\mathfrak{I}}_{0}(u)\right) \subseteq \overline{\operatorname{Graph}(\eta + du)}$ (observe that for weak solutions these two sets coincide);

6.
$$\Phi_{-t}^{L}\left(\widetilde{\mathfrak{I}}_{0}(u)\right) = \widetilde{\mathfrak{I}}_{t}(u) \text{ for all } t > 0;$$

7.
$$\bigcup_{t>0} \widetilde{\mathfrak{I}}_t(u) = \widetilde{\mathfrak{I}}_0(u);$$

8.
$$\widetilde{\mathfrak{I}}(u) = \bigcap_{t \ge 0} \Phi^L_{-t} \left(\widetilde{\mathfrak{I}}_0(u) \right);$$

9.
$$\widetilde{\mathfrak{I}}_0(u) \subseteq W^u\left(\widetilde{\mathfrak{I}}(u)\right).$$

10. $\pi: \widetilde{\mathfrak{I}}(u) \longrightarrow \pi(\widetilde{\mathfrak{I}}(u))$ is a bi-Lipschitz homeomorphism [Graph Theorem]. The same is true for $\widetilde{\mathfrak{I}}_t(u)$ for each t > 0.

Theorem 1.4.19 (Fathi). The Aubry and Mañé sets defined in (1.16) and (1.11) can be equivalently defined in the following ways:

$$\widetilde{\mathcal{A}}_{c} = \bigcap_{u \prec L_{\eta} + \alpha(c)} \widetilde{\mathfrak{I}}(u) = \bigcap_{u \prec L_{\eta} + \alpha(c)} \mathcal{L}^{-1} \left(\operatorname{Graph}(\eta + du) \right)$$
$$\widetilde{\mathcal{N}}_{c} = \bigcup_{u \prec L_{\eta} + \alpha(c)} \widetilde{\mathfrak{I}}(u).$$

Moreover, there exists $u_{\infty} \prec L_{\eta} + \alpha(c)$ such that $\widetilde{\mathcal{A}}_{c} = \widetilde{\mathfrak{I}}(u_{\infty}) \cap \mathcal{L}^{-1}(\mathcal{E}_{c}^{*}).$

For the last statement is sufficient to observe that the set of critically dominated functions is a separable subset of C(M). Let $\{u_n\}$ be a countable dense family of such functions and define u_{∞} as a convex combination of their normalization (with respect to a fixed point $x_0 \in M$), e.g., $u_{\infty}(x) = \sum_{n=0}^{\infty} \frac{1}{2^n} (u_n(x) - u_n(x_0))$.

Remark 1.4.20. Using this characterization, the graph property of the Aubry set (Theorem 1.3.16) follows easily from property 10 in proposition 1.4.18. Moreover, the non-emptiness of $\tilde{\mathcal{N}}_c$ is a result of the non-emptiness of $\tilde{\mathfrak{I}}(u)$. As far as the non-emptiness of $\tilde{\mathcal{A}}_c$ is concerned, one can deduce it from this characterization and proposition 1.4.27.

From theorem 1.4.19 one can also deduce another interesting property of critically dominated functions: their differentiability on the projected Aubry set (recall that apriori these functions are only Lipschitz, so they are differentiable almost everywhere).

Proposition 1.4.21. Let $u \prec L_{\eta} + \alpha(c)$. For each $x \in \mathcal{A}_c$, u is differentiable at xand $d_x u$ does not depend on u; namely, $d_x u = \frac{\partial L}{\partial v}(x, \pi_{|_{\mathcal{A}_c}}^{-1}(x))$.

In addition to the Aubry set and Mañé set, one can also recover the definition of Peierls barrier h_{η} (see (1.13)) in terms of these functions. Let us start by observing that, from definition 1.4.2, if $u \prec L_{\eta} + \alpha(c)$ then for each $x, y \in M$ and t > 0 we have that $u(y) - u(x) \leq h_{\eta,t}(x, y) + \alpha(c)t$, where $h_{\eta,t}$ has been defined in (1.12). This implies that for each $u \prec L_{\eta} + \alpha(c)$ and for each $x, y \in M$, $u(y) - u(x) \leq h_{\eta}(x, y)$, *i.e.*,

$$h_{\eta}(x,y) \ge \sup_{u \prec L_{\eta} + \alpha(c)} [u(y) - u(x)] \qquad \forall x, y \in M.$$

Moreover, if $u_{-} \in S_{\eta}^{-}$ and $u_{+} \in S_{\eta}^{+}$ are conjugate solutions, the same result holds: $u_{-}(y) - u_{+}(x) \leq h_{\eta}(x, y)$ and consequently

$$h_{\eta}(x,y) \ge \sup_{(u_{-},u_{+})} [u_{-}(y) - u_{+}(x)] \qquad \forall x, y \in M,$$

where (u_{-}, u_{+}) denotes conjugate weak KAM solutions. In addition to this, it is possible to show that the above inequality is actually an equality. In fact:

Proposition 1.4.22 (Fathi). For $x \in M$ let us define the function $h_{\eta}^{x} : M \to \mathbb{R}$ (resp. $h_{\eta,x} : M \to \mathbb{R}$) by $h_{\eta}^{x}(y) = h_{\eta}(x, y)$ (resp. $h_{\eta,x}(y) = h_{\eta}(y, x)$). For each $x \in M$, the function h_{η}^{x} (resp. $-h_{\eta,x}$) is a weak KAM solution of negative (resp. positive) type. Moreover, its conjugate function $u_{+}^{x} \in \mathcal{S}_{\eta}^{+}$ (resp. $u_{-}^{x} \in \mathcal{S}_{\eta}^{-}$) vanishes at x.

As a consequence of this:

Corollary 1.4.23. For each $x, y \in M$, we have the equality

$$h_{\eta}(x,y) = \sup_{(u_{-},u_{+})} \left[u_{-}(y) - u_{+}(x) \right],$$

where the supremum is taken over pairs of conjugate solutions. Moreover, for any given $x, y \in M$ this supremum is actually attained.

Observe that, since for any $u \prec L_{\eta} + \alpha(c)$ there exists a weak KAM solution of negative type u_{-} and a weak KAM solution of positive type u_{+} , such that $u_{-} = u = u_{+}$ on the projected Aubry set \mathcal{A}_{c} (see Theorem 1.4.15), one can get the following representations for Peierls barrier h_{η} on the projected Aubry set:

$$h_{\eta}(x,y) = \sup_{u \prec L_{\eta} + \alpha(c)} [u(y) - u(x)]$$
 (1.20)

for all $x, y \in \mathcal{A}_c$. This supremum is actually attained for any fixed $x \in \mathcal{A}_c$.

Moreover, a similar representation also holds for Mather's pseudodistance δ_c (see (1.14)). In fact, from the definition of $\delta_c(x, y)$ we immediately get:

$$\delta_{c}(x,y) = h_{\eta}(x,y) + h_{\eta}(y,x) =$$

$$= \sup_{u \prec L_{\eta} + \alpha(c)} (u(y) - u(x)) + \sup_{v \prec L_{\eta} + \alpha(c)} (v(x) - v(y)) =$$

$$= \sup_{u,v \prec L_{\eta} + \alpha(c)} [(u(y) - v(y)) - (u(x) - v(x))]$$
(1.21)

for all $x, y \in \mathcal{A}_c$. This supremum is also attained for any fixed $x, y \in \mathcal{A}_c$.

ADDENDA

In these addenda we want to say more about the regularity of critically dominated functions or η -critical subsolutions and prove a result about the recurrent points of the Euler-Lagrange flow restricted to the Mañé set that we mentioned in relation with theorem 1.3.17.

Regularity of critical subsolutions

We have remarked above in this section, that for $k < \alpha(c)$ there do not exist functions dominated by $L_{\eta} + k$, while for $k \ge \alpha(c)$ they do exist. Moreover, if $k > \alpha(c)$ these functions can be chosen to be C^{∞} (see also characterization of $\alpha(c)$ in remark 1.4.7). The critical case has totally different features. As a counterpart of their relation with the dynamics of the system, critical dominated functions have very rigid structural properties, that become an obstacle when someone tries to make them smoother. For instance, as we have recalled in proposition 1.4.21, if $u \prec L_{\eta} + \alpha(c)$ then its differential $d_x u$ exists on \mathcal{A}_c and it is prescribed over there. This means that although it is quite easy to make these functions smoother $(e.g., C^{\infty})$ out of the projected Aubry set, it is impossible to modify them on this set.

Nevertheless, Albert Fathi and Antonio Siconolfi [37] managed to prove that C^1 η -critical subsolutions do exist and are dense, in the following sense:

Theorem 1.4.24 (Fathi, Siconolfi). Let $u \prec L_{\eta} + \alpha(c)$. For each $\varepsilon > 0$, there exists a C^1 function $\tilde{u} : M \longrightarrow \mathbb{R}$ such that:

i) $\tilde{u} \prec L_{\eta} + \alpha(c);$

- ii) $\tilde{u}(x) = u(x)$ on \mathcal{A}_c ;
- *iii)* $|\tilde{u}(x) u(x)| < \varepsilon$ on $M \setminus \mathcal{A}_c$.

Moreover, one can choose \tilde{u} so that it is a strict η -critical subsolution, i.e., we have $H_{\eta}(x, d_x \tilde{u}) < a(c) \text{ on } M \setminus \mathcal{A}_c.$

Remark 1.4.25. This result has been extended by Patrick Bernard [13], who showed that every η -critical subsolution coincides, on the Aubry set, with a $C^{1,1}$ η -critical subsolution. In general, this is the best regularity that one can expect: it is easy in fact to construct examples in which C^2 η -critical subsolutions do not exist. For example, consider the case in which the Aubry set projects over all the manifold M and it is not a C^1 graph (e.g., on $M = \mathbb{T}$ take $L(x, v) = \frac{1}{2} ||v||^2 + \sin^2(\pi x)$ and $\eta = \frac{2}{\pi} dx$). In this case there is only one critical subsolution (up to constants), that is an actual solution: its differential is Lipschitz but not C^1 .

It is therefore clear that the structure of the Aubry set plays a crucial role. Patrick Bernard [14] proved that if the Aubry set is a union of finitely many hyperbolic periodic orbits or hyperbolic fixed points, then smoother subsolutions can be constructed. In particular, if the Hamiltonian is C^k , then these subsolutions will be C^k too. The proof of this result is heavily based on the hyperbolic structure of the Aubry set and the result is deduced from the regularity of its local stable and unstable manifolds.

Using the density of C^1 critically dominated functions, one can emprove some of the results in 1.19, theorem 1.4.19, (1.20) and (1.21). Let us denote by S^1_{η} the set of $C^1 \eta$ -critical subsolutions and $S^{1,1}_{\eta}$ the set of $C^{1,1} \eta$ -critical subsolutions. Then:

$$\widetilde{\mathcal{A}}_{c} := \bigcap_{u \in \mathcal{S}_{\eta}^{1}} \widetilde{\mathfrak{I}}(u) = \bigcap_{u \in \mathcal{S}_{\eta}^{1,1}} \widetilde{\mathfrak{I}}(u) = \\ = \bigcap_{u \in \mathcal{S}_{\eta}^{1}} \mathcal{L}^{-1} \left(\operatorname{Graph}(\eta + du) \right) = \bigcap_{u \in \mathcal{S}_{\eta}^{1,1}} \mathcal{L}^{-1} \left(\operatorname{Graph}(\eta + du) \right). \quad (1.22)$$

In particular, there exists a $C^{1,1}$ $\eta\text{-critical subsolution}~\tilde{u}$ such that:

$$\widetilde{\mathcal{A}}_{c} = \mathcal{L}^{-1} \left(\operatorname{Graph}(\eta + d\widetilde{u}) \cap \mathcal{E}_{c}^{*} \right) =$$
$$= \widetilde{\mathfrak{I}}(\widetilde{u}) \cap \mathcal{L}^{-1} \left(\mathcal{E}_{c}^{*} \right).$$
(1.23)

Moreover, for any $x, y \in \mathcal{A}_c$:

$$h_{\eta}(x,y) = \sup_{u \in \mathcal{S}_{\eta}^{1}} \{u(y) - u(x)\} = \sup_{u \in \mathcal{S}_{\eta}^{1,1}} \{u(y) - u(x)\}$$
(1.24)
$$\delta_{c}(x,y) = \sup_{u,v \in \mathcal{S}_{\eta}^{1}} \{(u-v)(y) - (u-v)(x)\} =$$

$$= \sup_{u,v \in \mathcal{S}_{\eta}^{1,1}} \{(u-v)(y) - (u-v)(x)\},$$
(1.25)

where all the above suprema are maxima for any fixed $x, y \in \mathcal{A}_c$.

Non-wandering points of the Mañé set

We want now to show that the non-wandering set of the Euler-Lagrange flow restricted to the Mañé set is contained in the Aubry set. We have already recalled this result for sketching a proof of the second equivalence in theorem 1.3.17. Let us first recall the definition of *non-wandering point* for a flow $\Phi_t : X \longrightarrow X$.

Definition 1.4.26. A point $x \in X$ is called non-wandering if for each neighborhood \mathcal{U} and each positive integer n, there exists t > n such that $f^t(\mathcal{U}) \cap \mathcal{U} \neq \emptyset$.

We will denote the set of *non-wandering* points for Φ_t by $\Omega(\Phi_t)$.

Proposition 1.4.27. If M is a compact manifold and L a Tonelli Lagrangian on TM, then $\Omega\left(\Phi_t^L|\widetilde{\mathcal{N}}_c\right) \subseteq \widetilde{\mathcal{A}}_c$ for each $c \in H^1(M; \mathbb{R})$.

Remarks.

1) Proposition 1.4.27 also shows that the Aubry set is non-empty. In fact, any continuous flow on a compact space possesses non-wandering points.

2) Since every point in the support of an invariant measure μ is non-wandering, then this also shows that $\widetilde{\mathcal{M}}_c \subseteq \widetilde{\mathcal{A}}_c$.

Proof. Let $(x, v) \in \Omega\left(\Phi_t^L | \widetilde{\mathcal{N}}_0\right)$. By the definition of non-wandering point, there exist a sequence $(x_k, v_k) \in \widetilde{\mathcal{N}}_c$ and $t_k \to +\infty$, such that $(x_k, v_k) \to (x, v)$ and $\Phi_{t_k}^L(x_k, v_k) \to (x, v)$ as $k \to +\infty$. From (1.22), for each (x_k, v_k) there exists a η critical subsolution u_k , such that the curve $\gamma_k(t) = \pi\left(\Phi_t^L(x_k, v_k)\right)$ is $(u_k, L_\eta, \alpha(c))$ - calibrated. Moreover, up to extracting a subsequence, we can assume that, on any compact interval, γ_k converge in the C^1 -topology to $\gamma(t) = \pi\left(\Phi_t^L(x, v)\right)$.

Pick now any critical subsolution u. If we show that γ is $(u, L_{\eta}, \alpha(c))$ - calibrated, using (1.22) we can conclude that $(x, v) \in \widetilde{\mathcal{A}}_c$. First of all, observe that, by the continuity of u,

$$u(\gamma_k(t_k)) - u(x_k) \xrightarrow{k \to \infty} 0$$
.

Using that η -critical subsolutions are equi-Lipschitz (as remarked after definition 1.4.2), we can also conclude that

$$u_k(\gamma_k(t_k)) - u_k(x_k) \stackrel{k \to \infty}{\longrightarrow} 0$$

and, therefore,

$$\int_0^{t_k} L_\eta(\gamma_k(s), \dot{\gamma}_k(s)) + \alpha(c) \, ds = u_k(\gamma_k(t_k)) - u_k(x_k) \xrightarrow{k \to \infty} 0 \,. \tag{1.26}$$

Let $0 \le a \le b$ and choose $t_k \ge b$. Observe now that

$$\begin{split} u(\gamma_{k}(b)) - u(\gamma_{k}(a)) &= u(\gamma_{k}(t_{k})) - u(x_{k}) - [u(\gamma_{k}(t_{k})) - u(\gamma_{k}(b))] - \\ &- [u(\gamma_{k}(a)) - u(x_{k})] \ge \\ &\ge u(\gamma_{k}(t_{k})) - u(x_{k}) - \int_{b}^{t_{k}} L_{\eta}(\gamma_{k}(s), \dot{\gamma}_{k}(s)) + \alpha(c) \, ds - \\ &- \int_{0}^{a} L_{\eta}(\gamma_{k}(s), \dot{\gamma}_{k}(s)) + \alpha(c) \, ds = \\ &= u(\gamma_{k}(t_{k})) - u(x_{k}) + \int_{a}^{b} L_{\eta}(\gamma_{k}(s), \dot{\gamma}_{k}(s)) + \alpha(c) \, ds - \\ &- \int_{0}^{t_{k}} L_{\eta}(\gamma_{k}(s), \dot{\gamma}_{k}(s)) + \alpha(c) \, ds ; \end{split}$$

taking the limit as $k \to \infty$ on both sides, one can conclude:

$$u(\gamma(b)) - u(\gamma(a)) \ge \int_{a}^{b} L_{\eta}(\gamma(s), \dot{\gamma}(s)) + \alpha(c) \, ds$$

and therefore, from the fact that $u \prec L_{\eta} + \alpha(c)$, it follows the equality. This shows that γ is $(u, L_{\eta}, \alpha(c))$ -calibrated on $[0, \infty)$. To show that it is indeed calibrated on all \mathbb{R} , one can make a symmetric argument, letting $(y_k, w_k) = \Phi_{t_k}^L(x_k, v_k)$ play the role of (x_k, v_k) in the previous argument. In fact, one has $(y_k, w_k) \to (x, v)$ and $\Phi_{-t_k}^L(y_k, w_k) \to (x, v)$ as $k \to +\infty$ and the very same argument works.

Chapter 2

On the structure of the quotient Aubry set

2.1 The quotient Aubry set

In John Mather's studies on the dynamics of Lagrangian systems and the existence of Arnold diffusion, it turns out to be useful to understand certain aspects of the *Aubry* set and, in particular, what is called the *quotient Aubry set*. In fact, in many variational methods that have been developed for constructing orbits with prescribed behaviors [11, 12, 25, 61, 63, 65, 78], the structure of this set plays a crucial role. In this chapter, we want to discuss some its topological properties, in particular its total disconnectedness [75].

Let us start this section by recalling the definition of the quotient Aubry set (see also remark 1.3.10). The setting will be the same as in chapter 1: let M be a compact and connected smooth manifold without boundary, $L : TM \longrightarrow \mathbb{R}$ a Tonelli Lagrangian (definition 1.1.1) and $H : T^*M \longrightarrow \mathbb{R}$ the associated optical Hamiltonian (definition 1.1.3). Moreover, for each $c \in H^1(M; \mathbb{R})$ and η closed 1-form with cohomology class $[\eta] = c$, we will denote L_{η} and H_{η} the corresponding modified Lagrangian and Hamiltonian (see remark 1.2.1).

In (1.15) we have defined the projected Aubry set (with cohomology c), as

$$\mathcal{A}_{c} = \{ x \in M : \delta_{c}(x, x) = 0 \} = \{ x \in M : h_{\eta}(x, x) = 0 \},\$$

where h_{η} is *Peierls Barrier* (see (1.13)) and δ_c its symmetrization, also called *Mather's* pseudometric (see (1.14)).

Definition 2.1.1 (Quotient Aubry set). The quotient Aubry set $(\bar{\mathcal{A}}_c, \bar{\delta}_c)$ is the metric space obtained by identifying two points in \mathcal{A}_c if their δ_c -pseudodistance is zero.

We will denote an element of this quotient by $\bar{x} = \{y \in \mathcal{A}_c : \delta_c(x, y) = 0\}$. These elements, that are also called *c-static classes* (see [23]), provide a partition of \mathcal{A}_c into compact subsets, that can be lifted to invariant subsets of T*M*. They are really interesting from a dynamical systems point of view, since they contain the α and ω limit sets of *c*-minimizing orbits. Using this definition, one can rephrase definition 1.3.14 in the following way: $\gamma : \mathbb{R} \longrightarrow M$ is a *c*-regular minimizer if it is a *c*-minimizer and the α and ω -limit sets are both contained in the same *c*-static class (*i.e.*, in the same element of this quotient set). In fact, if $\gamma : \mathbb{R} \longrightarrow M$ is a *c*-minimizer, then its α and ω -limit sets are each contained in a *c*-static class, which might be a-priori different (see remark 1.3.13).

Remark 2.1.2. In general one could ask if given two different *c*-static classes $\overline{x_{\alpha}}$ and $\overline{x_{\omega}}$, there exists a *c*-minimizer "connecting" them, *i.e.*, such that its α -limit set is contained in $\overline{x_{\alpha}}$ and its ω -limit set is contained in $\overline{x_{\omega}}$. In general, this is not true. Consider, for instance, a mechanical Lagrangian on the 1-dimensional torus: in this case the Aubry set is made of equilibria points (exactly the global minima of the potential) and it is definitely not true that there exist heteroclinic connections among any two of them.

A more correct way to formulate this question might be the following. Define a reflexive partial order \leq in $\overline{\mathcal{A}}_c$ by: a) \leq is reflexive; b) \leq is transitive; c) if there exists a *c*-minimizer γ connecting $\overline{x_{\alpha}}$ to $\overline{x_{\omega}}$, then $\overline{x_{\alpha}} \leq \overline{x_{\omega}}$. Gonzalo Contreras and Gabriel Paternain [25] proved that if the quotient Aubry set is finite, then for any given $\overline{x}, \overline{y} \in \overline{\mathcal{A}}_c$ we have that $\overline{x} \leq \overline{y}$. In other words, there exist $\overline{x_1} = \overline{x}, \ldots, \overline{x_n} = \overline{x}$ and *c*-minimizers $\gamma_1, \ldots, \gamma_{n-1}$ such that for each $k = 1 \ldots, n$ the α -limit set of γ_k is contained in $\overline{x_k}$ and its ω -limit set in $\overline{x_{k+1}}$. Therefore, between any two given static classes there exists a chain of *c*-minimizers connecting them.

John Mather in [64] proved a more general statement (it was proved in the the nonautonomous case). If $\overline{x}, \overline{y} \in \overline{A}_c$ and $\overline{x} \neq \overline{y}$, then there exists a subset $\mathfrak{S} = \mathfrak{S}_{\overline{x},\overline{y}} \subseteq \overline{A}_c$ and an order \leq on \mathfrak{S} such that: i) \overline{x} is the least and \overline{y} the greatest element of \mathfrak{S} ; ii) the topology of \mathfrak{S} associated to the metric $\overline{\delta}_c$ is the same as the topology associated to the order; iii) \mathfrak{S} is compact with respect to this topology; iv) if $\overline{x_1}, \overline{x_2} \in \mathfrak{S}$ and $\overline{x_1} \leq \overline{x_2}$, then there exists a *c*-minimizer $\gamma_{\overline{x_1}, \overline{x_2}}$ such that its α -limit set is contained in $\overline{x_1}$ and its ω -limit set in $\overline{x_2}$.

While in the case of twist maps (see for instance [8, 38] and references therein) there is a detailed structure theory for these sets, in more degrees of freedom quite few is known. In particular, it seems to be useful to understand whether the quotient Aubry set is "small" in some sense of dimension (*e.g.*, vanishing topological or Hausdorff dimension).

In [62] Mather showed that if the state space has dimension ≤ 2 (in the nonautonomous case) or the Lagrangian is the kinetic energy associated to a Riemannian metric and the state space has dimension ≤ 3 , then the quotient Aubry set is totally disconnected, *i.e.*, every connected component consists of a single point (in a compact metric space this is equivalent to vanishing topological dimension). In the autonomous case, with dim $M \leq 3$, the same argument shows that this quotient is totally disconnected as long as the Aubry set does not intersect the zero section of TM (this is the case when the cohomology class is large enough in norm). See also the addendum to section 2.2.

What happens in higher dimension? Unfortunately, this is generally not true. In fact, Burago, Ivanov and Kleiner in [18] provided an example that does not satisfy this property (they did not discuss it in their work, but it follows from the results therein). More strikingly, Mather provided in [64] several examples of quotient Aubry sets that are not only non-totally-disconnected, but even isometric to closed intervals. All these examples come from mechanical Lagrangians on $T\mathbb{T}^d$ (*i.e.*, the sum of the kinetic energy and a potential) with $d \geq 3$. In particular, for every $\varepsilon > 0$, he provided a potential $U \in C^{2d-3,1-\varepsilon}(\mathbb{T}^d)$, whose associated quotient Aubry set is isometric to an interval. As the author himself noticed, it is not possible to improve the differentiability of these examples, due to the construction carried out. See the addendum to section 2.2 for such counterexamples. It is interesting to notice that Mather's construction is quite similar, in spirit, to the one used by Whitney for constructing a counterexample to Sard's lemma [77]. We will see in section 2.2 that this relation is not accidental at all and will represent one of the key ingredients in the proof of our main result.

In the following sections we will discuss the problem of the total disconnectedness of $\bar{\mathcal{A}}_c$ in higher dimension and prove it under some extra assumptions on the Lagrangian. In particular, it will follow from our main result that the above counterexamples are optimal, in the sense that for more regular mechanical Lagrangians the quotient Aubry set - corresponding to the zero cohomology class - is totally disconnected.

2.2 Sard's Lemma and the quotient Aubry set

In this section we want to discuss the relation between Sard's lemma and the total disconnectedness of the quotient Aubry set. In order to make it clear, it is convenient to adopt weak KAM theory's point of view, introduced in section 1.4. In the following, we will denote by S_{η} the set of η -critical subsolutions (or critically dominated functions, see definition 1.4.2 and proposition 1.4.3), by S_{η}^{1} the set of C^{1} η -critical subsolutions. We have already remarked in section 1.4 and its addendum, that these sets are non-empty and in particular S_{η}^{1} and $S_{\eta}^{1,1}$ are both "dense" in S_{η} (see theorem 1.4.24 and [13, 37] for the exact statements). We also showed how to get an expression for Peierls Barrier h_{η} and Mather's pseudometric δ_{c} in terms of these functions (see (1.20), (1.21) and in particular (1.24) and (1.25)).

It turns out that is convenient to characterize the elements of $\bar{\mathcal{A}}_c$ (*i.e.*, the *c*-quotient classes) in terms of η -critical subsolutions. Let us consider the following set:

$$\mathcal{D}_c = \{ u - v : \ u, \ v \in \mathcal{S}_\eta \}; \tag{2.1}$$

observe that it depends only on the cohomology class c and not on η (differently from the set of critical subsolutions). Similarly, denote by \mathcal{D}_c^1 and $\mathcal{D}_c^{1,1}$, the sets corresponding, respectively, to the difference of C^1 and $C^{1,1}$ η -critical subsolutions. With this new notation (1.25) becomes:

$$\delta_{c}(x,y) = \sup_{w \in \mathcal{D}_{c}} (w(y) - w(x)) = \sup_{w \in \mathcal{D}_{c}^{1}} (w(y) - w(x)) = = \sup_{w \in \mathcal{D}_{c}^{1,1}} (w(y) - w(x))$$
(2.2)

for any $x, y \in \mathcal{A}_c$. Moreover, this suprema are achieved for any fixed $x, y \in \mathcal{A}_c$.

Let us start to study some properties of these functions.

Proposition 2.2.1. If $w \in \mathcal{D}_c$, then $d_x w = 0$ on \mathcal{A}_c . Therefore $\mathcal{A}_c \subseteq \bigcap_{w \in \mathcal{D}_c^{1,1}} \operatorname{Crit}(w)$, where $\operatorname{Crit}(w)$ is the set of critical points of w.

Proof. This is an immediate consequence of proposition 1.4.21; namely, if $u, v \in S_{\eta}$, then they are differentiable on \mathcal{A}_c and $d_x u = d_x v$.

Remark 2.2.2. It is actually possible to show that $\mathcal{A}_c = \bigcap_{w \in \mathcal{D}_c^{1,1}} \operatorname{Crit}(w)$.

Proposition 2.2.3. If $w \in \mathcal{D}_c$, then it is constant on any quotient class of $\bar{\mathcal{A}}_c$; namely, if $x, y \in \mathcal{A}_c$ and $\delta_c(x, y) = 0$, then w(x) = w(y).

Proof. It follows easily from (2.2). In fact:

$$0 = \delta_c(x, y) = \sup_{\tilde{w} \in \mathcal{D}_c} (\tilde{w}(y) - \tilde{w}(x)) \ge w(y) - w(x)$$

$$0 = \delta_c(y, x) = \sup_{\tilde{w} \in \mathcal{D}_c} (\tilde{w}(x) - \tilde{w}(y)) \ge w(x) - w(y).$$

For any $w \in \mathcal{D}_c^1$, let us define the following *evaluation function*:

$$\varphi_w : (\bar{\mathcal{A}}_c, \, \bar{\delta}_c) \longrightarrow (\mathbb{R}, \, |\cdot|)$$
$$\bar{x} \longmapsto w(x) \, .$$

- φ_w is well defined, *i.e.*, it does not depend on the element of the class at which w is evaluated;
- $\varphi_w(\bar{\mathcal{A}}_c) = w(\mathcal{A}_c) \subseteq w(\operatorname{Crit}(w));$
- φ_w is Lipschitz, with Lipschitz constant 1. In fact:

$$\varphi_w(\bar{x}) - \varphi_w(\bar{y}) = w(x) - w(y) \le \delta_c(x, y) = \bar{\delta}_c(\bar{x}, \bar{y})$$
$$\varphi_w(\bar{y}) - \varphi_w(\bar{x}) = w(y) - w(x) \le \delta_c(y, x) = \bar{\delta}_c(\bar{y}, \bar{x}).$$

Therefore:

$$|\varphi_w(\bar{x}) - \varphi_w(\bar{y})| \le \bar{\delta}_c(\bar{x}, \bar{y}) \,.$$

Remark 2.2.4. In some sense, (2.2) corresponds to reconstruct a metric knowing its 1-Lipschitz functions (*i.e.*, Lipschitz functions with Lipschitz constant equal to 1).

Now the relation with Sard's lemma can be stated more clearly.

Definition 2.2.5. A C^1 function $f : M \longrightarrow \mathbb{R}$ is of Morse-Sard type if $|f(\operatorname{Crit}(f))| = 0$, where $\operatorname{Crit}(f)$ is the set of critical points of f and $|\cdot|$ denotes the Lebesgue measure on \mathbb{R} .

Proposition 2.2.6. Let M be a compact connected manifold of dimension $d \ge 1$, La Tonelli Lagrangian and $c \in H^1(M; \mathbb{R})$. If each $w \in \mathcal{D}_c^{1,1}$ is of Morse-Sard type, then the quotient Aubry set $(\bar{\mathcal{A}}_c, \bar{\delta}_c)$ is totally disconnected.

Proof. Suppose by contradiction that $\overline{\mathcal{A}}_c$ is not totally disconnected; therefore it must contain a connected component $\overline{\Gamma}$ with at least two points \overline{x} and \overline{y} . In particular $\overline{\delta}_c(\overline{x}, \overline{y}) > 0$ for some $x \in \overline{x}$ and $y \in \overline{y}$. From the representation formula for δ_c , it follows that there exists $w \in \mathcal{D}_c^{1,1}$, such that $w(x) \neq w(y)$. This implies that the set $\varphi_w(\overline{\Gamma})$ is a connected set in \mathbb{R} with at least two different points, hence it is a non-degenerate interval and its Lebesgue measure is positive. But

$$\varphi_w(\overline{\Gamma}) \subseteq \varphi_w(\overline{\mathcal{A}}_c) = w(\mathcal{A}_c) \subseteq w(\operatorname{Crit}(w))$$

and we get a contradiction

$$0 < |\varphi_w(\overline{\Gamma})| \le |w(\mathcal{A}_c)| \le |w(\operatorname{Crit}(w))| = 0.$$

Remark 2.2.7. It is clear from the proof that it would be sufficient that Morse-Sard's property held for a dense subset of $\mathcal{D}_c^{1,1}$. Moreover, one could ask w to satisfy only the weaker condition $|w(\mathcal{A}_c)| = 0$.

This proposition and Sard's lemma easily prove total disconnectedness in dimension $d \leq 2$ (see also the addendum at the end of this section and [62], where the non-autonomous case was considered). In fact, it suffices to notice that Sard's lemma in dimension d holds for $C^{d-1,1}$ functions [10].

Corollary 2.2.8. Let M be a compact connected manifold of dimension $d \leq 2$. For any L Tonelli Lagrangian and $c \in H^1(M; \mathbb{R})$, the quotient Aubry set $(\bar{\mathcal{A}}_c, \bar{\delta}_c)$ is totally disconnected.

The main problem becomes now to understand under which conditions on L and c, these differences of subsolutions are of Morse-Sard type. Unfortunately, one cannot use the classical Sard's lemma, due to a lack of regularity of critical subsolutions. As we have already discussed in remark 1.4.25, in general the best regularity one can get is $C^{1,1}$. In fact, although it is always possible to smooth them up out of the Aubry set and obtain functions in $C^{\infty}(M \setminus \mathcal{A}_c) \cap C^{1,1}(M)$, the presence of the Aubry set (where the value of their differential is prescribed) represents an obstacle that it is impossible to overcome. On the other hand, in the proof of Sard's lemma regularity is used to gain a control on the behavior of the function near the critical set, *i.e.*, to control the *complexity* (à la Yomdin) of the function near this set. In absence of such a condition, one can try to resort to some other properties of the function that may allow one to get a similar control. In our case, these functions are related to Hamilton-Jacobi equation and we will see how the smoothness of the Hamiltonian, rather than the regularity of the subsolutions, may provide for our needs. There are several difficulties involved in pursuing such an approach in the general case, mostly related to the nature of the Aubry set; at any rate, we will see in the next section that this becomes possible with some extra assumptions on the Lagrangian.

ADDENDA

Total disconnectedness in low dimension

In this addendum we want to recall Mather's result in low dimension [60, 62].

Proposition 2.2.9.

- i) If M is a compact manifold of dimension ≤ 2 and $L : \mathbb{T}M \times \mathbb{T} \longrightarrow \mathbb{R}$ is a Tonelli Lagrangian, then $\bar{\mathcal{A}}_c$ is totally disconnected for all $c \in \mathrm{H}^1(M; \mathbb{R})$.
- ii) If M is a compact manifold of dimension ≤ 3 and $L : TM \longrightarrow \mathbb{R}$ is a Tonelli Lagrangian, such that $\widetilde{\mathcal{A}}_c$ does not intersect the zero section of TM, then $\overline{\mathcal{A}}_c$ is totally disconnected.
- iii) If L is the kinetic energy associated to a Riemannian metric on M and dim $M \leq 3$, then $\bar{\mathcal{A}}_c$ is totally disconnected for all $c \in \mathrm{H}^1(M; \mathbb{R})$.

In the following we want to give an idea of the proof of such results. One of the key ingredient in the proof will be the following result by John Mather (we refer to [60] for a proof).

Proposition 2.2.10. Let M be a smooth finite dimensional manifold with a Riemannian metric g and let X be a compact connected subset of M that admits a Lipschitz lamination¹ of codimension $d \ge 2$. Let $x, y \in X$ and $\varepsilon > 0$; then, there exists a sequence $x = x_0, \ldots, x_k = y$ of points in X such that $\sum_{i=0}^{k-1} \delta_M(x_i, x_{i+1})^d < \varepsilon$, where δ_M is the metric associated to the Riemannian metric g.

¹By Lipschitz lamination we mean an S-smooth lamination, in the sense of Thurston, where S is the class of bi-Lipschitz homeomorphisms; see for instance [60] for a more precise definition.

Proof (Proposition 2.2.9). Let us start recalling that from the graph property of the Aubry set (see theorem 1.3.16 or [59]), it follows that \mathcal{A}_c has a 1-dimensional Lipschitz lamination, which is defined by the flow on it, generated by the vector field $\pi^{-1}: \mathcal{A}_c \longrightarrow \widetilde{\mathcal{A}}_c$. Moreover - see remark 1.3.12 - recall that

$$\delta_c((x,\tau_0),(y,\tau_1)) \le C[d(x,y) + \|\tau_1 - \tau_0\|]^2$$
(2.3)

for each $(x, \tau_0), (y, \tau_1) \in \mathcal{A}_c$, where

$$\|\tau_1 - \tau_0\| = \inf \{ |t_1 - t_0| : t_i \in \mathbb{R}, t_i \equiv \tau_i \pmod{1}, i = 0, 1 \}.$$

i) If dim $M \leq 2$, the 1-dimensional lamination of \mathcal{A}_c is of codimension ≤ 2 . Hence, applying proposition 2.2.10 with d = 2, we get that if $\overline{(x,\tau)}$ and $\overline{(x',\tau')}$ are two points in the same connected component of $\overline{\mathcal{A}}_c$ and $\varepsilon > 0$, then there is a finite sequence of points $(x_0, \tau_0) = (x, \tau), \ldots, (x_k, \tau_k) = (x', \tau')$ in that connected component such that

$$\sum_{i=0}^{k-1} [d(x_i, x_{i+1}) + \|\tau_{i+1} - \tau_i\|]^2 < \varepsilon.$$

Therefore, using (2.3) and the triangle inequality for δ_c , we get:

$$\delta_c((x,\tau),(x',\tau')) \leq \sum_{i=0}^{k-1} \delta_c((x_i,\tau_i),(x_{i+1},\tau_{i+1})) \leq \\ \leq C \sum_{i=0}^{k-1} [d(x_i,x_{i+1}) + \|\tau_{i+1} - \tau_i\|]^2 < C\varepsilon$$

Since $\varepsilon > 0$ is arbitrary, it follows that $\delta_c((x, \tau), (x', \tau')) = 0$; this means that each connected component of $\overline{\mathcal{A}}_c$ is reduced to a single point.

ii) In the autonomous case, the argument used in *i*) to show that \mathcal{A}_c admits a

1-dimensional Lipschitz lamination still applies, provided that this section vanishes nowhere. This means that $\widetilde{\mathcal{A}}_c$ does not intersect the zero section of TM.

iii) In the case that L is the kinetic energy associated to a Riemannian metric on M and $c \neq 0$, it is obvious that $\widetilde{\mathcal{A}}_c$ does not intersect the zero section of TM. In this case \mathcal{A}_c admits a 1-dimensional Lipschitz lamination and, since dim $M \leq 3$, this lamination is of codimension 2. Hence, we can proceed just as in the proof of i).

In the case c = 0, the result is trivial since $\mathcal{A}_c = M$ and $\bar{\mathcal{A}}_c$ is one point.

Counterexamples in high dimension

As we have recalled in section 2.1, in [64] John Mather has constructed several examples of quotient Aubry sets that are isometric to closed intervals, therefore they are not totally disconnected. All these examples consist of mechanical Lagrangians on $T\mathbb{T}^d$ (*i.e.*, the sum of the kinetic energy and a potential) with $d \geq 3$. In particular, for every $\varepsilon > 0$, he provided a potential $U \in C^{2d-3,1-\varepsilon}(\mathbb{T}^d)$, whose associated quotient Aubry set is isometric to an interval. The basic strategy consists first in defining an appropriate connected compact subset $Z \subset \mathbb{R}^d$ on which the potential will attain its global minimum (it will then coincide with the Aubry set \mathcal{A}_0), and then constructing a function U that satisfies some suitable conditions. This construction is very technical and we skip all details, for which we refer to [64]. It is interesting to notice that Mather's construction is quite similar, in spirit, to the one used by Whitney for constructing a counterexample to Sard's lemma [77].

Using this relation with Whitney's counterexample, Albert Fathi explained in [31] how to construct similar counterexamples, but less technically involved. We think that it might be useful to reproduce his unpublished construction here. Let us start by recalling a result by Patrice Assouad [4] about the bi-Lipschitz embedding of compact metric spaces into an Euclidean space (see also [39] where this result is applied to construct a counterexample to Sard's lemma).

Definition 2.2.11. Let X be a topological space and δ a metric on it. (X, δ) is said to admit a bi-Lipschitz embedding to \mathbb{R}^N , if there exists a mapping $\Psi : X \to \mathbb{R}^N$ such that

$$C_1\delta(x,y) \le \|\Psi(x) - \Psi(y)\| \le C_2\delta(x,y)$$

for all $x, y \in X$ with some fixed constants $C_1, C_2 > 0$.

For instance, any subset of an Euclidean space admits a bi-Lipschitz embedding; moreover, this property is clearly invariant under bi-Lipschitz mappings. A difficult open problem consists in characterizing which metric spaces admit such an embedding.

Definition 2.2.12 (Doubling metric spaces). Let X be a topological space and δ a metric on it. (X, δ) is called doubling if there exists $K \in \mathbb{N}$ such that any ball $B(x, \varepsilon)$ is covered by at most K balls of radius $\frac{\varepsilon}{2}$, for all $\varepsilon > 0$.

Every subset of the Euclidean space is doubling and also the doubling condition is invariant under bi-Lipschitz mappings. From this, it follows immediately that a necessary condition for (X, δ) to be bi-Lipschitz embeddable to an Euclidean space, is that (X, δ) is doubling. Unfortunately the doubling condition is not sufficient [41]. However, the following result by Patrice Assouad shows that every doubling metric space admits an "almost" bi-Lipschitz embedding.

Theorem 2.2.13 (Assouad, [4]). Let (X, δ) be a doubling metric space. Then, for every 0 < s < 1 there exist N and a bi-Lipschitz embedding $\Psi : (X, \delta^s) \to \mathbb{R}^N$, i.e., a mapping such that

$$C_1\delta(x,y)^s \le \|\Psi(x) - \Psi(y)\| \le C_2\delta(x,y)^s$$

for some constants $C_1, C_2 > 0$, depending on s, and for all $x, y \in X$.

Let us see now how to relate this result to the counterexamples mentioned above.

Theorem 2.2.14 (Fathi). Let $r \ge 2$ and (X, δ) a doubling metric space. Then, there exist $N \in \mathbb{N}$ sufficiently large and a C^r mechanical Lagrangian $L : \mathbb{TT}^N \to \mathbb{R}$ such that the quotient Aubry set $(\bar{\mathcal{A}}_0, \bar{\delta}_0)$ is bi-Lipschitz equivalent to (X, δ) .

Remark 2.2.15. This construction, contrary to the one in [64], does not provide a clear relation between r and N. It will be only clear that N goes to $+\infty$, as rincreases, as it must be because of theorem 2.3.1 (see section 2.3).

Let us try to give a sketch of the proof of such a result.

Proof. From theorem 2.2.13, it follows that for each p > 1 there exist N and a bi-Lipschitz embedding $\Psi : (X, \delta) \to (\mathbb{R}^N, \|\cdot\|^p)$, *i.e.*, a mapping such that

$$C^{-1} \|\Psi(x) - \Psi(y)\|^{p} \le \delta(x, y) \le C \|\Psi(x) - \Psi(y)\|^{p}$$
(2.4)

for a positive constant C and for all $x, y \in X$. For the sake of simplicity, we can identify X with $\Psi(X)$, *i.e.*, assume that Ψ is the identity.

Let choose p > 1; we will see in the following how to choose p in terms of r. Observe that - as in all metric spaces - we can recover the metric in terms of the Lipschitz functions with Lipschitz constant 1:

 $\delta(x, y) = \sup\{u(y) - u(x) : u \text{ is Lipschitz with Lipschitz constant } 1\}.$

Moreover it follows from (2.4) that for any 1-Lipschitz function u we have:

$$|u(x) - u(y)| \le C ||x - y||^p \quad \text{for all } x, y \in X.$$

Therefore, these functions are (p-1)-flat on X and using Whitney extension theorem one can extend them to C^{p-1} functions defined over all \mathbb{R}^N , that are flat on X (see section 1.10 for a similar construction). To be more precise, if $u: X \to \mathbb{R}$ is 1-Lipschitz function (with respect to δ), then one can define a C^{p-1} function $\tilde{u}: \mathbb{R}^N \to \mathbb{R}$ such that

- i) $\tilde{u} = u$ on X;
- *ii)* $d_x \tilde{u}$ exists on X;
- *iii)* $d_x \tilde{u} = 0$ exactly on X;
- *iv*) $||d_x \tilde{u}|| \leq \tilde{C} \operatorname{dist}(x, X)^{p-1}$ for each point x in a sufficiently large ball B(0, R).

Next step consists in constructing a mechanical Lagrangian $L(x, v) = \frac{1}{2} ||v||^2 + U(x)$ on \mathbb{T}^N , such that these functions \tilde{u} are 0-critical subsolutions of the associated Hamilton-Jacobi equation. For this purpose, we just need to define a potential U that satisfies suitable growth condition in a neighborhood of X. Up to multiplying U by a constant to obtain the right constant on the left-hand side, one can get:

$$\tilde{C}^2 \operatorname{dist}(x, X)^{2(p-1)} \le U(x) \le \tilde{C}_1 \operatorname{dist}(x, X)^{2(p-1)},$$
(2.5)

where \tilde{C} is the same constant as in iv) and \tilde{C}_1 some other positive constant. The usual Whitney argument (see for instance section 2.5 for a similar method) allows one to construct such a function and to show that it is C^{2p-3} . Observe that the set on which U achieves its global minimum is exactly X and therefore the associated Aubry set $\mathcal{A}_0 = X$. It is easy, in fact, to show that for mechanical systems, the projected Aubry set coincides with the set on which the potential U achieves its global minimum (see for instance Lemma 2.3.8 for a similar proof). It remains to show that $(\mathcal{A}_0, \delta_0)$ is bi-Lipschitz equivalent to (X, δ) ; if this is the case, each static class will consist of just one point and $\mathcal{A}_0 = \bar{\mathcal{A}}_0$, and this will conclude the proof of the theorem (choosing p such that $2p - 3 \ge r$).

From property iv) and (2.5), it follows that on B(0, R):

$$\frac{1}{2} \|d_x u\|^2 \le C^2 \text{dist}(x, X)^{2p-2} \le U(x);$$

this shows that \tilde{u} can be modified to be a subsolution of the system, without changing it near the Aubry set. Therefore (see (1.21)):

 $\delta_0(x,y) \ge \sup\{\tilde{u}(y) - \tilde{u}(x): u \text{ is Lipschitz with Lipschitz constant } 1\} = \delta(x,y).$

Let us prove the other inequality. If v is a 0-critical subsolution, *i.e.*, $H(x, d_x v) \leq \alpha(0) = 0$, then from (2.5) we get: $\frac{1}{2} ||d_x u||^2 \leq \tilde{C}_1 \operatorname{dist}(x, X)^{2(p-1)}$. Let $x, y \in \mathcal{A}_0$ and $\gamma : [0, ||y - x||] \to \mathbb{R}^N$ be the unit-speed geodesic joining x to y; integrating, we obtain:

$$\begin{split} v(y) - v(x) &= \int_{0}^{\|y-x\|} d_{\gamma(s)} v(\dot{\gamma}(s)) \, ds \leq \int_{0}^{\|y-x\|} \|d_{\gamma(s)} v\| \, ds \leq \\ &\leq \int_{0}^{\|y-x\|} \sqrt{2\tilde{C}_{1}} \operatorname{dist}(\gamma(s), X)^{p-1} \, ds \leq \\ &\leq \sqrt{2\tilde{C}_{1}} \int_{0}^{\|y-x\|} \operatorname{dist}(\gamma(s), x)^{p-1} \, ds \leq \\ &\leq \sqrt{2\tilde{C}_{1}} \int_{0}^{\|y-x\|} \|y-x\|^{p-1} \, ds \leq \\ &\leq \sqrt{2\tilde{C}_{1}} \|y-x\|^{p} \leq C \sqrt{2\tilde{C}_{1}} \delta(x, y); \end{split}$$

this implies $\delta_0(x,y) \leq C\sqrt{2\tilde{C}_1}\delta(x,y)$ and concludes the proof of their bi-Lipschitz equivalence.

2.3 On the total disconnectedness in high dimension

Our main goal in this section is to show that, under suitable hypotheses on L, there is a well specified cohomology class c_L , for which $(\bar{\mathcal{A}}_{c_L}, \bar{\delta}_{c_L})$ is totally disconnected, *i.e.*, every connected component consists of a single point.

As we have already recalled at the end of section 1.1, the cotangent bundle T^*M can be naturally equipped with a canonical symplectic structure $\omega = \sum_{i=1}^{d} dx_i \wedge dp_i =$ $-d\lambda$, where $\lambda = \sum_{i=1}^{d} p_i dx_i$ is the *Liouville form* (or *tautological form*). Let us denote by \mathcal{L} the Legendre transform introduced in (1.4). Consider now the section of T^*M given by

$$\Lambda_L = \mathcal{L}(M \times \{0\}) = \left\{ \left(x, \frac{\partial L}{\partial v}(x, 0) \right) : \quad x \in M \right\},\$$

corresponding to the 1-form

$$\eta_L(x) = \frac{\partial L}{\partial v}(x,0) \cdot dx = \sum_{i=1}^d \frac{\partial L}{\partial v_i}(x,0) \, dx_i \, .$$

We would like this 1-form to be closed, that is equivalent to ask Λ_L to be a Lagrangian submanifold (see proposition 3.1.2), in order to consider its cohomology class $c_L = [\eta_L] \in \mathrm{H}^1(M; \mathbb{R})$. Observe that this cohomology class can be defined in a more intrinsic way; in fact the projection

$$\pi_{|\Lambda_L} : \Lambda_L \subset \mathrm{T}^* M \longrightarrow M$$

induces an isomorphism between the cohomology groups $\mathrm{H}^1(M;\mathbb{R})$ and $\mathrm{H}^1(\Lambda_L;\mathbb{R})$. The preimage of $[\lambda_{|\Lambda_L}]$ under this isomorphism is called the *Liouville class* of Λ_L and one can easily show that it coincides with c_L . We can now define this set:

 $\mathbb{L}(M) = \{L: \mathrm{T}M \longrightarrow \mathbb{R}: L \text{ is a Tonelli Lagrangian and } \Lambda_L \text{ is Lagrangian} \}.$

This set is non-empty and consists of Lagrangians of the form

$$L(x, v) = f(x) + \langle \eta(x), v \rangle_{x} + O(||v||^{2}),$$

with $f \in C^2(M)$ and $\eta \in C^2$ closed 1-form on M.

Examples.

• Mechanical and Symmetrical Lagrangians. The set $\mathbb{L}(M)$ includes mechanical Lagrangians, *i.e.*, Lagrangians of the form

$$L(x,v) = \frac{1}{2} \|v\|_x^2 + U(x),$$

namely the sum of the kinetic energy and a potential $U : M \longrightarrow \mathbb{R}$, and more generally symmetrical (or reversible) Lagrangians, i.e., Lagrangians L : $TM \longrightarrow \mathbb{R}$ such that

$$L(x,v) = L(x,-v),$$

for every $(x, v) \in TM$. In fact in both cases, $\frac{\partial L}{\partial v}(x, 0) \equiv 0$, therefore $\Lambda_L = M \times \{0\}$ coincides with the zero section of the cotangent space, that is clearly Lagrangian with $c_L = 0$.

Mañé's Lagrangians. Let X be a C² vector field on M. We discussed in section 1.1 how to embed its flow into the Euler-Lagrange flow of a Tonelli Lagrangian, namely L_X(x, v) = ¹/₂ ||v - X(x)||²/_x.

It is easy to check that $L_X \in \mathbb{L}(M)$ if and only if X is *irrotational*, *i.e.*, the asso-

ciated 1-form $X^{\flat} := g(X, \cdot)$ is closed. In particular, $c_{L_X} = [X^{\flat}] \in \mathrm{H}^1(M; \mathbb{R})$. As a special case, one can consider the Mañé's Lagrangians associated to gradient fields $X = \nabla f$, where $f \in C^3(M; \mathbb{R})$; observe that in this case $c_{L_{\nabla f}} = 0$.

We can now state our main result:²

Theorem 2.3.1 ([75]). Let M be a compact connected manifold of dimension $d \ge 1$ and let $L \in \mathbb{L}(M)$ be a Lagrangian such that $L(x,0) \in C^r(M)$, with $r \ge 2d - 2$ and $\frac{\partial L}{\partial v}(x,0) \in C^2(M)$. If c_L denotes the Liouville class of Λ_L , then the quotient Aubry set $(\bar{\mathcal{A}}_{c_L}, \bar{\delta}_{c_L})$ is totally disconnected, i.e., every connected component consists of a single point.

Remark 2.3.2. Using the same ideas as in [10], the proof of this result can be extended to the case $L(x, 0) \in C^{2d-3,1}(M)$.

Remark 2.3.3. Observe that theorem 2.3.1 can be also stated in this form:

Let M be a compact connected manifold of dimension $d \ge 1$ and let L a Tonelli Lagrangian such that $L(x,0) \in C^r(M)$, with $r \ge 2d - 2$ (or $L(x,0) \in C^{2d-3,1}$) and $\min_{v \in T_{xM}} L(x,v) = L(x,0)$ for all $x \in M$. Then, the quotient Aubry set $(\bar{\mathcal{A}}_0, \bar{\delta}_0)$ is totally disconnected.

Lagrangians satisfying this condition are called *unimodal*.

Theorem 2.3.1 easily implies:

Corollary 2.3.4 (Symmetrical Lagrangians). Let M be a compact connected manifold of dimension $d \ge 1$ and let L(x, v) be a symmetrical Tonelli Lagrangian on TM, such that $L(x, 0) \in C^r(M)$, with $r \ge 2d - 2$. Then, the quotient Aubry set $(\bar{\mathcal{A}}_0, \bar{\delta}_0)$ is totally disconnected.

More specifically,

 $^{^{2}}$ It is worthy mentioning that a similar result has been also proven independently by Albert Fathi, Alessio Figalli and Ludovic Rifford ([33], in preparation).

Corollary 2.3.5 (Mechanical Lagrangians). Let M be a compact connected manifold of dimension $d \ge 1$ and let $L(x, v) = \frac{1}{2} ||v||_x^2 + U(x)$ be a mechanical Lagrangian on TM, such that the potential $U \in C^r(M)$, with $r \ge 2d - 2$. Then, the quotient Aubry set $(\bar{\mathcal{A}}_0, \bar{\delta}_0)$ is totally disconnected.

Remark 2.3.6. This result is optimal in terms of the regularity asked to the potential U. In fact, as we have recalled in section 2.2, Mather [64] constructed examples of quotient Aubry sets isometric to the unit interval, corresponding to mechanical Lagrangians $L \in C^{2d-3,1-\varepsilon}(\mathbb{TT}^d)$, for any $0 < \varepsilon < 1$.

Moreover, theorem 2.3.1 implies the following result for Mañé's Lagrangians associated to *irrotational* vector fields.

Corollary 2.3.7 (Mañé's Lagrangians). Let M be a compact connected manifold of dimension $d \ge 1$, equipped with a C^{∞} Riemannian metric g. Let X be a C^r vector field on M and consider the associated Mañé's Lagragian $L_X(x,v) = \frac{1}{2} ||v - X(x)||_x^2$. If $r \ge 2d - 2$ then the quotient Aubry set $(\bar{\mathcal{A}}_{c_{L_X}}, \bar{\delta}_{c_{L_X}})$ is totally disconnected, where $X^{\flat} = g(X, \cdot)$ is the associated 1-form and $c_{L_X} = [X^{\flat}] \in \mathrm{H}^1(M; \mathbb{R})$.

Before proving the theorem 2.3.1, it will be useful to show some useful properties. In particular, we will show how the condition $L \in \mathbb{L}(M)$ implies many features of the Aubry set that do not appear in the general case. They will play a key role in the proof of our result.

Lemma 2.3.8. Let us consider $L \in \mathbb{L}(M)$, such that $\frac{\partial L}{\partial v}(x,0) \in C^2(M)$, and let H be the associated Hamiltonian.

1. Every constant function $u \equiv \text{const}$ is a η_L -critical subsolution. In particular, all η_L -critical subsolutions are such that $d_x u \equiv 0$ on \mathcal{A}_{c_L} . 2. For every $x \in M$,

$$\frac{\partial H_{\eta_L}}{\partial p}(x,0) = \frac{\partial H}{\partial p}(x,\eta_L(x)) = 0.$$

Proof.

1. The second part follows immediately from the fact that, if $u, v \in S_{\eta_L}$, then they are differentiable on \mathcal{A}_{c_L} and $d_x u = d_x v$ (see proposition 1.4.21 and [32]). Let us show that $u \equiv const$ is a η_L -critical subsolution; namely, that

$$H_{\eta_L}(x,0) \le \alpha(c_L)$$

for every $x \in M$. It is sufficient to observe:

• $H_{\eta_L}(x,0) = -L(x,0)$; in fact:

$$H_{\eta_L}(x,0) = H(x,\eta_L(x)) = H\left(x,\frac{\partial L}{\partial v}(x,0)\right) = \\ = \left\langle \frac{\partial L}{\partial v}(x,0), 0 \right\rangle_x - L(x,0) = \\ = -L(x,0).$$

• Let v be *dominated* by $L_{\eta_L} + \alpha(c_L)$ (see definition 1.4.2), *i.e.*, for each continuous piecewise C^1 curve $\gamma : [a, b] \longrightarrow M$ we have

$$v(\gamma(b)) - v(\gamma(a)) \le \int_a^b L_{\eta_L}(\gamma(t), \dot{\gamma}(t)) dt + \alpha(c_L)(b-a).$$

Then, considering the constant path $\gamma(t) \equiv x$, one can easily deduce that

$$\alpha(c_L) \ge \sup_{x \in M} (-L_{\eta_L}(x, 0)) = -\inf_{x \in M} L_{\eta_L}(x, 0);$$

therefore, for every $x \in M$:

$$\alpha(c_L) \ge -L_{\eta_L}(x,0) = -L(x,0) = H_{\eta_L}(x,0).$$

2. The inverse of the Legendre transform can be written in coordinates

$$\mathcal{L}^{-1}: \ \mathbf{T}^*M \longrightarrow \mathbf{T}M$$

$$(x,p) \longmapsto \left(x, \frac{\partial H}{\partial p}(x,p)\right).$$

Therefore,

$$(x,0) = \mathcal{L}^{-1}(\mathcal{L}(x,0)) = \mathcal{L}^{-1}\left(x,\frac{\partial L}{\partial v}(x,0)\right) =$$
$$= \mathcal{L}^{-1}((x,\eta_L(x))) = \left(x,\frac{\partial H}{\partial p}(x,\eta_L(x))\right).$$

In particular, using that for any η_L -critical subsolution u we have that $H_{\eta_L}(x, d_x u) = \alpha(c_L)$ on \mathcal{A}_{c_L} , we can easily deduce that:

$$\mathcal{A}_{c_L} \subseteq \{L(x,0) = -\alpha(c_L)\} = \{H(x,\eta_L(x)) = \alpha(c_L)\}$$

and

$$\alpha(c_L) = \sup_{x \in M} (-L(x,0)) = -\inf_{x \in M} L(x,0) =: e_0;$$

this quantity e_0 has already been introduced in [48, 22], where it is referred to as strict critical value. Observe that in general it satisfies

$$e_0 \leq \min_{c \in \mathrm{H}^1(M;\mathbb{R})} \alpha(c) = -\beta(0)$$

where β : $H_1(M; \mathbb{R}) \longrightarrow \mathbb{R}$ is Mather's β -function (see (1.5)); therefore, we are considering an extremal case in which $e_0 = \alpha(c_L) = \min \alpha(c)$; it follows also quite easily that $c_L \in \partial \beta(0)$, namely, it is a subgradient of β at 0.

A crucial step in the proof of our result will be the following lemma, that can be read as a sort of relaxed version of Sard's Lemma.

Lemma 2.3.9. Let $U \in C^r(M)$, with $r \ge 2d - 2$, be a non-negative function, vanishing somewhere and denote $\mathcal{A} = \{U(x) = 0\}$. If $u : M \longrightarrow \mathbb{R}$ is C^1 and satisfies $\|d_x u\|_x^2 \le U(x)$ in an open neighborhood of \mathcal{A} , then $|u(\mathcal{A})| = 0$ (where $|\cdot|$ denotes the Lebesgue measure on \mathbb{R}).

See section 2.4 for its proof. In particular, when we apply this lemma to our specific case, it implies this essential property, which can be also read as a Sard's lemma for critical subsolutions of Hamilton-Jacobi equation.

Proposition 2.3.10 (Sard's lemma for critical subsolutions). Under the hypotheses of theorem 2.3.1, if $u \in S_{\eta_L}$, then $|u(\mathcal{A}_{c_L})| = 0$. In particular, for any $w \in \mathcal{D}_{c_L}$ we have that $|w(\mathcal{A}_{c_L})| = 0$.

Proof. First of all, we can assume that $u \in S^1_{\eta_L}$, because of Fathi and Siconolfi's theorem 1.4.24. By Taylor's formula, it follows that there exists an open neighborhood W of \mathcal{A}_{c_L} , such that for all $x \in W$:

$$\alpha(c_L) \geq H_{\eta_L}(x, d_x u) = H_{\eta_L}(x, 0) + \frac{\partial H_{\eta_L}}{\partial p}(x, 0) \cdot d_x u + \int_0^1 (1-t) \frac{\partial^2 H_{\eta_L}}{\partial p^2}(x, t \, d_x u) (d_x u)^2 \, dt \, .$$

Let us observe the following.

• From the previous lemma, one has that

$$\frac{\partial H_{\eta_L}}{\partial p}(x,0) = 0 \quad \text{for every } x \in M.$$

• From the strict convexity hypothesis, it follows that there exists $\gamma > 0$ such that:

$$\frac{\partial^2 H}{\partial p^2}(x, t \, d_x u) (d_x u)^2 \ge \gamma \| d_x u \|_x^2$$

for all $x \in M$ and $0 \le t \le 1$.

Therefore, for $x \in W$:

$$\begin{aligned} \alpha(c_L) &\geq H_{\eta_L}(x, d_x u) \geq H_{\eta_L}(x, 0) + \frac{\gamma}{2} \| d_x u \|_x^2 = \\ &= -L(x, 0) + \frac{\gamma}{2} \| d_x u \|_x^2. \end{aligned}$$

The assertion will follow from the previous lemma, choosing

$$U(x) = \frac{2}{\gamma} (\alpha(c_L) + L(x, 0)).$$

In fact by hypothesis, $U \in C^r$ with $r \geq 2d - 2$; moreover, it satisfies all other conditions since

$$\alpha(c_L) = -\inf_{x \in M} L(x, 0)$$

and

$$\mathcal{A}_{c_L} \subseteq \{x \in W : L(x,0) = -\alpha(c_L)\} = \{x \in W : U(x) = 0\} =: \mathcal{A}$$

Therefore, the previous lemma allows us to conclude that $|u(\mathcal{A}_{c_L})| = 0$.

The same proof works for $w \in \mathcal{D}_c$, observing that $||d_x(u-v)||^2 \leq 2(||d_xu||^2 + ||d_xv||^2)$. It will be sufficient in this case to take $\widetilde{U}(x) = \frac{8}{\gamma}(\alpha(c_L) + L(x, 0))$.

Proof (Theorem 2.3.1). The proof follows immediately from proposition 2.2.6 and proposition 2.3.10.

2.4 Proof of Lemma 2.3.9 and Sard's lemma for critical subsolutions

In this section we want to give a proof of Lemma 2.3.9. As we have already remarked, this will imply a version of Sard's lemma for critical subsolutions of certain Hamilton-Jacobi equations (proposition 2.3.10). See also a related work by Ludovic Rifford [70].

Definition 2.4.1. Consider a function $f \in C^r(\mathbb{R}^d)$. We say that f is s - flat at $x_0 \in \mathbb{R}^d$ (with $s \leq r$), if all its derivatives, up to the order s, vanish at x_0 .

The proof of the Lemma 2.3.9 is based on the following modified version of *Kneser-Glaeser's Rough composition theorem* (see section 2.5).

Proposition 2.4.2. Let $V, W \subset \mathbb{R}^d$ be open sets, $A \subset V$, $A^* \subset W$ closed sets. Consider $U \in C^r(V)$, with $r \geq 2$, a non-negative function which is s-flat on $A \subset \{U(x) = 0\}$, with $s \leq r - 1$, and $g : W \longrightarrow V$ a C^{r-s} homeomorphism such that $g(A^*) \subset A$.

Then, for every open pre-compact set $W_1 \supset A^*$ properly contained in W, there exists

$$F: \mathbb{R}^d \longrightarrow \mathbb{R}$$

satisfying the following properties:

- i) $F \in C^{r-1}(\mathbb{R}^d);$
- ii) $F \ge 0;$
- iii) F(x) = U(g(x)) = 0 on A^* ;
- iv) F is s-flat on A^* ;
- v) $\{F(x) = 0\} \cap W_1 = A^*;$

vi) there exists a constant K > 0, such that $U(g(x)) \leq KF(x)$ on W_1 .

To prove Lemma 2.3.9, it will be enough to show that for every $x_0 \in M$, there exists a neighborhood Ω such that the conclusion of the lemma holds. For such a local result, we can assume that $M = \mathcal{U}$ is an open subset of \mathbb{R}^d , with $x_0 \in \mathcal{U}$. In the sequel, we will identify $T^*\mathcal{U}$ with $\mathcal{U} \times \mathbb{R}^d$ and for $x \in \mathcal{U}$ we identify $T^*_x\mathcal{U} = \{x\} \times \mathbb{R}^d$. Let us equip $\mathcal{U} \times \mathbb{R}^d$ with the natural coordinates $(x_1, \ldots, x_d, p_1, \ldots, p_d)$.

Before proceeding in the proof, let us point out that it is locally possible to replace the norm obtained by the Riemannian metric, by a constant norm on \mathbb{R}^d .

Lemma 2.4.3. For each $0 < \alpha < 1$ and $x_0 \in M$, there exists an open neighborhood Ω of x_0 , with $\overline{\Omega} \subset \mathcal{U}$ and such that

$$(1-\alpha) \|p\|_{x_0} \le \|p\|_x \le (1+\alpha) \|p\|_{x_0},$$

for every $p \in T_x^* \mathcal{U} \cong \mathbb{R}^d$ and each $x \in \overline{\Omega}$.

Proof. By continuity of the Riemannian metric, the norm $||p||_x$ tends uniformly to 1 on $\{p : ||p||_{x_0} = 1\}$, as x tends to x_0 . Therefore, for x near to x_0 and every $p \in \mathbb{R}^d \setminus \{0\}$, we have:

$$(1-\alpha) \le \left\| \frac{p}{\|p\|_{x_0}} \right\|_x \le (1+\alpha).$$

We can now prove the main result of this section.

Proof (Lemma 2.3.9). By choosing local charts and by Lemma 2.4.3, we can assume that $U \in C^r(\Omega)$, with Ω open set in \mathbb{R}^d , $\mathcal{A} = \{x \in \Omega : U(x) = 0\}$ and $u : \Omega \longrightarrow \mathbb{R}$ is such that $||d_x u||^2 \leq \beta U(x)$ in Ω , where β is a positive constant.

Define, for $1 \leq s \leq r$:

$$B_s = \{ x \in \mathcal{A} : U \text{ is } s \text{ - flat at } x \}$$

and observe that

$$\mathcal{A} = B_1 := \left\{ x \in \mathcal{A} : DU(x) = 0 \right\}.$$

We will prove the lemma by induction on the dimension d. Let us start with the following claim.

Claim 2.4.4. If $s \ge 2d - 2$, then $|u(B_s)| = 0$.

Proof. Let $C \subset \Omega$ be a closed cube with edges parallel to the coordinate axes. We will show that $|u(B_s \cap C)| = 0$. Since B_s can be covered by countably many such cubes, this will prove that $|u(B_s)| = 0$.

Let us start observing that, by Taylor's theorem, for any $x \in B_s \cap C$ and $y \in C$ we have

$$U(y) = R_s(x; y),$$

where $R_s(x; y)$ is Taylor's remainder. Therefore, for any $y \in C$

$$U(y) = o(||y - x||^{s}).$$

Let λ be the length of the edge of C. Choose an integer N > 0 and subdivide Cin N^d cubes C_i with edges $\frac{\lambda}{N}$, and order them so that one has $C_i \cap B_s \neq \emptyset$, for $1 \leq i \leq N_0 \leq N^d$. Hence,

$$B_s \cap C = \bigcup_{i=1}^{N_0} B_s \cap C_i.$$

Observe that for every $\varepsilon > 0$, there exists $\nu_0 = \nu_0(\varepsilon)$ such that, if $N \ge \nu_0$, $x \in B_s \cap C_i$ and $y \in C_i$, for some $0 \le i \le N_0$, then

$$U(y) \le \frac{\varepsilon^2}{4\beta (d\lambda^2)^d} \|y - x\|^s.$$

Fix $\varepsilon > 0$. Choose $x_i \in B_s \cap C_i$ and call $y_i = u(x_i)$. Define, for $N \ge \nu_0$, the

following intervals in \mathbb{R} :

$$E_i = \left[y_i - \frac{\varepsilon}{2N^d}, y_i + \frac{\varepsilon}{2N^d}\right].$$

Let us show that if N is sufficiently big, then $u(B_s \cap C) \subset \bigcup_{i=1}^{N_0} E_i$. In fact, if $x \in B_s \cap C$, then there exists $1 \leq i \leq N_0$, such that $x \in B_s \cap C_i$. Therefore,

$$\begin{aligned} |u(x) - y_i| &= |u(x) - u(x_i)| = \\ &= ||d_x u(\tilde{x})|| \cdot ||x - x_i|| \le \\ &\le \sqrt{\beta U(\tilde{x})} ||x - x_i|| \le \\ &\le \sqrt{\beta \frac{\varepsilon^2}{4\beta (d\lambda^2)^d}} ||\tilde{x} - x_i||^{\frac{s}{2}} ||x - x_i|| \le \\ &\le \frac{\varepsilon}{2(d\lambda^2)^{\frac{d}{2}}} ||x - x_i||^{\frac{s+2}{2}} \le \\ &\le \frac{\varepsilon}{2(d\lambda^2)^{\frac{d}{2}}} \left(\sqrt{d} \frac{\lambda}{N}\right)^{\frac{s+2}{2}}, \end{aligned}$$

where \tilde{x} is a point in the segment joining x and x_i . Since by hypothesis $s \ge 2d - 2$, then $\frac{s+2}{2} \ge d$. Hence, assuming that $N > \max\{\lambda \sqrt{d}, \nu_0\}$, one gets

$$|u(x) - y_i| \le \frac{\varepsilon}{2N^d}$$

and can deduce the above inclusion.

To prove the claim, it is now enough to observe:

$$|u(B_s \cap C)| \leq \left| \bigcup_{i=1}^{N_0} E_i \right| \leq \sum_{i=1}^{N_0} |E_i| \leq \varepsilon N_0 \frac{1}{N^d} \leq \varepsilon N^d \frac{1}{N^d} = \varepsilon .$$

From the arbitrariness of ε , the assertion follows easily.

This claim immediately implies that $u(B_{2d-2})$ has measure zero. In particular, this proves the case d = 1 (since in this case 2d - 2 = 0) and allows us to start the induction.

Let us suppose to have proven the result for d-1 and try to show it for d. Since

$$\mathcal{A} = (B_1 \setminus B_2) \cup (B_2 \setminus B_3) \cup \ldots \cup (B_{2d-3} \setminus B_{2d-2}) \cup B_{2d-2},$$

it remains to show that $|u(B_s \setminus B_{s+1})| = 0$ for $1 \le s \le 2d - 3 \le r - 1$.

Claim 2.4.5. Every $\tilde{x} \in B_s \setminus B_{s+1}$ has a neighborhood \tilde{V} , such that

$$|u((B_s \setminus B_{s+1}) \cap \tilde{V})| = 0.$$

Since $B_s \setminus B_{s+1}$ can be covered by countably many such neighborhoods, this implies that $u(B_s \setminus B_{s+1})$ has measure zero.

Proof. Choose $\tilde{x} \in B_s \setminus B_{s+1}$. By definition of these sets, all partial derivatives of order s of U vanish at this point, but there is one of order s+1 that does not. Assume (without any loss of generality) that there exists a function

$$w(x) = \partial_{i_1} \partial_{i_2} \dots \partial_{i_s} U(x)$$

such that

$$w(\tilde{x}) = 0$$
 but $\partial_1 w(\tilde{x}) \neq 0$.

Define

$$\begin{array}{rcl} h:\Omega & \longrightarrow & \mathbb{R}^d \\ & x & \longmapsto & (w(x), \, x_2, \, \dots, \, x_d) \,, \end{array}$$

where $x = (x_1, x_2, ..., x_d)$. Clearly, $h \in C^{r-s}(\Omega)$ and $Dh(\tilde{x})$ is non-singular; hence, there is an open neighborhood V of \tilde{x} such that

$$h:V\longrightarrow W$$

is a C^{r-s} isomorphism (with W = h(V)).

Let V_1 be an open precompact set, containing \tilde{x} and properly contained in V, and define $A = B_s \cap \overline{V_1}$, $A^* = h(A)$ and $g = h^{-1}$. If we consider W_1 , any open set containing A^* and properly contained in W, we can apply proposition 2.4.2 and deduce the existence of $F : \mathbb{R}^d \longrightarrow \mathbb{R}$ satisfying properties i)-vi).

Define

$$\hat{W} = \{(x_2, \ldots, x_d) \in \mathbb{R}^{d-1} : (0, x_2, \ldots, x_d) \in W_1\}$$

and

$$\hat{U}(x_2,\ldots,x_d) = C F(0,x_2,\ldots,x_d),$$

where C is a positive constant to be chosen sufficiently big. Observe that $\hat{U} \in C^{r-1}(\mathbb{R}^{d-1})$. Moreover, property v) of F and the fact that $A^* = h(A) \subseteq \{0\} \times \hat{W}$ imply that:

$$A^* = \{0\} \times \hat{B}_1,$$

where $\hat{B}_1 = \{(x_2, \dots, x_d) \in \hat{W} : F(0, x_2, \dots, x_d) = 0\}$. Denote

$$\hat{\mathcal{A}} := \{ (x_2, \dots, x_d) \in \hat{W} : \hat{U} = 0 \} = \hat{B}_1$$

and define the following function on \hat{W} :

$$\hat{u}(x_2,\ldots,x_d)=u(g(0,x_2,\ldots,x_d)).$$

We want to show that these functions satisfy the hypotheses for the (d-1)-dimensional

case. In fact:

- $\hat{U} \in C^{r-1}(\mathbb{R}^{d-1})$, with $r-1 \ge 2d-3 > 2(d-1)-2$;
- $\hat{u} \in C^1(\hat{W})$ (since g is in $C^{r-s}(W)$, where $1 \le s \le r-1$);
- if we denote by $\mu = \sup_{W_1} ||d_x g|| < +\infty$ (since g is C^1 on $\overline{W_1}$), then we have that for every point in \hat{W} :

$$\begin{aligned} \|d\hat{u}(x_{2},\ldots,x_{d})\|^{2} &\leq \|d_{x}u(g(0,x_{2},\ldots,x_{d}))\|^{2}\|d_{x}g(0,x_{2},\ldots,x_{d})\|^{2} \leq \\ &\leq \mu^{2}\|d_{x}u(g(0,x_{2},\ldots,x_{d}))\|^{2} \leq \\ &\leq \beta\mu^{2}U(g(0,x_{2},\ldots,x_{d})) \leq \\ &\leq \beta\mu^{2}KF(0,x_{2},\ldots,x_{d}) \leq \\ &\leq \hat{U}(x_{2},\ldots,x_{d}), \end{aligned}$$

if we choose $C > \beta \mu^2 K$, where K is the positive constant appearing in proposition 2.4.2, property vi).

Therefore, it follows from the inductive hypothesis that: $|\hat{u}(\hat{\mathcal{A}})| = 0$. Since

$$u(B_s \cap V_1) \subseteq u(A) = u(g(A^*)) = u(g(\{0\} \times \hat{B}_1)) = \\ = \hat{u}(\hat{B}_1) = \hat{u}(\hat{A}),$$

defining $\tilde{V} = V_1$, we may conclude that

$$|u(B_s \cap \tilde{V})| \le |\hat{u}(\hat{\mathcal{A}})| = 0.$$

This completes the proof of Lemma 2.3.9.

91

2.5 Proof of a modified version of Kneser-Glaeser's Rough composition theorem.

Now, let us prove proposition 2.4.2. We will prove a modified version of the result presented in [1], adapted to our different needs and setting.

Proof (Proposition 2.4.2). Let us start by defining a family of polynomials. Supposing that g is C^r and using the *s*-flatness hypothesis, we have, for $x \in A^*$ and k = 0, 1, ..., r:

$$f_k(x) = D^k(U \circ g)(x) = \sum_{s < q \le k} \sum \sigma_k D^q U(g(x)) D^{i_1}g(x) \dots D^{i_q}g(x), \qquad (2.6)$$

where the second sum is over all the q-tuples of integers $i_1, \ldots, i_q \ge 1$ such that $i_1 + \ldots + i_q = k$, and $\sigma_k = \sigma_k(i_1, \ldots, i_q)$.

The crucial observation is that (2.6) makes sense on A^* , even when g is C^{r-s} smooth (in fact $i_j \leq k - q + 1 \leq r - s$).

We would like to proceed in the fashion of *Whitney's extension theorem*, in order to find a smooth function F such that $D^k F = f_k$ on A^* and satisfying the stated conditions.

Remark 2.5.1. Note that, without any loss of generality, we can assume that W is contained in an open ball of diameter 1. The general case will then follow from this special one, by a straightforward partition of unity argument.

Let us start with some technical lemmata.

Lemma 2.5.2. For $x, x', x_0 \in A^*$ and k = 0, ..., r, we have:

$$f_k(x') = \sum_{i \le r-k} \frac{f_{k+i}(x)}{i!} (x'-x)^i + R_k(x,x') \,,$$

with

$$\frac{|R_k(x,x')|}{\|x'-x\|^{r-k}} \longrightarrow 0$$

as $x, x' \longrightarrow x_0$ in A^* .

Proof. Let $y \in A^*$, $y' \in V$ and q = s + 1, ..., r. By Taylor's formula, we may write

$$D^{q}U(y') = \sum_{\alpha \le r-q} \frac{D^{q+\alpha}U(y)}{\alpha!} (y'-y)^{\alpha} + I_{q}(y,y')(y'-y)^{r-q}$$

where

$$I_q(y,y') = \int_0^1 \frac{(1-t)^{r-q}}{(r-q)!} \{ D^r U(y+t(y'-y)) - D^r U(y) \} dt \,.$$

Similarly, for $i \leq r - s$ and $x, x' \in W$,

$$D^{i}g(x') = \sum_{\beta \le r-s-i} \frac{D^{i+\beta}g(x)}{\beta!} (x'-x)^{\beta} + J_{i}(x,x')(x'-x)^{r-s-i}$$

where

$$J_i(x,x') = \int_0^1 \frac{(1-t)^{r-s}}{(r-s)!} \{ D^{r-s}g(x+t(x'-x)) - D^{r-s}g(x) \} dt.$$

It follows from the formulae for I_q and J_i , that they are locally uniformly continuous in both variables and zero on the diagonal.

Now, take $x, x', x_0 \in A^*$ and let y = g(x) and y' = g(x'). By (2.6) we have

$$f_k(x') = \sum_{s < q \le k} \sum \sigma_k D^q U(y') D^{i_1} g(x') \dots D^{i_q} g(x') .$$

Substituting the expressions above, we get:

$$f_k(x') = \sum_{j=0}^{r-k} a_j^k (x'-x)^j + R_k(x,x'), \qquad (2.7)$$

where $R_k(x, x')$ is the sum of all terms T of one of the following forms:

- a) $a(x'-x)^{j}$ where j > r-k;
- b) $I_q(y, y')(y' y)^{r-q};$
- c) $J_{i_j}(x, x')(x' x)^{r-s-i_j}$.

We will show that if T is any term in $R_k(x, x')$, then

$$\frac{|T|}{\|x'-x\|^{r-k}} \longrightarrow 0 \qquad \text{as } x, \, x' \to x_0 \text{ in } A^*.$$

In fact, if T is of form a), this is immediate. If T has form b), then it follows easily from the fact that $q \leq k$, g is Lipschitz and I_q locally uniformly continuous and zero on the diagonal. For any term of the form c), $i_j \leq k - q + 1$ where $s + 1 \leq q$. Hence, $s + i_j \leq k$ and $r - s - i_j \geq r - k$. Thus, it can be deduced by the same reasoning as case b).

To conclude the proof of this lemma, we need to show that for each k = 0, 1, ..., rand j = 0, ..., r - k we have:

$$j!a_j^k = f_{k+j}(x). (2.8)$$

To see this, suppose for the moment that g is class C^r . Then so is $U \circ g$ and (2.7) is Taylor's formula for $f_k = D^k(U \circ g)$. Then, by the uniqueness of Taylor's formula, we have

$$j!a_j^k = D^{k+j}(U \circ g)(x) = f_{k+j}(x).$$

Since (2.8) is an identity in the derivatives of U and g, and points x and x', and since one can always find a C^r function with prescribed derivatives at any finite set of points, it follows that (2.8) holds in general.

Define, for $x \in A^*$ and $y \in \mathbb{R}^d$

$$P(x,y) = \sum_{i=s+1}^{r} \frac{f_i(x)}{i!} (y-x)^i$$

and its k-th derivative

$$P_k(x,y) = \sum_{i \le r-k} \frac{f_{i+k}(x)}{i!} (y-x)^i.$$

Lemma 2.5.3. For $x \in A^*$ and $y \in W_1$,

$$U(g(y)) = P(x, y) + R(x, y),$$

where $|R(x, y)| \le C ||y - x||^{r}$.

Proof. The proof follows the same idea of Lemma 2.5.2. By Taylor's formula for U,

$$U(g(y)) = \sum_{q=s+1}^{r} \frac{D^q U(g(x))}{q!} (g(y) - g(x))^q + I(g(x), g(y))(g(x) - g(y))^r.$$

Obviously,

$$|I(g(x), g(y))(g(x) - g(y))^r| \le C_1 ||y - x||^r,$$

therefore it is sufficient to estimate the first term.

Observe that:

$$g(y) = g(x) + \sum_{i=1}^{r-s} D^{i}g(x)(y-x)^{i} + J(x,y)(y-x)^{r-s}.$$

Hence, the first term in the above sum becomes:

$$\sum_{q=s+1}^{r} \frac{D^{q}U(g(x))}{q!} \left[\sum_{i=1}^{r-s} D^{i}g(x)(y-x)^{i} + J(x,y)(y-x)^{r-s} \right]^{q} = \\ = \sum_{k=s+1}^{r} a_{k}(y-x)^{k} + \hat{R}(x,y) = \\ = P(x,y) + \hat{R}(x,y) ,$$

since

$$a_k = \sum_{s+1 \le q \le k} \sum D^q U(g(x)) D^{i_1} g(x) \dots D^{i_q} g(x) = \frac{f_k(x)}{k!} \,.$$

The remainder terms consist of:

- terms containing $(y x)^k$, with k > r;
- terms of the binomial product, containing $J(x, y)(y x)^{r-s}$. They are of the form:

$$\dots (y-x)^{(r-s)j+\sum_{i=1}^{r-s}i\alpha_i}$$

where $\alpha_i \ge 0$ and $\sum \alpha_i = q - j$. Since $q \ge s + 1$ and $s \le r - 1$, then:

$$(r-s)j + \sum_{i=1}^{r-s} i\alpha_i \geq (r-s)j + \sum_{i=1}^{r-s} \alpha_i = = (r-s)j + q - j = = rj - sj + q - j \ge \geq rj - (s+1)j + s + 1 = = r + r(j-1) - (s+1)(j-1) = = r + (r-s-1)(j-1) \ge r.$$

Therefore, for $x \in A^*$ and $y \in W_1$

$$\left|\hat{R}(x,y)\right| \le C_2 \|y-x\|^r$$

and the lemma follows taking $C = C_1 + C_2$.

Next step will consist in creating a *Whitney's partition*. We will start by covering $W_1 \setminus A^*$ with an infinite collection of cubes K_j , such that the size of each K_j is roughly proportional to its distance from A^* .

First, let us fix some notation. We will write $a \prec b$ instead of "there exists a positive real constant M, such that $a \leq Mb$ " and $a \approx b$ as short for $a \prec b$ and $b \prec a$. Let $\lambda = \frac{1}{4\sqrt{d}}$; this choice will come in handy later. For any closed cube K (with edges parallel to the coordinate axes), K^{λ} will denote the $(1 + \lambda)$ - dilation of K about its center.

Let $\|\cdot\|$ be the euclidean metric on \mathbb{R}^d and

$$d(y) = d(y, A^*) = \inf\{\|y - x\| : x \in A^*\}.$$

If $\{K_j\}_j$ is the sequence of closed cubes defined below, with edges of length e_j , let d_j be its distance from A^* , *i.e.*,

$$d_j = d(A^*, K_j) = \inf\{\|y - x\| : x \in A^*, y \in K_j\}$$

One can show the following classical lemma.

Lemma 2.5.4. There exists a sequence of closed cubes $\{K_j\}_j$ with edges parallel to the coordinate axes, that satisfies the following properties:

- i) the interiors of the K_j 's are disjoint;
- ii) $W_1 \setminus A^* \subset \bigcup_j K_j;$

- iii) $e_j \approx d_j;$
- iv) $e_j \approx d(y)$ for all $y \in K_j^{\lambda}$;
- v) $e_j \approx d(z)$ for all $z \in W_1 \setminus A^*$, such that the ball with center z and radius $\frac{1}{8}d(z)$ intersects K_j^{λ} ;
- vi) each point of $W_1 \setminus A^*$ has a neighborhood intersecting at most N of the K_j^{λ} , where N is an integer depending only on d.

Proof. Divide \mathbb{R}^d into cubes of unit edge. Accept all cubes which intersect $W_1 \setminus A^*$ and are at distance $\geq \frac{1}{2}$ from A^* . Bisect each of the rejected cubes into 2^d parts and accept all those cubes which intersect $W_1 \setminus A^*$ and are at distance $\geq \frac{1}{4}$ from A^* . Repeat indefinitely to get the sequence $\{K_j\}_j$.

Properties i) and ii) are immediate. To prove iii) we show that:

$$\frac{e_j}{2} \le d_j \le (1 + 2\sqrt{d})e_j \quad \text{for all } j \,.$$

That $\frac{e_j}{2} \leq d_j$ follows easily from the definition. Let $e_j = 2^{-k}$. For k = 0, we have $d_j \leq 1$, since the diameter of W is less than 1 (see remark above). For k > 0, K_j is one of the 2^d dyadic cubes of a cube K^* and K^* was rejected. Then $e^* = 2e_j = 2^{-k+1}$ and

$$d^* = d(A, K^*) < 2^{-k} = e_j.$$

From the triangle inequality and the fact that the diameter of a cube is \sqrt{d} times the length of its edge, it follows that:

$$d_j \le (1 + 2\sqrt{d})e_j.$$

This completes the proof of iii).

To prove iv), let us show that

$$\frac{e_j}{4} \le d(y) \le (4+\lambda)\sqrt{d}e_j,$$

whenever $y \in K_j^{\lambda}$.

Choose $y \in K_j^{\lambda}$. Let $p_j \in K_j$ satisfy $d(y, K_j) = ||y - p_j||$. Then

$$\|y - p_j\| \le \lambda \sqrt{d} e_j = \frac{1}{4} e_j.$$

By iii),

$$\frac{e_j}{2} \le d_j \le d(p_j) \le d(y) + ||y - p_j||.$$

Hence, $\frac{1}{4}e_j \leq d(y)$. Now, let $q_j \in K_j$ satisfy $d(q_j) = d_j$. Then, by iii) and geometric considerations,

$$d(y) \le ||y - q_j|| + d(q_j) \le \sqrt{d}(1 + \lambda)e_j + (1 + 2\sqrt{d})e_j.$$

Hence, $d(y) \leq (4 + \lambda)\sqrt{d}e_j$. This completes the proof of iv). Now for v), let $a = \frac{1}{4}$ and $b = (4 + \lambda)\sqrt{d}$ and y in the ball of radius $\frac{1}{2}ad(z) = \frac{1}{8}d(z)$

and center z, and $y \in K_j^{\lambda}$. By iv), $ae_j \leq d(y) \leq be_j$. Since

$$d(y) \le ||y - z|| + d(z)$$
 and $d(z) \le ||y - z|| + d(y)$

we have

$$\frac{a}{2}e_j \le d(z) \le \left(b + \frac{a}{2}\right)e_j.$$

This proves v).

Now, let us prove vi). From v), it follows that there are constants p and q, depending only on d, such that for every $z \in W_1 \setminus A^*$, the K_j^{λ} 's intersecting the ball about zwith radius $\frac{1}{8}d(z)$ are contained in a ball of radius pd(z) about z and have edges at most qd(z). Let N be the maximum number of cubes with edges $\geq \frac{q}{p}$ that fit in the unit ball; it depends only on d and at most N of the K_j^{λ} 's intersect the ball of radius $\frac{1}{8}d(z)$ and center z. This completes the proof of vi).

Now, let us construct a partition of unity on $W_1 \setminus A^*$. Let Q be the unit cube centered at the origin. Let η be a C^{∞} bump function defined on \mathbb{R}^d such that

$$\eta(y) = \begin{cases} 1 & \text{for } y \in Q \\ 0 & \text{for } y \notin Q^{\lambda} \end{cases}$$

and $0 \le \eta \le 1$. Define

$$\eta_j(y) = \eta\left(\frac{y-c_j}{e_j}\right) \,,$$

where c_j is the center of K_j and e_j is the length of its edge, and consider

$$\sigma(y) = \sum_{j} \eta_j(y) \,.$$

Then, $1 \leq \sigma(y) \leq N$ for all $y \in W_1 \setminus A^*$. Clearly, for each $k = 0, 1, 2, \ldots$ we have that $D^k \eta_j(y) \prec e_j^{-k}$, for all $y \in W_1 \setminus A^*$. Hence, by properties iv) and vi) of Lemma 2.5.4, we have that for each $k = 0, 1, \ldots, r$:

$$D^k \eta_j(y) \prec d(y)^{-k}$$
 for all $y \in W_1 \setminus A^*$

and

$$D^k \sigma(y) \prec d(y)^{-k}$$
 for all $y \in W_1 \setminus A^*$.

Define

$$\varphi_j(y) = \frac{\eta_j(y)}{\sigma(y)}$$

These functions satisfy the following properties:

i) each φ_j is C^{∞} and supported on K_j^{λ} ;

- ii) $0 \le \varphi_j(y) \le 1$ and $\sum_j \varphi_j(y) = 1$, for all $y \in W_1 \setminus A^*$;
- iii) every point of $W_1 \setminus A^*$ has a neighborhood on which all but at most N of the φ_j 's vanish identically;
- iv) for each $k = 0, 1, ..., r, D^k \varphi_j(y) \prec d(y)^{-k}$ for all $y \in W_1 \setminus A^*$; namely, there are constants M_k such that $D^k \varphi_j(y) \leq M_k d(y)^{-k}$;
- v) there is a constant α and points $x_j \in A^*$, such that:

$$||x_j - y|| \le \alpha d(y),$$
 whenever $\varphi_j(y) \ne 0.$

This follows from properties iii) and iv) of Lemma 2.5.4.

We can now construct our function F. Observe that, from lemma 2.5.3:

$$0 \le U(g(y)) = P(x_j, y) + R(x_j, y) \le P(x_j, y) + C ||y - x_j||^r;$$

therefore $P(x_j, y) \ge -C ||y - x_j||^r$.

First, define

$$\hat{P}_j(y) = P(x_j, y) + 2C ||y - x_j||^r$$

where C is the same constant as in Lemma 2.5.3; for what said above,

$$\hat{P}_{j}(y) \ge C \|y - x_{j}\|^{r} > 0 \quad \text{in } W_{1} \setminus \{x_{j}\}.$$
(2.9)

•

Hence, construct F in the following way:

$$F(y) = \begin{cases} 0 & y \in A^* \\ \sum_j \varphi_j(y) \hat{P}_j(y) & y \in \mathbb{R}^d \setminus A^* \end{cases}$$

We claim that this satisfies all the stated properties i)-vi). In particular, properties

ii), iii) and v) follow immediately from the definition of F and (2.9). Moreover, $F \in C^{\infty}(\mathbb{R}^d \setminus A^*)$. We need to show that $D^k F = f_k$ (for k = 0, 1, ..., r - 1) on ∂A^* (namely, the boundary of A^*) and that $D^{r-1}F$ is continuous on it. The main difficult in the proof, is that $D^k F$ is expressed as a sum containing terms

$$D^{k-m}\varphi_i(y)P_m(x_i,y),$$

where $\varphi_j(y) \neq 0$. Even if y is close to some $x_0 \in A^*$, it could be closer to A^* and hence the bound given by property iv) of φ_j might become large. One can overcome this problem by choosing a point $x^* \in A^*$, so that $||x^* - y||$ is roughly the same as d(y) and hence, x_j is close to x^* .

Lemma 2.5.5. For every $\eta > 0$, there exists $\delta > 0$ such that for all $y \in W_1 \setminus A^*$, $x, x^* \in A^*$ and $x_0 \in \partial A^*$, we have

$$||P_k(x,y) - P_k(x^*,y)|| \le \eta \, d(y)^{r-k} \le \eta ||y - x_0||^{r-k},$$

whenever $k \leq r$ and

$$\begin{cases} \|y - x\| < \alpha d(y) \\\\ \|y - x^*\| < \alpha d(y) \\\\ \|y - x_0\| < \delta \,, \end{cases}$$

where α is the same constant as in v) above.

Proof. Let $\varepsilon > 0$ to be defined below. From Lemma 2.5.2, it follows that there exists $\delta > 0$ such that

$$||R_{k+q}(x^*, x)|| \le \varepsilon ||x^* - x||^{r-k-q} \le \varepsilon (2\alpha d(y))^{r-k-q}.$$

In fact $d(y) \leq ||y - x_0||$, hence by making y close to x_0 (*i.e.*, δ small), one can make x and x^* close to x_0 .

Observe that, for $x, x^* \in A^*$ and $y \in \mathbb{R}^d$ we have

$$P_k(x,y) = P_k(x^*,y) + \sum_{q \le r-k} \frac{R_{k+q}(x^*,x)}{q!} (y-x)^q,$$

as it can be easily shown using the definition of P_k , the fact that the $f_k(x^*)$ are multilinear maps and the analogous of the binomial theorem.

Therefore,

$$\begin{aligned} \|P_k(x,y) - P_k(x^*,y)\| &\leq \sum_{q \leq r-k} \|R_{k+q}(x^*,x)(y-x)^q\| \leq \\ &\leq \sum_{q \leq r-k} \varepsilon (2\alpha d(y))^{r-k} = \\ &= \varepsilon (r-k) (2\alpha d(y))^{r-k} \,. \end{aligned}$$

Hence, take $\varepsilon = \frac{\eta}{(r-k)(2\alpha)^{r-k}}$ when $r \neq k$, and $\varepsilon = \eta$ for r = k. This concludes the proof of the lemma.

Lemma 2.5.6. For every $\eta > 0$, there exist $0 < \delta < 1$ and a constant E, such that for all $y \in W_1 \setminus A^*$, $x^* \in A^*$ and $x_0 \in \partial A^*$, we have

$$||D^k F(y) - P_k(x^*, y)|| \le E \, d(y)^{r-k} \le \eta \, d(y)^{r-k-1},$$

whenever $k \leq r-1$ and

$$\begin{cases} \|y - x^*\| < \alpha d(y) \\ \|y - x_0\| < \delta \,. \end{cases}$$

Proof. Let $S_{j,k}(x^*, y) = \partial_k \hat{P}_j(y) - P_k(x^*, y)$. From Lemma 2.5.5 (with $\eta = \varepsilon$, to be

defined later) and the definition of $\hat{P}_j,$ we get:

$$||S_{j,k}(x^*, y)|| \leq ||\partial_k \hat{P}_j(y) - P_k(x_j, y)|| + ||P_k(x_j, y) - P_k(x^*, y)|| \leq C_k d(y)^{r-k} + \varepsilon d(y)^{r-k} = (C_k + \varepsilon) d(y)^{r-k}.$$

Then,

$$F(y) - P(x^*, y) = \sum_{j} \varphi_j(y) S_{j,0}(x^*, y)$$

and hence

$$D^{k}F(y) - P_{k}(x^{*}, y) = \sum_{j} \sum_{i \leq k} \binom{k}{i} D^{k-i}\varphi_{j}(y)S_{j,i}(x^{*}, y).$$

Therefore, choosing ε sufficiently small:

$$\begin{aligned} \|D^k F(y) - P_k(x^*, y)\| &\leq \sum_j \sum_{i \leq k} \binom{k}{i} \|D^{k-i} \varphi_j(y)\| \cdot \|S_{j,i}(x^*, y)\| \leq \\ &\leq \sum_j \sum_{i \leq k} \binom{k}{i} M_{k-i} d(y)^{-k+i} (C_k + \varepsilon) d(y)^{r-i} \leq \\ &\leq E d(y)^{r-k} \leq \eta d(y)^{r-k-1}. \end{aligned}$$

Lemma 2.5.7. For every $\eta > 0$, there exists $0 < \delta < 1$ such that, for all $y \in W_1 \setminus A^*$, $x^* \in A^*$ and $x_0 \in \partial A^*$, we have

$$||P_k(x^*, y) - P_k(x_0, y)|| \le \eta ||y - x_0||^{r-k},$$

whenever $k \leq r$ and

$$\begin{cases} ||y - x^*|| < \alpha d(y) \\ ||y - x_0|| < \delta. \end{cases}$$

Proof. The proof goes as the one of Lemma 2.5.5, observing that $||x^* - x_0|| \le (1 + \alpha)||y - x_0||$ and

$$P_k(x_0, y) - P_k(x^*, y) = \sum_{q \le r-k} \frac{R_{k+q}(x^*, x_0)}{q!} (y - x)^q.$$

Claim 2.5.8. For every $x_0 \in \partial A^*$ and k = 0, 1, ..., r - 1:

$$D^k F(x_0) = f_k(x_0) \,.$$

Moreover, $D^{r-1}F$ is continuous at $x_0 \in \partial A^*$.

Proof. This claim follows easily from the lemmata above. We proceed by induction on k. For k = 0, it follows immediately from the definition of F (defining $f_0 = U \circ g$). Assume k < r - 1 and $D^k F(x_0) = f_k(x_0)$. We will show that for every $\varepsilon > 0$, there exists $\delta > 0$ such that

$$||D^{k}F(y) - f_{k}(x_{0}) - f_{k+1}(x_{0})(y - x_{0})|| \le \varepsilon ||y - x_{0}||, \qquad (2.10)$$

whenever $||y - x_0|| < \delta$.

Choose $\varepsilon > 0$. Take $\eta = \frac{\varepsilon}{3}$ in Lemma 2.5.6 and 2.5.7 and the corresponding $\delta_1, \delta_2 > 0$. Since

$$||P_k(x_0, y) - f_k(x_0) - f_{k+1}(x_0)(y - x_0)|| \le \sum_{2 \le i \le r-k} \frac{||f_{k+i}(x_0)||}{i!} ||y - x_0||^i,$$

we may choose $\delta_3 > 0$ so small that

$$||P_k(x_0, y) - f_k(x_0) - f_{k+1}(x_0)(y - x_0)|| \le \eta ||y - x_0||,$$

whenever $||y - x_0|| < \delta_3$.

By Lemma 2.5.2, we know that we can choose δ_4 so small that (2.10) holds when $||y - x_0|| < \delta_4$ and $y \in A^*$. Now take $\delta = \min\{\delta_1, \delta_2, \delta_3, \delta_4, 1\}$. Choose y such that $||y - x_0|| < \delta$. If $y \in A^*$, we have completed the proof. If $y \in W_1 \setminus A^*$, choose $x^* \in A^*$ with $||y - x^*|| < \alpha d(y)$. Then, by Lemma 2.5.6 and 2.5.7 and the above estimate, we have:

$$\begin{split} \|D^{k}F(y) - f_{k}(x_{0}) - f_{k+1}(x_{0})(y - x_{0})\| &\leq \\ &\leq \|D^{k}F(y) - P_{k}(x^{*}, y)\| + \|P_{k}(x^{*}, y) - P_{k}(x_{0}, y)\| + \\ &+ \|P_{k}(x_{0}, y) - f_{k}(x_{0}) - f_{k+1}(x_{0})(y - x_{0})\| \leq \\ &\leq \eta \|y - x_{0}\|^{r-k-1} + \eta \|y - x_{0}\|^{r-k} + \eta \|y - x_{0}\|^{r-k} \leq \\ &\leq 3\eta \|y - x_{0}\| = \\ &= \varepsilon \|y - x_{0}\|. \end{split}$$

This proves iv). Moreover,

Claim 2.5.9. $D^{r-1}F$ is continuous at $x_0 \in \partial A^*$.

Proof. We must show that for every $\varepsilon > 0$, there exists $\delta > 0$ such that

$$\|D^{r-1}F(y) - f_{r-1}(x_0)\| \le \varepsilon,$$

whenever $||y - x_0|| < \delta$.

In fact, using Lemma 2.5.6 and 2.5.7 and the same argument as in the previous claim

(with $\eta = \frac{\varepsilon}{2}$), one can get:

$$\begin{split} \|D^{r-1}F(y) - f_{r-1}(x_0)\| &\leq \\ &\leq \|D^{r-1}F(y) - P_{r-1}(x^*, y)\| + \|P_{r-1}(x^*, y) - f_{r-1}(x_0, y)\| = \\ &\leq \|D^{r-1}F(y) - P_{r-1}(x^*, y)\| + \|P_{r-1}(x^*, y) - P_{r-1}(x_0, y)\| \leq \\ &\leq \eta + \eta \|y - x_0\| \leq \\ &\leq \eta (1+\delta) \leq 2\eta = \varepsilon \,. \end{split}$$

This proves that $F \in C^{r-1}(\mathbb{R}^d)$ and completes the proof of i) and iv).

It remains to show that property vi) holds, namely that there exists a constant K > 0, such that $U(g(x)) \leq KF(x)$ on W_1 . Obviously, this holds at every point in A^* , for every choice of K (since both functions vanish there).

Claim 2.5.10. There exists a constant K > 0, such that $\frac{U \circ g}{F} \leq K$ on $W_1 \setminus A^*$.

Proof. Since F > 0 on $W_1 \setminus A^*$, it is sufficient to show that $\frac{U \circ g}{F}$ is uniformly bounded by a constant, as d(y) goes to zero.

Let us start observing that, for $y \in K_j^{\lambda}$,

$$\hat{P}_j(y) \ge C \|y - x_j\|^r \ge C d(y)^r;$$

therefore:

$$F(y) = \sum_{j} \varphi_{j}(y) \hat{P}_{j}(y) \ge$$
$$\ge \sum_{j} \varphi_{j}(y) C d(y)^{r} =$$
$$= C d(y)^{r}.$$

Moreover, if $x^* \in A^*$ such that $d(y) = ||y - x^*||$, lemmata 2.5.3 and 2.5.6 imply:

$$\begin{aligned} |U(g(y)) - F(y)| &\leq |U(g(y)) - P(x^*, y)| + |P(x^*, y) - F(y)| \leq \\ &\leq Cd(y)^r + Ed(y)^r = (C + E)d(y)^r \,. \end{aligned}$$

Hence,

$$\begin{array}{lcl} \frac{U(g(y))}{F(y)} &=& \frac{U(g(y)) - F(y) + F(y)}{F(y)} \leq \\ &\leq& 1 + \frac{|U(g(y)) - F(y)|}{F(y)} \leq \\ &\leq& 1 + \frac{(C+E)d(y)^r}{Cd(y)^r} \leq \\ &\leq& 2 + \frac{E}{C} =: \tilde{K} \,. \end{array}$$

This proves property vi) and concludes the proof of the proposition.

108

Chapter 3

Dynamics and Symplectic geometry

3.1 Symplectic geometry of the phase space

The main goal of this chapter is to highlight some relations between the dynamics of the system and the underlying symplectic geometry of the phase space. As we have recalled in the addendum at the end of section 1.1, the cotangent bundle T^*M may be equipped with a canonical symplectic structure

$$\omega = \sum_{i=1}^d dx_i \wedge dp_i,$$

where $(\mathcal{U}, x_1, \ldots, x_d)$ is a local coordinate chart for M and $(T^*\mathcal{U}, x_1, \ldots, x_d, p_1, \ldots, p_d)$ the associated cotangent coordinates. It is easy to show that ω is a symplectic form (i.e., it is non-degenerate and closed). In particular, one can check that ω is intrinsically defined (i.e., independent of the choice of the coordinate charts), by considering the 1-form on $T^*\mathcal{U}$

$$\lambda = \sum_{i=1}^d p_i \, dx_i \, ,$$

which satisfies $\omega = -d\lambda$ and is coordinate-independent; in fact, in terms of the natural projection

$$\begin{array}{rccc} \pi: \mathrm{T}^*M & \longrightarrow & M \\ & (x,p) & \longmapsto & x \end{array}$$

the form λ may be equivalently defined pointwise by

$$\lambda_{(x,p)} = (d\pi_{(x,p)})^* p \in \mathcal{T}^*_{(x,p)}\mathcal{T}^*M.$$
(3.1)

The 1-form λ is called the *Liouville form* (or the *tautological form*).

We also remarked that, since ω is non-degenerate and closed, the Hamiltonian vector field X_H is uniquely determined by:

$$\omega\left(X_H(x,p),\cdot\right) = d_x H(\cdot) \,.$$

One can easily check that, in local coordinates, the above equation is equivalent to Hamilton's equation (1.2).

A distinguished role in the study of the geometry of a symplectic space is played by the so-called *Lagrangian submanifolds*.

Definition 3.1.1 (Lagrangian submanifolds). Let Λ be a d-dimensional C^1 submanifold of (T^*M, ω) . We say that Λ is Lagrangian if for any $(x, p) \in \Lambda$, $T_{(x,p)}\Lambda$ is a Lagrangian subspace, i.e., $\omega|_{T_{(x,p)}\Lambda} = 0$.

We will mainly be concerned with Lagrangian graphs, that is Lagrangian manifolds $\Lambda \subset T^*M$ such that $\Lambda = \{(x, \eta(x)), x \in M\}$. In (T^*M, ω) there exists an interesting well-known relation between Lagrangian graphs and closed 1-forms (recall that a 1-form can be interpreted as a section of T^*M). **Proposition 3.1.2.** Let $\Lambda = \{(x, \eta(x)), x \in M\}$ be a smooth section of T^*M . Λ is Lagrangian if and only if η is a closed 1-form.

Proof. Let us consider:

$$s_{\eta}: M \longrightarrow \mathrm{T}^*M$$

 $x \longmapsto (x, \eta(x)).$

We want to prove first that $s_{\eta}^* \lambda = \eta$, where λ is the tautological form introduced above and $s_{\eta}^* \lambda$ denotes its pull-back. Recalling that $\lambda(x, p) = (d\pi(x, p))^* p$, we get:

$$(s_{\eta}^{*}\lambda)(x) = (ds_{\eta}(x))^{*}\lambda(x,p) = (ds_{\eta}(x))^{*}(d\pi(x,p))^{*}\eta(x) =$$
$$= d(\pi \circ s_{\eta}(x,p))^{*}\eta(x) = \eta(x) ,$$

where in the last equality we used that $\pi \circ s_{\eta}$ is the identity map. Using this property, the claim follows immediately. In fact:

$$\begin{split} \Lambda \text{ is Lagrangian } & \Longleftrightarrow \quad \omega \big|_{\mathrm{TA}} = 0 \quad \Longleftrightarrow \quad s_{\eta}^* \omega = 0 \quad \Longleftrightarrow \quad s_{\eta}^* d\lambda = 0 \\ & \longleftrightarrow \quad ds_{\eta}^* \lambda = 0 \quad \Longleftrightarrow \quad d\eta = 0 \quad \Longleftrightarrow \quad \eta \text{ is closed.} \end{split}$$

In the light of this relation, one can define the *cohomology class* (or *Liouville class*) of Λ to be the cohomology class of the closed 1-form representing it.

Remark 3.1.3. Observe that this cohomology class can be defined in a more intrinsic way; in fact, the projection

$$\pi_{|\Lambda_L} : \Lambda_L \subset \mathrm{T}^*M \longrightarrow M$$

induces an isomorphism between the cohomology groups $H^1(M; \mathbb{R})$ and $H^1(\Lambda_L; \mathbb{R})$.

The preimage of $[\lambda_{|\Lambda_L}]$ under this isomorphism is called the *Liouville class* of Λ_L and one can easily check that these two definitions coincide.

This characterization motivates the following extension of the notion of Lagrangian graph to the continuous case.

Definition 3.1.4 (Continuous Lagrangian graphs). A continuous section Λ of T^*M is a C^0 -Lagrangian graph if it locally coincides with the graph of an exact differential. As above, one can define its cohomology class.

Let us now consider a Hamiltonian $H : T^*M \longrightarrow \mathbb{R}$ and Φ_t^H the associated Hamiltonian flow. We want to analyze the relation between the property of being Lagrangian and the dynamics on it.

Proposition 3.1.5. Let Λ be a Lagrangian submanifold. Then Λ is invariant if and only if $H|_{\Lambda} \equiv \text{const.}$

Proof. $[\Longrightarrow]$ The Hamiltonian vector field X_H is defined by $\omega(X_H, \cdot) = dH$. Since Λ is invariant, $X_H|_{\Lambda}$ is tangent to Λ . But Λ is Lagrangian, therefore $0 = \omega(X_H, V) = dH \cdot V$ for any $V \in T\Lambda$, and this implies that H is constant on Λ . [$\Leftarrow=$] Since H is constant on Λ , we have that $0 = dH \cdot V = \omega(X_H, V)$, for every $V \in T\Lambda$. Since Λ is Lagrangian, X_H belongs to $T\Lambda$ itself and therefore Λ is invariant. \Box

Remark 3.1.6. It follows immediately from the above property that invariant Lagrangian graphs correspond to η -critical subsolutions of Hamilton-Jacobi equation, for some 1-form η , whose cohomology class coincides with that of Λ . Therefore, if Λ is an invariant Lagrangian graph, then the value of the constant in proposition 3.1.5 is given by $\alpha(c_{\Lambda})$, where α is the α -function associated to H and c_{Λ} the cohomology class of Λ . See also remark 1.4.7.

In the next sections we will study more in depth the minimizing properties of Lagrangian graphs and deduce some uniqueness results within a fixed cohomology or homology class (see section 3.3). Before concluding this section, let us state and prove a remarkable observation due to Michael Herman [42] for the case $M = \mathbb{T}^d$. We will use it in section 3.4 to discuss some features of *KAM tori*.

Proposition 3.1.7. Given a Hamiltonian H, let $\mathcal{T} \subset \mathbb{T}^d \times \mathbb{R}^d$ be an invariant graph over \mathbb{T}^d such that the Hamiltonian flow on \mathcal{T} is conjugated to a flow R_t on \mathbb{T}^d , which is transitive, i.e., with a dense orbit. Then \mathcal{T} is Lagrangian.

Proof. Let $\varphi : \mathbb{T}^d \to \mathcal{T}$ be the conjugation: $\varphi^{-1} \circ \Phi_t^H \circ \varphi = R_t, \forall t \in \mathbb{R}$. Consider the inclusion $i_{\mathcal{T}}$ of \mathcal{T} into $\mathbb{T}^d \times \mathbb{R}^d$. We want to prove that $\omega|_{\mathcal{T}} = i_{\mathcal{T}}^* \omega \equiv 0$. Let us start by proving that the restriction of the symplectic form $i_{\mathcal{T}}^* \omega$ is invariant under the Hamiltonian flow Φ_t^H . In fact,

$$\Phi_t^{H^*}(i_{\mathcal{T}}^*\omega) = \left(i_{\mathcal{T}} \circ \Phi_t^H\right)^* \omega = \left(\Phi_t^H \circ i_{\mathcal{T}}\right)^* \omega = i_{\mathcal{T}}^* \left(\Phi_t^{H^*}\omega\right) = i_{\mathcal{T}}^*\omega,$$

where we used that \mathcal{T} is invariant $(\Phi_t^H \circ i_{\mathcal{T}} = i_{\mathcal{T}} \circ \Phi_t^H)$ and Φ_t^H is a symplectomorphism for any $t \in \mathbb{R}$ (*i.e.*, $\Phi_t^{H^*} \omega = \omega$). Consider now the 2-form on \mathbb{T}^d given by $\omega_1 = \varphi^*(i_{\mathcal{T}}^*\omega)$. Let us show that ω_1 is invariant under R_t ; in fact,

$$(R_t)^*\omega_1 = (R_t)^* \left(\varphi^* \left(i_T^*\omega\right)\right) = \left(\varphi \circ R_t\right)^* i_T^*\omega = \left(\Phi_t^H \circ \varphi\right)^* i_T^*\omega =$$
$$= \varphi^* \left(\Phi_t^{H^*} \left(i_T^*\omega\right)\right) = \varphi^* \left(i_T^*\omega\right) = \omega_1,$$

where we used that $\varphi^{-1} \circ \Phi_t^H \circ \varphi = R_t$. Since R_t is transitive, then ω_1 invariant implies ω_1 constant: $\omega_1 = \sum_{i < j} a_{ij} dx_i \wedge dx_j$. But ω_1 is exact (since $\omega = -d\lambda$ is exact), therefore $\omega_1 = \varphi^*(i_T^*\omega) \equiv 0$ and (using that φ is invertible) the result is proved.

3.2 Symplectic aspects of the Aubry set and the quotient Aubry set

In this section we want to discuss purely symplectic definitions of the Aubry set and the quotient Aubry set, which follow quite easily from weak KAM theory approach. In fact, reinterpreting the notion of critical subsolutions from a more geometric prospect, one can deduce many symplectic properties of these sets; in particular we will deduce that they are "invariant" under exact symplectomorphism. Similar results have also been discussed in [15, 69].

As we have seen in section 1.4, the Aubry set \mathcal{A}_c^* can be equivalently defined as:

$$\mathcal{A}_{c}^{*} = \bigcap_{u \in \mathcal{S}_{\eta}^{1}} \operatorname{Graph}(\eta + du)$$
(3.2)

where S_{η}^{1} denotes the set of η -critical subsolutions (see theorem 1.4.19 and (1.22)).

Let us give the following definition.

Definition 3.2.1 (Subcritical Lagrangian graphs). Given a Lagrangian graph Λ with Liouville class c, we will say that Λ is c-subcritical, or simply subcritical, if $\Lambda \subset \{(x,p) \in T^*M : H(x,p) \leq \alpha(c)\}, \text{ where } \alpha : H^1(M;\mathbb{R}) \longrightarrow \mathbb{R} \text{ is Mather's}$ α -function. Given a subcritical Lagrangian graph Λ with Liouville class c, we will call $\Lambda_{crit} = \{(x,p) \in \Lambda : H(x,p) = \alpha(c)\}$ its critical part.

Observe that c-critical Lagrangian graphs correspond to η -critical subsolutions, for any η closed 1-form with cohomology c. Therefore, we can deduce from weak KAM theory (in particular theorem 1.4.24) that they do exist and that Lipschitz graphs are "dense" (in the sense of theorem 1.4.24). In terms of these Lagrangian graphs, (3.2) becomes:

$$\mathcal{A}_{c}^{*} = \bigcap_{c-\text{subcritical }\Lambda} \Lambda = \bigcap_{c-\text{subcritical }\Lambda} \Lambda_{crit}$$
(3.3)

where the intersections are over the set of all *c*-critical Lagrangian graphs. In particular, there will exist a *c*-critical Lagrangian graph $\tilde{\Lambda}$ such that $\mathcal{A}_c^* = \tilde{\Lambda}_{crit} = \tilde{\Lambda} \cap \mathcal{E}_c^*$, where \mathcal{E}_c^* is Mañé critical energy level (see (1.23)). Observe that, as discussed in [69], this property can be also interpreted as a "non-removable intersection property" for such particular Lagrangian graphs.

In terms of c-critical Lagrangian graphs one can also get the following geometric characterization for the Mañé critical energy level \mathcal{E}_c^* and, consequently, for $\alpha(c)$: it is the only energy level of H such that the bounded region that it "encloses" does not contain any c-Lagrangian graphs in its interior, while each of its open neighborhoods does. In particular, it follows from the existence of critically dominated functions (see remark 1.4.5) that there exist continuous (even Lipschitz) c-Lagrangian graphs contained in the closure of this region bounded by \mathcal{E}_c^* : these are exactly what we called c-critical Lagrangian graphs. Moreover, it is immediate from the above definitions that if any energy level of the Hamiltonian contains a c-Lagrangian graph, then this energy level is the Mañé c-critical one: in fact any higher energy level will bound a region containing in its interior a c-Lagrangian graph.

Remark 3.2.2. Observe that the above characterizations provide definitions for the Aubry set and the Mañé critical energy level, which do not depend on the form of the Hamiltonian, except for its energy level. Therefore, if H' is another Hamiltonian that has \mathcal{E}_c^* as energy level, then \mathcal{E}_c^* will be its Mañé *c*-critical energy level. Moreover, it follows from (3.3) that H' will also have the same Aubry set \mathcal{A}_c^* . These properties are not surprising. In fact, it is easy to show that on a given energy level, the flows of all Hamiltonians that possess it as an energy level, is the same up to time-reparameterization.

Now, we want to show how the above concepts allow one to get also a geometric definition of the quotient Aubry set; this is not discussed in [69]. Let us start with

Mather's pseudodistance δ_c defined in (1.14). For any $x, y \in \mathcal{A}_c$ and Λ_1, Λ_2 *c*-critical Lagrangian graphs, let us consider a closed path $\lambda_{x,y}^{\Lambda_1,\Lambda_2}$ obtained by joining x to y on Λ_1 and then going back to x staying on Λ_2 . Using (1.25) it is immediate to check that:

$$\delta_c(x,y) = \sup_{\Lambda_1,\Lambda_2 \atop c - \text{critical}} \int_{\lambda_{x,y}^{\Lambda_1,\Lambda_2}} \lambda$$
(3.4)

where λ is the Liouville form defined in (3.1) and the integral does not depend on the chosen path $\lambda_{x,y}^{\Lambda_1,\Lambda_2}$. As a consequence of this definition, one gets a characterization of *c*-static classes and, therefore, of the quotient Aubry set $\bar{\mathcal{A}}_c$. From a geometric point of view, two points in the same *c*-static class are points for which the integral $\int_{\lambda_{x,y}^{\Lambda_1,\Lambda_2}} \lambda$ is zero for all possible loops constructed as above. It would be interesting to understand if this property has a geometric meaning too; Floer homology might provide the useful ground for tackling this question.

Let us now deduce, from these characterizations, the following result concerning the invariance of the Aubry set, the quotient Aubry set and the Mañé critical energy level under exact symplectomorphisms. Let us first recall that a diffeomorphism $\Psi : T^*M \longrightarrow T^*M$ is a symplectomorphism if it preserves the symplectic form ω , *i.e.*, $\Psi_*\omega = \omega$; in particular this is equivalent to say that $\Psi_*\lambda - \lambda$ is a closed 1-form. We will say that Ψ is an *exact symplectomorphism* if $\Psi_*\lambda - \lambda$ is exact.

Theorem 3.2.3. Let H be an optical Hamiltonian on T^*M and let $\Psi : T^*M \longrightarrow T^*M$ be an exact symplectomorphism. Consider the new Hamiltonian $H' = H \circ \Psi^{-1}$. Then for all $c \in H^1(M; \mathbb{R})$: ¹

- $\mathcal{E}_c^{*\prime} = \Psi(\mathcal{E}_c^*)$ and therefore $\alpha'(c) = \alpha(c);$
- $\mathcal{A}_c^{*\prime} = \Psi(\mathcal{A}_c^*);$

¹We will indicate with a *prime* all quantities associated to H' (*e.g.*, $\mathcal{E}_c^{*'}, \mathcal{A}', \delta_c'$, etc...)

δ_c(x, y) = δ'_c(Ψ(x), Ψ(y)); therefore it maps c-static classes of H' into c-static classes of H and the induced map Ψ : Ā_c → Ā'_c is an isometry.

A similar result has also been proven, with different techniques, by Patrick Bernard in [15].

Proof. It is enough to observe that exact symplectomorphisms transform *c*-Lagrangian graphs into *c*-Lagrangian graphs. The proof will then easily follow from (3.3), (3.4) and the characterization of Mañé *c*-critical energy level. \Box

3.3 Minimizing properties of Lagrangian graphs

In this section we will analyze the minimizing properties of invariant measures supported on Lagrangian graphs and deduce some uniqueness results for invariant Lagrangian graphs within a fixed cohomology or homology class (we will clearify what we mean by homology class later in this section; see definition 3.3.5).

An interesting result is the following characterization of minimizing measures, introduced in section 1.2.

Lemma 3.3.1. Let μ be an invariant probability measure on TM and $\mu^* = \mathcal{L}_*\mu$ its push-forward to T^*M , via the Legendre transform \mathcal{L} . Then, μ is a Mather's measure if and only if $\operatorname{supp} \mu^*$ is contained in the critical part of a subcritical Lagrangian graph. In particular, any invariant probability measure μ^* on T^*M , whose support is contained in an invariant Lagrangian graph with Liouville class c, is the image, via the Legendre transform, of a c-action minimizing measure.

Proof. (i) If μ is a Mather's measure with cohomology class c, then the support of μ^* is contained in $\mathcal{L}(\widetilde{\mathcal{M}}_c) \subseteq \mathcal{A}_c^*$. By (1.22), \mathcal{A}_c^* is given by the intersection of all c-subcritical Lagrangian graphs, so $\operatorname{supp} \mu^*$ is contained in at least one c-subcritical Lagrangian graph Λ . In particular $\operatorname{supp} \mu^*$ is contained in the critical part of Λ ,

simply because $\mathcal{L}(\widetilde{\mathcal{M}}_c)$ is in the energy level $\mathcal{E}_c^* = \{(x, p) \in \mathrm{T}^*M : H(x, p) = \alpha(c)\},$ see Theorem 1.2.7.

(*ii*) Let us fix η to be a closed 1-form with $[\eta] = c$. Since we are assuming that Λ is a *c*-subcritical Lagrangian graph, we can write $\Lambda = \{(x, \eta(x) + du(x)), x \in M\}$, where $u : M \to \mathbb{R}$ is C^1 . By Theorem 1.3.17, in order to show that μ is a *c*-action minimizing measure, it is enough to show that $\sup \mu \subseteq \widetilde{\mathcal{N}}_c$, *i.e.*, that any orbit in $\sup \mu$ is a *c*minimizer. To this purpose, let us consider $(x, v) \in \operatorname{supp} \mu$ and let $\gamma(t) \equiv \pi(\Phi_t(x, v))$, where Φ_t is the Euler-Lagrange flow and π the canonical projection on M. Given any interval $[a, b] \subset \mathbb{R}$, let us consider the difference $u(\gamma(b)) - u(\gamma(a))$ and rewrite it as:

$$u(\gamma(b)) - u(\gamma(a)) = \int_{a}^{b} d_{\gamma(s)} u(\gamma(s)) \dot{\gamma}(s) ds =$$

=
$$\int_{a}^{b} \left[L_{\eta}(\gamma(s), \dot{\gamma}(s)) + H_{\eta}(\gamma(s), d_{\gamma(s)}u) \right] ds, \qquad (3.5)$$

where the second equality follows from the definition of the Hamiltonian as the Legendre-Fenchel transform of the Lagrangian and the fact that $\gamma(s)$ is an orbit of the Euler-Lagrange flow. Note that along the orbit $H_{\eta}(\gamma(s), d_{\gamma(s)}u) = \alpha(c)$, because $\operatorname{supp} \mu$ is invariant and $\operatorname{supp} \mu^*$ is in the critical part of Λ . Then

$$\int_{a}^{b} L_{\eta}(\gamma(s), \dot{\gamma}(s)) ds = u(\gamma(b)) - u(\gamma(a)) - \alpha(c)(b-a) .$$
(3.6)

On the other hand, any other curve $\gamma_1 : [a, b] \to M$ such that $\gamma_1(a) = \gamma(a)$ and $\gamma_1(b) = \gamma(b)$ satisfies:

$$u(\gamma(b)) - u(\gamma(a)) = \int_{a}^{b} d_{\gamma_{1}(s)} u(\gamma_{1}(s)) \dot{\gamma}(s) \, ds \leq \\ \leq \int_{a}^{b} \left[L_{\eta}(\gamma_{1}(s), \dot{\gamma}_{1}(s)) + H_{\eta}(\gamma_{1}(s), d_{\gamma_{1}(s)}u) \right] ds \qquad (3.7)$$

where the second inequality follows again by the duality between Hamiltonian and

Lagrangian. Note that now $H_{\eta}(\gamma_1(s), d_{\gamma_1(s)}u) \leq \alpha(c)$, because $\Lambda = \{(x, \eta(x) + du(x))\}$ is subcritical. Then

$$\int_{a}^{b} L_{\eta}(\gamma_{1}(s), \dot{\gamma}_{1}(s)) ds \ge u(\gamma(b)) - u(\gamma(a)) - \alpha(c)(b-a)$$
(3.8)

and this proves that γ is a *c*-minimizer (see also proposition 1.4.9).

Let us finally observe that the Hamiltonian is constant on any invariant Lagrangian graph $\Lambda = \{(x, \eta + du)\}, i.e., H(x, \eta + du) = k$ (see proposition 3.1.5). Then u is a classical solution of the Hamilton-Jacobi equation with cohomology class c. As explained in remark 3.1.6, $k = \alpha(c)$ and this shows that Λ coincides with its critical part. By the result proved in item (*ii*), if μ^* is supported on Λ , then μ is a c-action minimizing measure and this proves the last claim in the statement of the lemma.

Remark 3.3.2. Note that the fact that the orbits on an invariant Lagrangian graph are action-minimizing can be also deduced from a classical result by Weierstrass, as already pointed out by Jürgen Moser (see remark in [57]). In fact, Weierstrass method or the use of the Hamilton-Jacobi Equation (that we are using in our proof) are essentially two sides of the same coin.

This result easily implies an already-known uniqueness result for Lagrangian graphs supporting invariant measures of full support, in a fixed cohomology class (see also [57], in which a different proof is presented).

Theorem 3.3.3. If $\Lambda \subset T^*M$ is a Lagrangian graph on which the Hamiltonian dynamics admits an invariant measure μ^* with full support, then $\Lambda = \mathcal{L}(\widetilde{\mathcal{M}}_c) = \mathcal{A}_c^*$, where c is the cohomology class of Λ . Therefore, if Λ_1 and Λ_2 are two Lagrangian graphs as above, with the same cohomology class, then $\Lambda_1 = \Lambda_2$. In other words, for any given $c \in H^1(M; \mathbb{R})$, there exists at most one invariant Lagrangian graph Λ with cohomology class c, that carries an invariant measure whose support is the whole of Λ .

Proof. By Lemma 3.3.1, the measure $\mu = \mathcal{L}_*^{-1}\mu^*$ is *c*-minimizing. This means that $\mathcal{L}^{-1}(\Lambda) = \operatorname{supp} \mu \subseteq \widetilde{\mathcal{M}}_c \subseteq \widetilde{\mathcal{A}}_c$, where the last inclusion follows from Theorem 1.3.17. Note however that, by Theorems 1.2.7 and 1.3.16, $\widetilde{\mathcal{M}}_c$ and $\widetilde{\mathcal{A}}_c$ are graphs over their bases and, since $\operatorname{supp} \mu$ is a graph over the whole M, it follows that

$$\mathcal{L}^{-1}(\Lambda) = \operatorname{supp} \mu = \widetilde{\mathcal{M}}_c = \widetilde{\mathcal{A}}_c \,. \tag{3.9}$$

One can deduce something more from the above proof.

Theorem 3.3.4. If Λ and μ are as in Theorem 3.3.3 and ρ is the rotation vector of $\mu = \mathcal{L}^{-1}\mu^*$, then $\Lambda = \mathcal{L}(\widetilde{\mathcal{M}}^{\rho})$. Therefore, if Λ_1 and Λ_2 are two Lagrangian graphs supporting measures of full support and the same rotation vector ρ , then $\Lambda_1 = \Lambda_2$. Moreover, Mather's β -function is differentiable at ρ with $\partial\beta(\rho) = c$, where c is the cohomology class of Λ .

Proof. The first claim follows from the fact that $\widetilde{\mathcal{M}}^{\rho}$ is a graph over M and that by definition $\widetilde{\mathcal{M}}^{\rho} \supseteq$ supp $\mu = \mathcal{L}^{-1}(\Lambda)$. As far as the differentiability of β at ρ is concerned, suppose that $c' \in \mathrm{H}^1(M; \mathbb{R})$ is a subderivative of β at ρ . Hence, $\beta(\rho) = \langle c', \rho \rangle - \alpha(c')$ and this implies that $\mathfrak{M}^{\rho} \subseteq \mathfrak{M}_{c'}$; in fact, for any $\mu \in \mathfrak{M}^{\rho}$:

$$\int_{\mathrm{T}M} (L - \hat{\eta}') \, d\mu = \int_{\mathrm{T}M} L \, d\mu - \int_{\mathrm{T}M} \hat{\eta}' \, d\mu = \beta(\rho) - \langle c', \rho \rangle = -\alpha(c') \,,$$

where η' is a closed 1-form of cohomology c'. As a result, $\widetilde{\mathcal{M}}^{\rho} = \mathcal{L}^{-1}(\Lambda) \subseteq \widetilde{\mathcal{M}}_{c'}$. The graph property of $\widetilde{\mathcal{M}}_{c'}$ and of $\widetilde{\mathcal{A}}_{c'}$ implies that $\widetilde{\mathcal{A}}_{c'} = \widetilde{\mathcal{M}}_{c'} = \mathcal{L}^{-1}(\Lambda)$ and $\mathcal{A}^*_{c'} = \Lambda$. As a consequence, c' = c. In fact, by Theorem 1.4.24 and (1.22), there exists a C^1 function $v : \mathcal{M} \longrightarrow \mathbb{R}$, such that $\Lambda = \{(x, \eta' + dv) : x \in \mathcal{M}\}$, where η' is a closed 1-form with $[\eta'] = c'$. This, by definition, means that c' is the cohomology class of Λ and therefore c' = c.

Now, recall that to any invariant probability measure μ on TM one can associate an element $\rho(\mu)$ of the homology group $H_1(M; \mathbb{R})$, known as *rotation vector* or *Schwartzman asymptotic cycle* (see sections 1.2 and 3.5). This allows us to define the homology class of certain invariant Lagrangian graphs.

Definition 3.3.5 (Schwartzman uniquely/strictly ergodic Lagrangian graphs).

A Lagrangian graph Λ is called Schwartzman uniquely ergodic if all invariant measures supported on Λ have the same rotation vector ρ , which will be called homology class of Λ . Moreover, if there exists an invariant measure with full support, Λ will be called Schwarztman strictly ergodic.

See section 3.5 for a more detailed description of such flows. We can now state and prove the following uniqueness result.

Theorem 3.3.6 ([34]). Let Λ be a Schwartzman strictly ergodic invariant Lagrangian graph with homology class ρ . The following properties are satisfied:

- (i) if $\Lambda \cap \mathcal{A}_c^* \neq \emptyset$, then $\Lambda = \mathcal{A}_c^*$ and $c = c_\Lambda$, where c_Λ is the cohomology class of Λ .
- (ii) the Mather function α is differentiable at c_{Λ} and $\partial \alpha(c_{\Lambda}) = \rho$.

Therefore,

- (iii) any invariant Lagrangian graph that carries a measure with rotation vector ρ is equal to the graph Λ;
- (iv) any invariant Lagrangian graph is either disjoint from Λ or equal to Λ .

Proof. (*i*) From Theorem 3.3.3, it follows that $\Lambda = \mathcal{A}_{c_{\Lambda}}^*$. Let us show that it does not intersect any other Aubry set. Suppose by contradiction that Λ intersects another

Aubry set \mathcal{A}_c^* . By Theorem 3.3.4, $\Lambda = \mathcal{L}^{-1}\left(\widetilde{\mathcal{M}}^{\rho}\right)$, then $\widetilde{\mathcal{M}}^{\rho} \cap \widetilde{\mathcal{A}}_c \neq \emptyset$ and, because of the Lemma 1.3.18 and the graph property of $\widetilde{\mathcal{A}}_c$, we can conclude that $\mathcal{A}_c^* = \Lambda$. The same argument used in the proof of Theorem 3.3.4 allows us to conclude that $c = c_{\Lambda}$. (*ii*) Suppose that $h \in \partial \alpha(c_{\Lambda})$. The previous lemma implies that $\widetilde{\mathcal{M}}^h \subseteq \Lambda$; the Schwartzman unique ergodicity property of Λ implies $h = \rho$. Therefore α is differentiable at c_{Λ} and $\partial \alpha(c_{\Lambda}) = \rho$.

To prove (iv), let Λ_1 be an invariant Lagrangian graph and call c_1 its cohomology class. If the compact invariant set $\Lambda \cap \Lambda_1$ is not empty, then we can find a probability measure μ^* invariant under the flow and whose support is contained in this intersection. Since μ^* is contained in the Lagrangian graph Λ_1 , by Lemma 3.3.1, it is c_1 -minimizing. Hence, the support of μ^* is contained in $\mathcal{A}_{c_1}^*$. This shows that the intersection $\Lambda \cap \mathcal{A}_{c_1}^*$ contains the support of μ^* and is therefore not empty. By (i), $\Lambda = \mathcal{A}_{c_1}^*$. Moreover, note that $\mathcal{A}_{c_1}^* \subseteq \Lambda_1$, because, on the one hand, $\Lambda_1 = \text{Graph}(\eta_1 + du_1)$, with $[\eta_1] = c_1$ and u_1 a classical solution to the Hamilton-Jacobi equation (see proof of Lemma 3.3.1), and, on the other hand, $\mathcal{A}_{c_1}^* = \bigcap_{u \in S_{\eta}^1} \text{Graph}(\eta_1 + du_1)$, see (1.22). Therefore, $\Lambda = \Lambda_1$, since they are both graphs over the base.

To prove (*iii*), consider an invariant Lagrangian graph Λ_1 , with cohomology class c_1 , which carries an invariant measure μ^* whose rotation vector is ρ . By Lemma 3.3.1, the measure μ^* is c_1 -minimizing. Therefore, we have $\widetilde{\mathcal{M}}^{\rho} \cap \widetilde{\mathcal{A}}_{c_1} \neq \emptyset$. By Proposition 1.3.18, it follows that $\mathcal{L}^{-1}(\Lambda) = \widetilde{\mathcal{M}}^{\rho} \subseteq \widetilde{\mathcal{A}}_{c_1} \subseteq \mathcal{L}^{-1}(\Lambda_1)$. Again, this forces the equality $\Lambda = \Lambda_1$ because of the graph property.

Finally, observe that Lemma 1.2.6, Theorem 1.3.17 and Proposition 1.3.18 imply the following property.

Corollary 3.3.7. Mather's α function is differentiable at c if and only if the restriction of the Euler-Lagrange flow to $\widetilde{\mathcal{A}}_c$ is Schwartzman uniquely ergodic (see section 3.5 for the definition and a discussion of Schwartzman ergodic flows).

3.4 Global uniqueness of KAM tori and Herman's tori

In the case $M = \mathbb{T}^d$, it is natural to ask for the implications of the results in section 3.3 for KAM theory. In this section we will use these results to discuss the problem of uniqueness of KAM tori and, more generally, of the invariant tori belonging to the closure of the set of KAM tori (or *Herman's Tori*).

Let us start by recalling what we mean by KAM torus (recall that in this case the homology group $H_1(\mathbb{T}^d; \mathbb{R})$ can be canonically identified with \mathbb{R}^d).

Definition 3.4.1 (KAM Torus). $\mathcal{T} \subset \mathbb{T}^d \times \mathbb{R}^d$ is a (maximal) KAM torus with rotation vector ρ if:

i) $\mathcal{T} \subset \mathbb{T}^d \times \mathbb{R}^d$ is a continuous graph over \mathbb{T}^d ;

ii) \mathcal{T} is invariant under the Hamiltonian flow Φ_t^H generated by H;

iii) the Hamiltonian flow on \mathcal{T} is conjugated to a uniform rotation on \mathbb{T}^d ; i.e., there exists a diffeomorphism $\varphi : \mathbb{T}^d \to \mathcal{T}$ such that $\varphi^{-1} \circ \Phi_t^H \circ \varphi = R_{\rho}^t, \forall t \in \mathbb{R}$, where $R_{\rho}^t : x \to x + \rho t$.

KAM theory concerns with the existence of KAM tori for quasi-integrable Hamiltonian systems of the form $H(x, p) = H_0(p) + \varepsilon f(x, p)$, where: (x, p) are local coordinates on $\mathbb{T}^d \times \mathbb{R}^d$, ε is a "small" parameter and f(x, p) a smooth function. If $\varepsilon = 0$, the system is integrable, in the sense that the dynamics can be explicitly solved: in particular each torus $\mathbb{T}^d \times \{p_0\}$ is invariant and the motion on it corresponds to a rotation with frequency $\rho(p_0) = \frac{\partial H_0}{\partial p}(p_0)$. The question addressed by KAM theory is whether this foliation of the phase space into invariant tori, on which the motion is quasi-periodic, persists when $\varepsilon \neq 0$. In 1954 Kolmogorov [44] proved (and Arnol'd [2] and Moser [66, 67] reproved it later in different contexts and with different techniques) that, in spite of the generic disappearence of the invariant submanifolds filled by periodic orbits, as already pointed out by Poincaré, for small ε it is always possible to find KAM tori corresponding to "strongly non-resonant", i.e., Diophantine, rotation vectors. Let us recall here the definition and some properties of Diophantine vectors:

Definition 3.4.2. Given $\gamma, \tau > 0$, we say that $\rho \in \mathbb{R}^d$ is a (γ, τ) -Diophantine vector if and only if $|\langle \rho, \nu \rangle| \geq \gamma |\nu|^{-\tau}$, $\forall \nu \in \mathbb{Z}^d \setminus \{0\}$.

Remark 3.4.3. The set of (γ, τ) -Diophantine vectors will be denoted by $\mathcal{D}(\gamma, \tau)$. Note that, if $\tau < d-1$, $\mathcal{D}(\gamma, \tau) = \emptyset$, while for $\tau > d-1$, the Diophantine vectors have full measure in \mathbb{R}^d , that is $\lim_{R\to\infty} \mu_0 \left(\bigcup_{\gamma>0} \mathcal{D}(\gamma, \tau) \cap B_R\right) / \mu_0(B_R) = 1$, where μ_0 is the Lesbegue measure and B_R is the ball of radius R centered at 0; for $\tau = d-1$, $\bigcup_{\gamma>0} \mathcal{D}(\gamma, \tau)$ has measure zero but Hausdorff dimension d.

The celebrated KAM Theorem (in one of its several versions) not only shows the existence of such tori, but also provides an explicit method to construct them.

Theorem 3.4.4 (Kolmogorov–Arnol'd–Moser, [73]). Let $d \ge 2$, $\tau > d - 1$, $\gamma > 0$, $\ell > 2\tau + 2$, M > 0 and r > 0 be given. Let $B_r \in \mathbb{R}^d$ be the open ball of radius r centered at the origin. Let $H \in C^{\ell}(\mathbb{T}^d \times B_r)$ be of the form

$$H(x,p) = H_0(p) + \varepsilon f(x,p) \tag{3.10}$$

with $|H_0|_{C^{\ell}} \leq M$, $|f|_{C^{\ell}} \leq M$, $\left|\frac{\partial^2 H_0}{\partial p^2}\right| \geq M^{-1}$ and $\rho = \frac{\partial H_0}{\partial p}(0) \in \mathcal{D}(\gamma, \tau)$. Then, for any $s < \ell - 2\tau - 1$, there exists $\varepsilon_0 > 0$ such that for any $\varepsilon \leq \varepsilon_0$ the Hamiltonian (3.10) admits a $C^{s,s+\tau}$ KAM torus with rotation vector ρ , i.e., a $C^{s+\tau}$ invariant torus such that the Hamiltonian flow on it is C^s -conjugated with a rotation with frequency ρ .

Remark 3.4.5. 1) If $H \in C^{\infty}$ then the KAM torus mentioned in the theorem above is C^{∞} . If H is real analytic then the KAM torus is real analytic.

2) As already mentioned, the proof of this theorem is constructive and it actually

contains much more information than those summarized in the above statement. For instance, in the analytic case, the proof consists in an iterative method allowing one to construct order by order the series defining the conjugation function (and to prove convergence of the formal series). In the differentiable case the proof is based on the idea of iteratively approximating differentiable functions by analytic ones and then using the inductive approximation scheme of the analytic case. However, although these proofs provide an explicit construction of a KAM torus , these construction depends a priori on a number of arbitrary choices that one has to make along the proof – e.g., the choice of cutoffs one needs to introduce in the iterative approximation scheme.

3) The invariant torus constructed in the proof of the KAM theorem is locally unique, in the sense that for any prescribed (and admissible) s there is at most one $C^{s,s+\tau}$ KAM torus with rotation vector ρ within a C^s -distance $\delta(d, s, C, \tau)$ to the one constructed in the proof of the KAM Theorem, see [17, 72, 73]. Note that the C^s -distance δ within which one can prove uniqueness of the KAM torus in a prescribed regularity class depends both on the irrationality properties of ρ and on the regularity class s itself. It is then a priori possible that even for small ε there exist different KAM tori, within a prescribed C^1 -distance from the one constructed in the proof of the theorem, possibly less regular than that torus. Quite surprisingly, even in the analytic case, we are not aware of any proof of "global" uniqueness of the invariant analytic torus one manages to construct is unique within the class of analytic tori – however nothing a priori guarantees that less regular invariant tori with the same rotation vector exist.

The question arisen in remark (3) is our main motivation for the study of the problem of global uniqueness of KAM tori. Corollary 3.4.6 below settles the question and shows that, at least in the case of optical Hamiltonians, it is not possible to have two different KAM tori with the same rotation vector. Note that the assumption of strict convexity of the Hamiltonian is necessary to exclude trivial sources of non-uniqueness: for instance, in the context of quasi-integrable Hamiltonians, global uniqueness could be lost simply because the unperturbed Hamiltonian induces a map $p \rightarrow \partial_p H_0(p)$ from actions to frequencies that is not one to one. Let us also remark that, apparently, the Hamiltonian considered in KAM theorem is not a Tonelli Hamiltonian, since the latter, by definition, is defined globally on the whole $\mathbb{T}^d \times \mathbb{R}^d$. However any C^ℓ strictly convex Hamiltonian defined on $\mathbb{T}^d \times B_r$ for some r > 0 can be extended to a global C^ℓ optical Hamiltonian. Then in the statement of the KAM Theorem above it is actually enough to assume H to be a C^ℓ optical Hamiltonian, locally satisfying the (in)equalities listed after (3.10).

Given the proof of proposition 3.3.6, it follows the following corollary.

Corollary 3.4.6 (Global uniqueness of KAM tori, [34]). Every optical Hamiltonian H on T^*T^d possesses at most one Lagrangian KAM torus for any given rotation vector ρ . In particular, if H and ρ satisfy the assumptions of the KAM Theorem, then there exists one and only one KAM torus with rotation vector ρ .

Proof. Since the Lagrangian KAM torus \mathcal{T} admits an invariant measure μ^* of full support, which is the image via the conjugation φ of the uniform measure on \mathbb{T}^d , then the claims follow from Theorem 3.3.4. Note that for rationally independent rotation vectors, Herman's remark in proposition 3.1.7, implies that \mathcal{T} is automatically Lagrangian.

An interesting generalization of the result of Corollary 3.4.6 concerns the invariant tori belonging to the C^0 -closure $\overline{\Upsilon}$ of the set Υ of all Lagrangian KAM tori. Note that for quasi-integrable systems, Υ is not empty. The set Υ can be seen as a subset of Lip(\mathbb{T}^d , \mathbb{R}^d). This follows from Theorem 3.3.4 and from Mather's graph theorem, see Theorems 1.2.7, 1.2.9, and the results in [57]. Moreover, any family of invariant Lagrangians graphs on which the function α (or H) is bounded gives rise to a family of functions in $\operatorname{Lip}(\mathbb{T}^d, \mathbb{R}^d)$ with uniformly bounded Lipschitz constant (see section 1.4). By Ascoli-Arzelà theorem, it follows that $\overline{\Upsilon}$ is also a subset of $\operatorname{Lip}(\mathbb{T}^d, \mathbb{R}^d)$, consisting of functions whose graphs are invariant C^0 -Lagrangian tori. Michael Herman [43] showed that, for a generic Hamiltonian H close enough to an integrable Hamiltonian H_0 , the dynamics on the generic tori in $\overline{\Upsilon}$ is not conjugated to a rotation. These "new" tori therefore represent the majority, in the sense of topology, and hence most invariant tori cannot be obtained by the KAM algorithm. More precisely, Herman showed that in $\overline{\Upsilon}$ there exists a dense G_{δ} set (*i.e.*, a dense countable intersection of open sets) of invariant Lagrangian graphs on which the dynamics is strictly ergodic and weakly mixing, and for which the rotation vector is not Diophantine. These invariant graphs are therefore not obtained by the KAM theorem, however our uniqueness result do still apply to these graphs since strict ergodicity implies Schwartzman strict ergodicity.

More generally, given any Tonelli Lagrangian on \mathbb{T}^d , we consider the set $\tilde{\Upsilon}$ of invariant Lagrangian graphs on which the dynamics of the flow is topologically conjugated to an *ergodic* linear flow on \mathbb{T}^d (of course, far from the canonical integrable Lagrangian the set $\tilde{\Upsilon}$ may be empty). The dynamics on any of the invariant graphs in $\tilde{\Upsilon}$ is strictly ergodic. Since the set of strictly ergodic flows on a compact set is a G_{δ} in the C^0 topology, see for example [35, Corollaire 4.5], it follows that there exists a dense G_{δ} subset \mathcal{G} of the C^0 closure of $\tilde{\Upsilon}$ in $\operatorname{Lip}(\mathbb{T}^d, \mathbb{R}^d)$, such that the dynamics on any $\Lambda \in \mathcal{G}$ is strictly ergodic. Therefore we have the following proposition.

Proposition 3.4.7 (Uniqueness of Herman's Tori, [34]). There exists a dense G_{δ} set \mathcal{G} in the C^0 closure of $\tilde{\Upsilon}$ consisting of strictly ergodic invariant Lagrangian graphs. Any $\Lambda \in \mathcal{G}$ satisfies the following properties:

- (i) the invariant graph Λ has a well-defined rotation vector $\rho(\Lambda)$.
- (ii) Any invariant Lagrangian graph that intersects Λ coincides with Λ .

 (iii) Any Lagrangian invariant graph that carries an invariant measure whose rotation is ρ(Λ) coincides with Λ.

3.5 Schwartzman unique and strict ergodicity

In Section 1.2 we have introduced the concept of rotation vector of a measure. This is closely related to the notion of Schwartzman asymptotic cycle of a flow, introduced by Sol Schwartzman in [74], as a first attempt to develop an algebraic topological approach to the study of dynamics. In this section, we would like to provide some examples and investigate some properties of what we have called *Schwartzman uniquely ergodic flows* (see section 3.4).

First, we would like to discuss more in depth the concept of rotation vector and Schwartzman asymptotic cycle. One can provide a different description of the Schwartzman asymptotic cycle of a flow. This is also known as the flux homomorphism in volume preserving and symplectic geometry, see [9][Chapter 3]. We will use the description given in [26][pages 67-70]. This definition has the technical advantadge of not relying on the Krylov-Bogolioubov theory of generic orbits in a dynamical system, although a more geometrical definition showing that "averaged" pieces of long orbits converge almost everywhere in the first homology group for any invariant measure is certainly more heuristic and intuitive.

Let us start with some standard facts. As usual we set $\mathbb{T} = \mathbb{R}/\mathbb{Z}$. The space \mathbb{T} is a topological group for the addition. An important feature of \mathbb{T} is that the canonical projection $\pi : \mathbb{R} \to \mathbb{T}$ is a covering map. Therefore given any continuous path $\gamma : [a, b] \to \mathbb{T}$, with $a \leq b$, we can find a continuous lift $\bar{\gamma} : [a, b] \to \mathbb{R}$ such that $\gamma = \pi \bar{\gamma}$. Any two such lifts differ by an integer. It follows that the quantity $\bar{\gamma}(b) - \bar{\gamma}(a)$ does not depend on the lift. We will set

$$\mathcal{V}(\gamma) = \bar{\gamma}(b) - \bar{\gamma}(a) \in \mathbb{R}.$$

This quantity remains constant on the homotopy class, with fixed end-points, of the path γ . Moreover, if γ is a closed path, *i.e.*, we have $\gamma(a) = \gamma(b)$ then $\mathcal{V}(\gamma) \in \mathbb{Z}$, and, if such a closed path is homotopic to 0 (with fixed end-points) then $\mathcal{V}(\gamma) = 0$. It is also clear that for $c \in [a, b]$, we have

$$\mathcal{V}(\gamma|[a,b]) = \mathcal{V}(\gamma|[a,c]) + \mathcal{V}(\gamma|[c,b]).$$

Note also that two continuous paths $\gamma_1, \gamma_2 : [a, b] \to \mathbb{T}$ can be added by the formula

$$(\gamma_1 + \gamma_2)(t) = \gamma_1(t) + \gamma_2(t).$$

For this addition we have

$$\mathcal{V}(\gamma_1 + \gamma_2) = \mathcal{V}(\gamma_1) + \mathcal{V}(\gamma_2).$$

Another important property of the map \mathcal{V} is its continuity on the functional space $C^0([a, b], \mathbb{T})$, endowed with the topology of uniform convergence. Let θ be the closed 1-form on \mathbb{T} whose lift to \mathbb{R} is the usual differential form dt on \mathbb{R} , where dt is the differential of the identity map $\mathbb{R} \to \mathbb{R}, t \mapsto t$. It is well-known that when $\gamma : [a, b] \to \mathbb{T}$ is C^1 , we have

$$\mathcal{V}(\gamma) = \int_{\gamma} \theta = \int_{a}^{b} \theta_{\gamma(t)}(\dot{\gamma}(t)) dt$$

If X is a topological space and $F: X \times [a, b] \to \mathbb{T}$ is a given map, we will define $\mathcal{V}(F): X \to \mathbb{R}$ by

$$\forall x \in X, \ \mathcal{V}(F)(x) = \mathcal{V}(F_x),$$

where $F_x : [a, b] \to \mathbb{T}$ is defined by $F_x(t) = F(x, t)$. The continuity of \mathcal{V} on $C^0([a, b], \mathbb{T})$ implies that $\mathcal{V}(F)$ is continuous. Furthermore, the continuity of \mathcal{V} on $C^0([a, b], \mathbb{T})$ also implies that the map $C^0(X \times [a, b], \mathbb{T}) \to C^0(X, \mathbb{T}), F \mapsto \mathcal{V}(F)$ is continuous, when we provide the spaces of continuous maps with the compact open topology.

If F can be lifted to a continuous map $\overline{F}: X \times [a, b] \to \mathbb{R}$ with $F = \pi \overline{F}$, then

$$\mathcal{V}(F)(x) = \bar{F}(x,b) - \bar{F}(x,a).$$

Suppose now that X is a topological space and that $(\phi_t)_{t\in\mathbb{R}}$ is a continuous flow on X. We will define $\Phi: X \times [0,1] \to X$ by $\Phi(x,t) = \phi_t(x)$. If $f: X \to \mathbb{T}$ is continuous, we set

$$\mathcal{V}(f,\phi_t) = \mathcal{V}(f \circ \Phi) : X \to \mathbb{R}.$$

There is another another way to define $\mathcal{V}(f, \phi_t)$ which is used in [26]. The function $F(f, \Phi) : X \times [0, 1] \to \mathbb{T}$, defined by

$$F(f,\Phi)(x,t) = f(\phi_t(x)) - f(x)$$

is continuous and identically 0 on $X \times \{0\}$, it is therefore homotopic to a constant and can be lifted to a continuous map $F(f, \Phi) : X \times [0, 1] \to \mathbb{R}$, with $F(f, \Phi) | X \times \{0\}$ identically 0. We have

$$\mathcal{V}(f,\phi_t)(x) = F(f,\Phi)(x,1).$$

Note that if f is homotopic to 0 then it can be lifted continuously to $\bar{f}: X \to \mathbb{R}$. In that case $\bar{F}(f, \Phi) = \bar{f}\Phi - \bar{f}$ and

$$\mathcal{V}(f,\phi_t)(x) = \bar{f}(\phi_1(x)) - \bar{f}.$$

If μ is a measure with compact support and invariant under the flow ϕ_t , for a continuous $f : X \to \mathbb{T}$, we can define $\mathcal{S}(\mu, \phi_t)(f)$, or simply $\mathcal{S}(\mu)(f)$ when ϕ_t is fixed, by

$$\mathcal{S}(\mu)(f) = \int_X \mathcal{V}(f, \phi_t)(x) \, d\mu(x).$$

It is not difficult to verify that for $f_1, f_2: X \to \mathbb{T}$, we have

$$\mathcal{S}(\mu)(f_1 + f_2) = \mathcal{S}(\mu)(f_1) + \mathcal{S}(\mu)(f_2).$$

Moreover, if $f: X \to \mathbb{T}$ is homotopic to 0, it can be lifted to $\overline{f}: X \to \mathbb{R}$ and

$$\mathcal{S}(\mu)(f) = \int_X [\bar{f}(\phi_1(x)) - \bar{f}(x)] \, d\mu(x) = \int_X \bar{f}(\phi_1(x)) \, d\mu(x) - \int_X \bar{f}(x) \, d\mu(x) = 0,$$

since μ is invariant by ϕ_1 . Therefore, if we denote by $[X, \mathbb{T}]$ the set of homotopy classes of continuous maps from X to \mathbb{T} , which is an additive group, the map $\mathcal{S}(\mu)$ is a well-defined additive homomorphism from the additive group $[X, \mathbb{T}]$ to \mathbb{R} . When X is a good space (like a manifold or a locally finite polyhedron), it is well-known that $[X, \mathbb{T}]$ is canonically identified with the first cohomology group $\mathrm{H}^1(X; \mathbb{Z})$. In that case $\mathcal{S}(\mu)$ is in $\mathrm{Hom}(\mathrm{H}^1(X; \mathbb{Z}), \mathbb{R})$. Since the first cohomology group with real coefficients $\mathrm{H}^1(X; \mathbb{R})$ is $\mathrm{H}^1(X; \mathbb{Z}) \otimes \mathbb{R}$, we can view $\mathcal{S}(\mu)$ as an element of the dual $\mathrm{H}^1(X; \mathbb{R})^*$ of the \mathbb{R} -vector space $\mathrm{H}^1(X; \mathbb{R})$. When $\mathrm{H}^1(X; \mathbb{R})$ is finite-dimensional then $\mathrm{H}^1(X; \mathbb{R})^*$ is in fact equal to the first homology group $\mathrm{H}_1(X; \mathbb{R})$, and therefore $\mathcal{S}(\mu)$ defines an element of $\mathrm{H}_1(X; \mathbb{R})$, *i.e.*, a 1-cycle. This 1-cycle $\mathcal{S}(\mu)$ is called the *Schwartzman asymptotic cycle* of μ . Note that $\mathrm{H}^1(X; \mathbb{R})$ is finite dimensional when X is a finite polyhedron or a compact manifold. It should be also noted that for a manifold M the projection $TM \to M$ is a homotopy equivalence. Therefore $\mathrm{H}^1(TM; \mathbb{R}) = \mathrm{H}^1(M; \mathbb{R})$ is finite dimensional when M is a compact manifold.

We want now to study the behavior of Schwartzman asymptotic cycles under semi-conjugacy.

Proposition 3.5.1. Suppose that $\phi_t^i : X_i \to X_i, i = 1, 2$ are two continuous flows. Suppose also that $\psi : X_1 \to X_2$ is a continuous semi-conjugation between the flows, *i.e.*, $\psi \circ \phi_t^1 = \phi_t^2 \circ \psi$, for every $t \in \mathbb{R}$. Given a probability measure μ with compact support on X_1 invariant under ϕ_t^1 , then, for every continuous map $f : X_2 \to \mathbb{T}$, we have

$$\mathcal{S}(\psi_*\mu,\phi_t^2)([f]) = \mathcal{S}(\mu,\phi_t^1)([f\circ\psi]),$$

where $\psi_*\mu$ is the image of μ under ψ . In particular, if we are in the situation where Hom $([X_i, \mathbb{T}]) \equiv H_1(X_i; \mathbb{R}), i = 1, 2$, we obtain

$$\mathcal{S}(\psi_*\mu, \phi_t^2) = \mathrm{H}_1(\psi)(\mathcal{S}(\mu, \phi_t^1)).$$

Proof. Notice that $f\psi\phi_t^1(x) - f\psi(x) = f\phi_t^2(\psi(x)) - f(\psi(x))$. Therefore by uniqueness of the liftings, $\mathcal{V}(f\psi,\phi_t^1)(x) = \mathcal{V}(f,\phi_t^2)(\psi(x))$. An integration with respect to μ finishes the proof.

We would like now to relate the Schwartzman asymptotic cycles to the rotation vectors $\rho(\mu)$ defined for Lagrangian flows. We first consider the case of a C^1 flow ϕ_t on the manifold N. We call X the continuous vector field on N generating ϕ_t , *i.e.*,

$$\forall x \in N, \quad X(x) = \frac{d\phi_t(x)}{dt}\Big|_{t=0}$$

By the flow property $\phi_{t+t'} = \phi_t \circ \phi_{t'}$, this implies

$$\forall x \in N, \ \forall t \in \mathbb{R}, \quad \frac{d\phi_t(x)}{dt} = X(\phi_t(x)).$$

In the case of a manifold N, the identification of $[N, \mathbb{T}]$ with $\mathrm{H}^1(N; \mathbb{Z})$ is best described with the de Rham cohomology. We consider the natural map $I_N : [X, \mathbb{T}] \to$ $\mathrm{H}^1(X; \mathbb{R})$ defined by

$$I_N([f]) = [f^*\theta],$$

where [f] on the left hand side denotes the homotopy class of the C^{∞} map $f: N \to \mathbb{T}$, and $[f^*\theta]$ on the right hand side is the cohomology class of the pullback by f of the closed 1-form on \mathbb{T} whose lift to \mathbb{R} is dt. Note that any homotopy class in $[N, \mathbb{T}]$ contains smooth maps because C^{∞} maps are dense in C^0 maps (for the Whitney topology). Therefore the map I_N is indeed defined on the whole of $[N, \mathbb{T}]$. As it is well-known, this map I_N induces an isomorphism of $[N, \mathbb{T}]$ on $\mathrm{H}^1(N; \mathbb{Z}) \subset \mathrm{H}^1(N; \mathbb{R}) =$ $\mathrm{H}^1(N; \mathbb{Z}) \otimes \mathbb{R}$.

Given a C^{∞} map $f : N \to \mathbb{T}$, the C^1 flow ϕ_t on N, and $x \in N$, we compute $\mathcal{V}(f, \phi_t)(x)$. If $\gamma_x : [0, 1] \to N$ is the path $t \mapsto \phi_t(x)$, by definition, we have $\mathcal{V}(f, \phi_t)(x) = \mathcal{V}(f \circ \gamma_x)$. Since γ_x is C^1 , we get

$$\mathcal{V}(f,\phi_t)(x) = \int_{f \circ \gamma_x} \theta = \int_{\gamma_x} f^* \theta.$$

Since $\gamma_x(t) = \phi_t(x)$, we have $\dot{\gamma}_x(t) = X(\phi_t(x))$. It follows that

$$\mathcal{V}(f,\phi_t)(x) = \int_0^1 (f^*\theta)_{\phi_t(x)}(X[\phi_t(x)]) \, dt$$

Recall that the interior product $i_X \omega$ of a differential form ω with X is given by

$$(i_X\omega)_x(\cdots) = \omega_x(X(x),\cdots).$$

When ω is a differential 1-form then $i_X \omega$ is a function. With this notation, we get

$$\mathcal{V}(f,\phi_t)(x) = \int_0^1 (i_X f^*\theta)(\phi_t(x)) \, dt.$$

Therefore if μ is an invariant measure for ϕ_t , which we will assume to have a compact support, we obtain

$$\mathcal{S}(\mu) = \int_N \int_0^1 (i_X f^* \theta)(\phi_t(x)) \, dt d\mu(x).$$

Since $i_X f^* \theta$ is continuous, and we are assuming that μ has a compact support, we have

$$\mathcal{S}(\mu) = \int_0^1 \int_N (i_X f^*\theta)(\phi_t(x)) \, d\mu(x) dt.$$

By invariance of μ under ϕ_t , we get $\int_N (i_X f^* \theta)(\phi_t(x)) d\mu(x) = \int_N (i_X f^* \theta)(x) d\mu(x)$, and therefore

$$\mathcal{S}(\mu) = \int_0^1 \int_N (i_X f^*\theta)(x) \, d\mu(x) dt = \int_N (i_X f^*\theta) \, d\mu.$$

This shows that as an element of $H^1(M; \mathbb{R})^*$, the Schwarztman asymptotic cycle $\mathcal{S}(\mu)$ is given by

$$\mathcal{S}(\mu)([\omega]) = \int_N i_X \omega \, d\mu.$$

We can now easily compute Schwartzman asymptotic cycles for linear flows on \mathbb{T}^d . Such a flow is determined by a constant vector field $\alpha \in \mathbb{R}^d$ on \mathbb{T}^d (here we use the canonical trivialisation of the tangent bundle of \mathbb{T}^d), the associated flow $R_t^{\alpha} : \mathbb{T}^d \to \mathbb{T}^d$ is defined by $R_t^{\alpha}(x) = x + [t\alpha]$, where $[t\alpha]$ is the class in $\mathbb{T}^d = \mathbb{R}^d/\mathbb{Z}^d$ of the vector $t\alpha \in \mathbb{R}^d$. If ω is a 1-form with constant coefficients, *i.e.*, $\omega = \sum_{i=1}^d a_i dx_i$, with $a_i \in \mathbb{R}$, the interior product $i_{\alpha}\omega$ is the constant function $\sum_{i=1}^d \alpha_i a_i$. Therefore, it follows that $\mathcal{S}(\mu) = \alpha \in \mathbb{R}^d \equiv \mathrm{H}_1(\mathbb{T}^d; \mathbb{R})$.

We now compute Schwartzman asymptotic cycles for Euler-Lagrange flows. In this case N = TM and ϕ_t is an Euler-Lagrange flow ϕ_t^L of some Lagrangian L. If we call X_L the vector field generating ϕ_t^L , since this flow is obtained from a second order ODE on M, we get

$$\forall x \in M, \forall v \in T_x M, \quad T\pi(X_L(x, v)) = v,$$

where $T\pi : T(TM) \to TM$ denotes the canonical projection. Since this projection π is a homotopy equivalence, to compute $S(\mu)$ we only need to consider forms of the type $\pi^*\omega$ where ω is a closed 1-form on the base M. In this case $(i_{X_L}\pi^*\omega)(x,v) = \omega_x(T\pi(X_L(x,v))) = \omega_x(v)$. Therefore, for any probability measure μ on TM with compact support and invariant under ϕ_t^L , we obtain

$$\mathcal{S}(\mu)[\pi^*\omega] = \int_{TM} \omega_x(v) \, d\mu(x,v) = \int_{TM} \hat{\omega} \, d\mu.$$

This is precisely $\rho(\mu)$ as it was defined in section 1.2. Note that the only property we have used is the fact that ϕ_t is the flow of a second order ODE on the base M.

To simplify things, in the remainder of this section, we will assume that X is a compact space, for which we have $[X, \mathbb{T}] = \mathrm{H}^1(X; \mathbb{Z})$, and $\mathrm{H}_1(X; \mathbb{Z})$ is finitely generated. In that case, the dual space $\mathrm{H}^1(X; \mathbb{R})^*$ is $\mathrm{H}_1(X; \mathbb{R})$, and for every flow ϕ_t on X and every probability measure μ on X invariant under ϕ_t , the Schwartzman asymptotic cycle is an element of the finite dimensional-vector space $\mathrm{H}_1(X; \mathbb{R})$.

Suppose that x is a periodic point of ϕ_t or period T > 0. One can define an invariant probability measure μ_{x,t_0} for ϕ_t by

$$\int_X g(x) \, d\mu_{x,t_0} = \frac{1}{t_0} \int_0^{t_0} g((\phi_t(x)) \, dt,$$

where $g: X \to \mathbb{R}$ is a measurable function. It is easy to verify that $\mathcal{S}(\mu_{x,t_0})$ is equal in $H_1(X;\mathbb{R})$ to the homology class $[\gamma_{x,t_0}]/t_0$, where γ_{x,t_0} is the loop $t \mapsto \phi_t(x), t \in [0, t_0]$. When x is a fixed point of ϕ_t , then the Dirac mass δ_x at x is invariant under ϕ_t , and in that case $\mathcal{S}(\delta_x) = 0$.

Definition 3.5.2. For a flow ϕ_t on X, we denote by $\mathcal{S}(\phi_t)$ the set of all Scwartzman asymptotic cycles $\mathcal{S}(\mu)$, where μ is an arbitrary probability measure on X invariant under ϕ_t .

Since X is compact, note that for the weak topology the set $\mathfrak{M}(X)$ of probability Borel measures on X is compact and convex. It is even metrizable, since we are assuming X metrizable. Furthermore the subset $\mathfrak{M}(X, \phi_t) \subseteq \mathfrak{M}(X)$ of probability measures invariant under ϕ_t is, as it is well-known, compact convex and non-empty. Therefore $\mathcal{S}(\phi_t)$ is a compact convex non-empty subset of $H_1(X;\mathbb{R})$.

For the case of a linear flow R^{α} on \mathbb{T}^d , we have shown above that $\mathcal{S}(R_t^{\alpha}) = \{\alpha\} \subset \mathbb{R}^d \equiv \mathrm{H}_1(\mathbb{T}^d; \mathbb{R}).$

The following corollary is an easy consequence of Proposition 3.5.1.

Corollary 3.5.3. For i = 1, 2, suppose that ϕ_t^i is a continuous flow on the compact space X_i , which satisfies $\operatorname{Hom}([X_i, \mathbb{T}], \mathbb{R}) \equiv \operatorname{H}_1(X_i; \mathbb{R})$. If $\psi : X_1 \to X_2$ is a topological conjugacy between ϕ_t^1 and ϕ_t^2 (i.e., the map ψ is a homeomorphism that satisfies $\psi \phi_t^1 = \phi_t^2 \psi$, for all $t \in \mathbb{R}$), then we have

$$\mathcal{S}(\phi_t^2) = \mathrm{H}_1(\psi)[\mathcal{S}(\phi_t^1)].$$

We denote by $\mathfrak{F}(X)$ the set of continuous flows on X. We can embed $\mathfrak{F}(X)$ in $C^0(X \times [0,1], X)$ by the map $\phi_t \mapsto F^{\phi_t} \in C^0(X \times [0,1], X)$, where

$$F^{\phi_t}(x,t) = \phi_t(x).$$

The topology on $C^0(X \times [0,1], X)$ is the compact open (or uniform) topology, and we endow $\mathfrak{F}(X)$ with the topology inherited from the embedding given above.

Lemma 3.5.4. The map $\phi_t \mapsto \mathcal{S}(\phi_t)$ is upper semi-continuous on $\mathfrak{F}(X)$. This means that for each open subset $U \subseteq H_1(X; \mathbb{R})$, the set $\{\phi_t \in \mathfrak{F}(X) \mid \mathcal{S}(\phi_t) \subset U\}$ is open in $\mathfrak{F}(X)$.

Proof. Since the topology on $C^0(X \times [0,1], X)$ is metrizable, if this were not true we could find an open set $U \subset H_1(X; \mathbb{R})$ and a sequence ϕ_t^n of continuous flows on X converging uniformly to a flow ϕ_t , with $\mathcal{S}(\phi_t) \subset U$, and $\mathcal{S}(\phi_t^n)$ is not contained in U. This means that for each n we can find a probability measure μ_n on X invariant under ϕ_t^n and such that its Schwartzman asymptotic cycle $\mathcal{S}(\mu_n, \phi_t^n)$ for ϕ_n^t is not in the open set U. Since $\mathfrak{M}(X)$ is compact for the weak topology, extracting a subsequence if necessary, we can assume that $\mu_n \to \mu$. It is not difficult to show that μ is invariant under the flow ϕ_t . We now show that $\mathcal{S}(\mu_n, \phi_t^n) \to \mathcal{S}(\mu, \phi_t)$. This will yield a contradiction and finish the proof because $\mathcal{S}(\mu_n, \phi_t^n)$ is in the closed set $H_1(X; \mathbb{R}) \setminus U$, for every n, and $\mathcal{S}(\mu, \phi_t) \in U$.

To show that the linear maps $\mathcal{S}(\mu_n, \phi_t^n) \in \mathrm{H}_1(X; \mathbb{R}) = \mathrm{H}^1(X; \mathbb{R})^*$ converge to the linear map $\mathcal{S}(\mu, \phi_t)$, it suffices to show that $\mathcal{S}(\mu_n, \phi_t^n)([f]) \to \mathcal{S}(\mu, \phi_t)([f])$, for every $[f] \in [X, \mathbb{T}] = \mathrm{H}^1(X; \mathbb{Z}) \subset \mathrm{H}^1(X; \mathbb{R}) = \mathrm{H}^1(X; \mathbb{Z}) \otimes \mathbb{R}$. Fix now a continuous map $f: X \to \mathbb{T}$. Denote by $F_n, F: X \times [0, 1] \to \mathbb{T}$ the maps defined by

$$F_n(x,t) = f(\phi_t^n(x)) - f(x)$$
 and $F(x,t) = f(\phi_t(x)) - f(x)$

By the uniform continuity of f on the compact metric space X, the sequence F_n converges uniformly to F. Since $F_n | X \times \{0\} \equiv 0$, if we call $\tilde{F}_n : X \times [0,1] \to \mathbb{R}$ the lift of F_n such that $\tilde{F}_n | X \times \{0\} \equiv 0$, then the sequence \tilde{F}_n also converges uniformly to \tilde{F} , that is the lift of F such that $\tilde{F} | X \times \{0\} \equiv 0$. Since the μ_n are probability measures, we have

$$\left| \int_X \tilde{F}_n(x,1)\mu_n(x) - \int_X \tilde{F}(x,1)\mu_n(x) \right| \le \|\tilde{F}_n - \tilde{F}\|_{\infty} \longrightarrow 0.$$

Since $\mu_n \to \mu$ weakly, we also have

$$\left| \int_X \tilde{F}(x,1)\mu_n(x) - \int_X \tilde{F}(x,1)\mu(x) \right| \longrightarrow 0.$$

Therefore $\mathcal{S}(\mu_n, \phi_t^n)([f]) = \int_X \tilde{F}_n(x, 1)\mu_n(x) \to \mathcal{S}(\mu, \phi_t)([f]) = \int_X \tilde{F}(x, 1)\mu_0(x).$

Definition 3.5.5. [Schwartzman unique ergodicity] We say that a flow ϕ_t is Schwartzman uniquely ergodic if $S(\phi_t)$ is reduced to one point.

By the computation done above linear flows on the torus \mathbb{T}^d are Schwartzman uniquely ergodic. Of course, all uniquely ergodic flows (*i.e.*, flows having exactly one

invariant probability measure) are also Schwartzman uniquely ergodic. Moreover, by Corollary 3.5.3, any flow topologically conjugate to a Schwartzman uniquely ergodic flow is itself Schwartzman uniquely ergodic.

Theorem 3.5.6. The set $\mathfrak{S}(N)$ of Schwartzman uniquely ergodic flows is a G_{δ} in $\mathfrak{F}(X)$.

Proof. Fix some norm on $\mathrm{H}^1(X; \mathbb{R})$. We will measure diameters of subsets of $\mathrm{H}^1(X; \mathbb{R})$ with respect to that norm. Fix $\epsilon > 0$. Call \mathcal{U}_{ϵ} the set of flows ϕ_t such that the diameter of $\mathcal{S}(\phi_t) \subset \mathrm{H}_1(X; \mathbb{R})$ is $< \epsilon$. If $\phi_t^0 \in \mathcal{U}_{\epsilon}$, we can find U an open subset of $\mathrm{H}_1(X; \mathbb{R})$ of diameter $< \epsilon$ and containing $\mathcal{S}(\phi_t^0)$. By the lemma above the set $\{\phi_t \in \mathfrak{F}(X) \mid \mathcal{S}(\phi_t) \subset U\}$ is open in $\mathfrak{F}(X)$, contains ϕ_t^0 and is contained in \mathcal{U}_{ϵ} . The set of Schwartzman uniquely ergodic flows is then $\bigcap_{n\geq 1} \mathcal{U}_{1/n}$.

Proposition 3.5.7. Let $\phi_t : X \longrightarrow X$ be a continuous flow on the compact path connected space X. Suppose that there exist $t_i \uparrow +\infty$ such that $\phi_{t_i} \longrightarrow \phi$ in C(X, X)(with the C⁰-topology). Then, ϕ_t is Schwartzman uniquely ergodic. In particular, periodic flows and (uniformly) recurrent flows are Schwartzman uniquely ergodic (in both cases $\phi = \mathrm{Id}$).

Proof. Fix a continuous map $f: X \to \mathbb{T}$. Consider the function $F: X \times [0, +\infty) \to \mathbb{T}, (x,t) \mapsto f(\phi_t(x)) - f(x)$. We have F(x,0) = 0, for every $x \in X$. Call $\overline{F}: X \times [0, +\infty) \to \mathbb{R}$ the (unique) continuous lift of F such that $\overline{F}(x,0) = 0$, for every $x \in X$. The definition of the Schwartzman asymptotic cycle gives

$$\mathcal{S}(\mu)([f]) = \int_X \bar{F}(x,1) \, d\mu(x),$$

for every probability measure invariant under ϕ_t . We claim that we have

$$\forall t, t' \ge 0, \forall x \in X, \quad \bar{F}(x, t+t') = \bar{F}(\phi_t(x), t') + F(x, t).$$

In fact, if we fix t and we consider each side of the equality above as a (continuous) function of (x, t') with values in \mathbb{R} , we see that the two sides are equal for t' = 0, and that they both lift the function

$$(x,t') \mapsto f(\phi_{t+t'}(x)) - f(x) = f(\phi'_t(\phi_t(x)) - f(\phi_t(x)) + f(\phi_t(x)) - f(x))$$

with values in \mathbb{T} . By induction, it follows easily that

$$\forall k \in \mathbb{N}, \quad \bar{F}(x,k) = \sum_{j=0}^{k-1} \bar{F}(\phi_j(x),1).$$

Therefore, if $t \ge 0$ and [t] is its integer part, we also obtain

$$\bar{F}(x,t) = \bar{F}(\phi_{[t]}(x), t - [t]) + \sum_{j=0}^{[t]-1} \bar{F}(\phi_j(x), 1).$$
(3.11)

It follows that

$$\forall t \ge 0, \forall x \in X, \quad |\bar{F}(x,t)| \le ([t]+1) \|\bar{F}|X \times [0,1]\|_{\infty}.$$
 (3.12)

By compactness $\|\bar{F}|X \times [0,1]\|_{\infty}$ is finite. If we integrate equality (3.11) with respect to a probability measure μ on X invariant under the flow ϕ_t , we obtain

$$\int_X \bar{F}(x,t) \, d\mu(x) = \int_X \bar{F}(x,t-[t]) \, d\mu(x) + [t] \int_X \bar{F}(x,1) \, d\mu(x) + [t] \int_X \bar{F}(x,1) \, d\mu(x) \, d\mu(x) + [t] \int_X \bar{F}(x,1) \, d\mu(x) \, d\mu(x) + [t] \int_X \bar{F}(x,1) \, d\mu(x) \, d\mu(x) \, d\mu(x) + [t] \int_X \bar{F}(x,1) \, d\mu(x) \, d\mu(x$$

Therefore we have

$$\mathcal{S}(\mu)([f]) = \lim_{t \to +\infty} \int_X \frac{\bar{F}(x,t)}{t} \, d\mu(x). \tag{3.13}$$

Suppose now that we set $\gamma_x(s) = \phi_s(x)$; we have $\overline{F}(x,t) = \mathcal{V}(f\gamma_x|[0,t])$. Fix now some point $x_0 \in X$, and consider $t_i \to +\infty$ such that $\phi_{t_i} \to \phi$ in the C^0 topology. Since $\overline{F}(x_0,t)/t$ is bounded in absolute value by $2\|\overline{F}|X \times [0,1]\|_{\infty}$, for $t \ge 1$, extracting a subsequence if necessary, we can assume that $\overline{F}(x_0,t_i)/t_i \to c \in \mathbb{R}$. If $x \in X$, we can find a continuous path $\gamma : [0,1] \to M$ with $\gamma(0) = x_0$ and $\gamma(1) = x$. The map $\Gamma : [0,1] \times [0,t] \to \mathbb{T}, (s,s') \to \phi_{s'}(\gamma(s))$ is continuous, therefore we can lift it to a continuous function with values in \mathbb{R} , and this implies the equality

$$\mathcal{V}(\Gamma|[0,1] \times \{0\}) + \mathcal{V}(\Gamma|\{1\} \times [0,t]) - \mathcal{V}(\Gamma|[0,1] \times \{1\}) - \mathcal{V}(\Gamma|\{0\} \times [0,t]) = 0.$$

This can be rewritten as

$$\mathcal{V}(f\gamma_x|[0,t]) - \mathcal{V}(f\gamma_{x_0}|[0,t]) = \mathcal{V}(f\phi_t\gamma) - \mathcal{V}(f\gamma),$$

which translates to

$$\bar{F}(x,t) - \bar{F}(x_0,t) = \mathcal{V}(f\phi_t\gamma) - \mathcal{V}(f\gamma).$$

Since $\phi_{t_i} \to \phi$ uniformly, by continuity of \mathcal{V} , the left hand-side remains bounded as $t = t_i \to +\infty$. It follows that $(\bar{F}(x, t_i) - \bar{F}(x_0, t_i))/t_i \to 0$. Hence for every $x \in X$, we also have that $\bar{F}(x, t_i)/t_i$ tends to the same limit c as $\bar{F}(x_0, t_i)/t_i$. Since, by (3.12), $\bar{F}(x, t)/t$ is uniformly bounded for $t \geq 1$, by Lebesgue's dominated convergence we obtain from (3.13) that $\mathcal{S}(\mu)([f]) = c$, where c is independent of the invariant measure μ . This is of course true for any $f : X \to \mathbb{T}$. Therefore $\mathcal{S}(\mu)$ does not depend on the invariant measure μ .

An interesting property of Schwartzman uniquely ergodic flows (which also shows that they have some kind of rigidity) is the following proposition, that follows immediately from the definition of Schwartzman unique ergodicity and what we remarked above about the asymptotic cycles of fixed and periodic points (see also [74]). **Proposition 3.5.8.** Suppose that ϕ_t is a Schwartzman uniquely ergodic flow on X. If there exists either a fixed point or a closed orbit homologous to zero, then all closed orbits are homologous to zero. In the remaining case, if C_1 and C_2 are closed orbits with periods τ_1 and τ_2 , then $\frac{C_1}{\tau_1}$ and $\frac{C_2}{\tau_2}$ are homologous. Since $[C_1]$ and $[C_2]$ are in $H_1(X;\mathbb{Z})$, it follows in this case that the ratio of the periods of any two closed orbits must be rational. Consequently, for any continuous family of periodic orbits of ϕ_t , all orbits have the same period.

Definition 3.5.9. [Schwartzman strict ergodicity] We say that a flow ϕ_t is Schwartzman strictly ergodic if it is Schwartzman uniquely ergodic and it has an invariant measure μ of full support (i.e., $\mu(U) > 0$ for every non-empty open subset U of X).

Linear flows on the torus \mathbb{T}^d are Schwartzman strictly ergodic (they preserve Lebesgue measure). Of course, all strictly ergodic flows (*i.e.*, flows having exactly one invariant probability measure, and the support of this measure is full) are also Schwartzman strictly ergodic. A minimal flow which is Schwartzman uniquely ergodic is in fact Schwartzman strictly ergodic (because all invariant measures have full support). Moreover, any flow topologically conjugate to a Schwartzman strictly ergodic flow is also Schwartzman strictly ergodic.

Bibliography

- Ralph Abraham and Joel Robbin. Transversal mappings and flows. An appendix by Al Kelley. W. A. Benjamin, Inc., New York-Amsterdam, 1967.
- [2] Vladimir I. Arnol'd. Proof of a theorem of A. N. Kolmogorov on the preservation of conditionally periodic motions under a small perturbation of the Hamiltonian. Uspehi Mat. Nauk, 18(5 (113)):13–40, 1963.
- [3] Vladimir I. Arnol'd. Instability of dynamical systems with many degrees of freedom. Dokl. Akad. Nauk SSSR, 156:9–12, 1964.
- [4] Patrice Assouad. Plongements lipschitziens dans Rⁿ. Bull. Soc. Math. France, 111(4):429–448, 1983.
- [5] Serge Aubry. The twist map, the extended Frenkel-Kontorova model and the devil's staircase. *Phys. D*, 7(1-3):240–258, 1983. Order in chaos (Los Alamos, N.M., 1982).
- [6] Serge Aubry and P. Y. Le Daeron. The discrete Frenkel-Kontorova model and its extensions. I. Exact results for the ground-states. *Phys. D*, 8(3):381–422, 1983.
- [7] J. M. Ball and V. J. Mizel. One-dimensional variational problems whose minimizers do not satisfy the Euler-Lagrange equation. Arch. Rational Mech. Anal., 90(4):325–388, 1985.

- [8] Victor Bangert. Mather sets for twist maps and geodesics on tori. In Dynamics reported, Vol. 1, volume 1 of Dynam. Report. Ser. Dynam. Systems Appl., pages 1–56. Wiley, Chichester, 1988.
- [9] Augustin Banyaga. The structure of classical diffeomorphism groups, volume 400 of Mathematics and its Applications. Kluwer Academic Publishers Group, Dordrecht, 1997.
- [10] Sean M. Bates. Toward a precise smoothness hypothesis in Sard's theorem. Proc. Amer. Math. Soc., 117(1):279–283, 1993.
- [11] Patrick Bernard. Connecting orbits of time dependent Lagrangian systems. Ann. Inst. Fourier (Grenoble), 52(5):1533–1568, 2002.
- [12] Patrick Bernard. The dynamics of pseudographs in convex Hamiltonian systems. To appear on J. Amer. Math. Soc., 2007.
- [13] Patrick Bernard. Existence of C^{1,1} critical sub-solutions of the Hamilton-Jacobi equation on compact manifolds. To appear on Ann. Sci. École Norm. Sup. (4), 2007.
- [14] Patrick Bernard. Smooth critical sub-solutons of the Hamilton-Jacobi equation. Math. Res. Lett., 14(3):503-511, 2007.
- [15] Patrick Bernard. Symplectic aspects of Mather theory. Duke Math. J., 136(3):401–420, 2007.
- [16] Patrick Bernard and Gonzalo Contreras. A generic property of families of Lagrangian systems. To appear on Ann. of Math. (2), 2007.
- [17] Henk Broer and Floris Takens. Unicity of KAM tori. Ergodic Theory Dynam. Systems, 27(3):713–724, 2007.

- [18] D. Burago, S. Ivanov, and B. Kleiner. On the structure of the stable norm of periodic metrics. *Math. Res. Lett.*, 4(6):791–808, 1997.
- [19] Ana Cannas da Silva. Lectures on symplectic geometry, volume 1764 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 2001.
- [20] Mario J. Dias Carneiro. On minimizing measures of the action of autonomous Lagrangians. Nonlinearity, 8(6):1077–1085, 1995.
- [21] Chong-Qing Cheng and Jun Yan. Existence of diffusion orbits in a priori unstable Hamiltonian systems. J. Differential Geom., 67(3):457–517, 2004.
- [22] Gonzalo Contreras, Jorge Delgado, and Renato Iturriaga. Lagrangian flows: the dynamics of globally minimizing orbits. II. Bol. Soc. Brasil. Mat. (N.S.), 28(2):155–196, 1997.
- [23] Gonzalo Contreras and Renato Iturriaga. Global minimizers of autonomous Lagrangians. 22º Colóquio Brasileiro de Matemática. [22nd Brazilian Mathematics Colloquium]. Instituto de Matemática Pura e Aplicada (IMPA), Rio de Janeiro, 1999.
- [24] Gonzalo Contreras, Renato Iturriaga, Gabriel P. Paternain, and Miguel Paternain. Lagrangian graphs, minimizing measures and Mañé's critical values. *Geom. Funct. Anal.*, 8(5):788–809, 1998.
- [25] Gonzalo Contreras and Gabriel P. Paternain. Connecting orbits between static classes for generic Lagrangian systems. *Topology*, 41(4):645–666, 2002.
- [26] Albert Fathi. Structure of the group of homeomorphisms preserving a good measure on a compact manifold. Ann. Sci. École Norm. Sup. (4), 13(1):45–93, 1980.

- [27] Albert Fathi. Solutions KAM faibles conjuguées et barrières de Peierls. C. R. Acad. Sci. Paris Sér. I Math., 325(6):649–652, 1997.
- [28] Albert Fathi. Théorème KAM faible et théorie de Mather sur les systèmes lagrangiens. C. R. Acad. Sci. Paris Sér. I Math., 324(9):1043–1046, 1997.
- [29] Albert Fathi. Orbites hétéroclines et ensemble de Peierls. C. R. Acad. Sci. Paris Sér. I Math., 326(10):1213–1216, 1998.
- [30] Albert Fathi. Sur la convergence du semi-groupe de Lax-Oleinik. C. R. Acad. Sci. Paris Sér. I Math., 327(3):267–270, 1998.
- [31] Albert Fathi. Sard, Whitney, Assouad and Mather. Talk at *Recent and Future developments in Hamiltonian Systems*, Institut Henri Poincaré, Paris (France), 2005.
- [32] Albert Fathi. Weak KAM theorem and Lagrangian dynamics. Cambridge Univ. Press. (To appear), 2007.
- [33] Albert Fathi, Alessio Figalli, and Ludovic Rifford. On the Hausdorff dimension of the Mather quotient. In preparation, 2007.
- [34] Albert Fathi, Alessandro Giuliani, and Alfonso Sorrentino. Uniqueness of invariant Lagrangian graphs in a homology class or a cohomology class. Preprint, 2007.
- [35] Albert Fathi and Michael R. Herman. Existence de difféomorphismes minimaux. In Dynamical systems, Vol. I—Warsaw, pages 37–59. Astérisque, No. 49. Soc. Math. France, Paris, 1977.
- [36] Albert Fathi and John N. Mather. Failure of convergence of the Lax-Oleinik semi-group in the time-periodic case. Bull. Soc. Math. France, 128(3):473–483, 2000.

- [37] Albert Fathi and Antonio Siconolfi. Existence of C¹ critical subsolutions of the Hamilton-Jacobi equation. *Invent. Math.*, 155(2):363–388, 2004.
- [38] Giovanni Forni and John N. Mather. Action minimizing orbits in Hamiltonian systems. In Transition to chaos in classical and quantum mechanics (Montecatini Terme, 1991), volume 1589 of Lecture Notes in Math., pages 92–186. Springer, Berlin, 1994.
- [39] Piotr Hajlasz. Whitney's example by way of Assouad's embedding. Proc. Amer. Math. Soc., 131(11):3463–3467 (electronic), 2003.
- [40] Gustav A. Hedlund. Geodesics on a two-dimensional Riemannian manifold with periodic coefficients. Ann. of Math. (2), 33(4):719–739, 1932.
- [41] Juha Heinonen. Lectures on analysis on metric spaces. Universitext. Springer-Verlag, New York, 2001.
- [42] Michael R. Herman. Inégalités "a priori" pour des tores lagrangiens invariants par des difféomorphismes symplectiques. Inst. Hautes Études Sci. Publ. Math., (70):47–101 (1990), 1989.
- [43] Michael R. Herman. On the dynamics of Lagrangian tori invariant by symplectic diffeomorphisms. In Progress in variational methods in Hamiltonian systems and elliptic equations (L'Aquila, 1990), volume 243 of Pitman Res. Notes Math. Ser., pages 92–112. Longman Sci. Tech., Harlow, 1992.
- [44] Andrey N. Kolmogorov. On conservation of conditionally periodic motions for a small change in Hamilton's function. *Dokl. Akad. Nauk SSSR (N.S.)*, 98:527–530, 1954.
- [45] Pierre-Louis Lions, George C. Papanicolau, and Srinivasa R. S. Varadhan. Homogenization of Hamilton-Jacobi equation. Unpublished preprint, 1987.

- [46] Ricardo Mañé. On the minimizing measures of Lagrangian dynamical systems. Nonlinearity, 5(3):623–638, 1992.
- [47] Ricardo Mañé. Generic properties and problems of minimizing measures of Lagrangian systems. Nonlinearity, 9(2):273–310, 1996.
- [48] Ricardo Mañé. Lagrangian flows: the dynamics of globally minimizing orbits. Bol. Soc. Brasil. Mat. (N.S.), 28(2):141–153, 1997.
- [49] Daniel Massart. Normes stables des surfaces. C. R. Acad. Sci. Paris Sér. I Math., 324(2):221–224, 1997.
- [50] Daniel Massart. Vertices of Mather's Beta function, II. Preprint on ArXiv, 2006.
- [51] John N. Mather. Existence of quasiperiodic orbits for twist homeomorphisms of the annulus. *Topology*, 21(4):457–467, 1982.
- [52] John N. Mather. More Denjoy minimal sets for area preserving diffeomorphisms. Comment. Math. Helv., 60(4):508–557, 1985.
- [53] John N. Mather. A criterion for the nonexistence of invariant circles. Inst. Hautes Études Sci. Publ. Math., (63):153–204, 1986.
- [54] John N. Mather. Modulus of continuity for Peierls's barrier. In Periodic solutions of Hamiltonian systems and related topics (Il Ciocco, 1986), volume 209 of NATO Adv. Sci. Inst. Ser. C Math. Phys. Sci., pages 177–202. Reidel, Dordrecht, 1987.
- [55] John N. Mather. Destruction of invariant circles. Ergodic Theory Dynam. Systems, 8*(Charles Conley Memorial Issue):199–214, 1988.
- [56] John N. Mather. Differentiability of the minimal average action as a function of the rotation number. Bol. Soc. Brasil. Mat. (N.S.), 21(1):59–70, 1990.

- [57] John N. Mather. Action minimizing invariant measures for positive definite Lagrangian systems. Math. Z., 207(2):169–207, 1991.
- [58] John N. Mather. Variational construction of orbits of twist diffeomorphisms. J. Amer. Math. Soc., 4(2):207–263, 1991.
- [59] John N. Mather. Variational construction of connecting orbits. Ann. Inst. Fourier (Grenoble), 43(5):1349–1386, 1993.
- [60] John N. Mather. A property of compact, connected, laminated subsets of manifolds. Ergodic Theory Dynam. Systems, 22(5):1507–1520, 2002.
- [61] John N. Mather. Arnol'd diffusion. I. Announcement of results. Sovrem. Mat. Fundam. Napravl., 2:116–130 (electronic), 2003.
- [62] John N. Mather. Total disconnectedness of the quotient Aubry set in low dimensions. Comm. Pure Appl. Math., 56(8):1178–1183, 2003. Dedicated to the memory of Jürgen K. Moser.
- [63] John N. Mather. Graduate course on Arnol'd diffusion. Princeton University, 2003-2008.
- [64] John N. Mather. Examples of Aubry sets. Ergodic Theory Dynam. Systems, 24(5):1667–1723, 2004.
- [65] John N. Mather. Arnol'd diffusion. II. Unpublished Manuscript, 2007.
- [66] Jürgen Moser. On invariant curves of area-preserving mappings of an annulus. Nachr. Akad. Wiss. Göttingen Math.-Phys. Kl. II, 1962:1–20, 1962.
- [67] Jürgen Moser. Convergent series expansions for quasi-periodic motions. Math. Ann., 169:136–176, 1967.

- [68] Jürgen Moser. Monotone twist mappings and the calculus of variations. Ergodic Theory Dynam. Systems, 6(3):401–413, 1986.
- [69] Gabriel P. Paternain, Leonid Polterovich, and Karl Friedrich Siburg. Boundary rigidity for Lagrangian submanifolds, non-removable intersections, and Aubry-Mather theory. *Mosc. Math. J.*, 3(2):593–619, 745, 2003.
- [70] Ludovic Rifford. On viscosity solutions of certain hamilton-jacobi equations: Regularity results and generalized sard's theorems. To appear on *Comm. Partial Differential Equations*, 2007.
- [71] R. Tyrrell Rockafellar. Convex analysis. Princeton Mathematical Series, No. 28.
 Princeton University Press, Princeton, N.J., 1970.
- [72] Dietmar Salamon and Eduard Zehnder. KAM theory in configuration space. Comment. Math. Helv., 64(1):84–132, 1989.
- [73] Dietmar A. Salamon. The Kolmogorov-Arnold-Moser theorem. Math. Phys. Electron. J., 10:Paper 3, 37 pp. (electronic), 2004.
- [74] Sol Schwartzman. Asymptotic cycles. Ann. of Math. (2), 66:270–284, 1957.
- [75] Alfonso Sorrentino. On the total disconnectedness of the quotient Aubry set. Ergodic Theory Dynam. Systems, 28(1):267–290, 2008.
- [76] Hassler Whitney. Analytic extensions of differentiable functions defined in closed sets. Trans. Amer. Math. Soc., 36(1):63–89, 1934.
- [77] Hassler Whitney. A function not constant on a connected set of critical points. Duke Math. J., 1(4):514–517, 1935.
- [78] Zhihong Xia. Arnold diffusion: a variational construction. In Proceedings of the International Congress of Mathematicians, Vol. II (Berlin, 1998), number Extra Vol. II, pages 867–877 (electronic), 1998.